

Received December 26, 2020, accepted January 27, 2021, date of publication February 4, 2021, date of current version February 17, 2021. Digital Object Identifier 10.1109/ACCESS.2021.3057099

A Framework of Software Project Scope Definition Elements: An ISM-DEMATEL Approach

ISMA UL HASSAN[®] AND SOHAIL ASGHAR[®], (Member, IEEE) Computer Science Department, COMSATS University Islamabad, Islamabad 45550, Pakistan

Corresponding author: Isma Ul Hassan (issmaahassan@comsats.edu.pk)

ABSTRACT Software project scope definition is complicated due to the diversity and magnitude of the information needed. An inadequate scope definition often results in project failure as it continues to emerge as the major cause of delays, changes/rework, and cost and schedule overruns. Literature mentions different tools and methods to verify, quantify, and control software scope definition. However, none of these methods and tools help in defining a complete scope. Since a well-defined scope in the early stages is a core ingredient for project success, therefore, previously we developed a method that includes 45 elements of the software project scope definition. Although these elements are noteworthy for project scope, some may influence others; and should be concentrated on more when defining the scope of the software projects. The objective of this present study is that it builds on our previous research and uses Interpretive Structural Modeling (ISM) to extract the interrelationships among the elements and Decision Making Trial and Evaluation Laboratory (DEMATEL) to determine the intensity of these relationships. Experts from academia and the software development industry were consulted to identify the relationships among the elements. ISM-DEMATEL approach indicates that the project manager's competence is the main driver. Further, requirements, stakeholders' expectations, cost estimates, project schedule, resource estimation, project summary, communication, consultation, top management support are among the most influential elements that are critical to focus on. Moreover, the findings of the study provide project managers with a better understanding of the elements and their interconnections thus, helping in achieving a better scope definition.

INDEX TERMS Interpretive structural modeling (ISM), decision making trial and evaluation laboratory (DEMATEL), scope definition, scope elements, scope management, software projects.

I. INTRODUCTION

Over the years, the success of project management has been measured using the three metrics; time, cost and quality also called as "iron triangle" or "triple constraint". While a project is considered to be successful if it satisfies the criteria of on-time, within budget, and according to the scope [1]. Software Project Management (SPM), a subdiscipline of project management plays a significant role in the development and successful completion of software projects. It involves the following activities: project planning, scheduling, budgeting, and risk analysis, etc. [2]. Project planning, which entails all the information about the project, concerning project deliverables, time, cost, communication, resources, etc., is believed to be a major contributor to the

The associate editor coordinating the review of this manuscript and approving it for publication was Francisco J. Garcia-Penalvo¹⁰.

success of the project [2], [3]. During the planning phase, acquiring a precise definition of the project through the scope statement is recognized as one of the most important project management tasks, as it dictates the project's success or failure [2], [4], [5].

Project scope signifies what needs to be done in the project and scope management is the management of what needs to be done [6]. Defining project scope, including all the work in sufficient detail to facilitate project execution, is considered complicated owing to the diversity and magnitude of information needed. Project scope definition requires two kinds of information: project-related information and product-related information [7]. It includes the do's and don'ts of the projects, in other terms, it is the prospect of the project for all the stakeholders [5], [8]. Project scope definition has long been the subject of discussion for its significant relevance to project success [4], [9]–[11]. A project's success or failure heavily depends on its scope definition [4], [9], [12]-[14]. To ensure smooth project execution, a well-defined project scope is considered critical [4]. It helps in successful resource allocations, better estimations, and provides a clear project roadmap [2]. Due to the dependence of schedule and budget on it, the importance of project scope increases further. It takes precedence over the constraints of schedule and budget as it is the most critical aspect of any project to recognize, and if the project scope is well understood, the other two essential constraints can be managed accordingly [15]. Consequently, a poorly defined project scope imposes a variety of adverse effects on the project schedule, cost, and quality [4], [15]. It has been found that incomplete or poor scope definition extends the project time and efforts, disrupts project rhythm, and lowers the morale of the project team [4], [6], [16], [17].

In software development projects, catering to the user, business or organizational needs has a significant influence on project success. Software scoping is a critical project process and project success is heavily dependent on the defined scope [16]-[26]. For several years, software project management has become the focus of attention because of the significant percentage of project failures in the software industry [27]. One of the most crucial reasons for software project failure is not meeting the needs/requirements of the user [16], [25], [28]-[30]. Lack of a clear scope definition has been recognized as a major cause of software project failure [4], [9], [17], [31]–[36]. Few of the reasons associated with software scope definition are partial requirements, unclear objectives, changing requirements, impractical customer expectations [4], [13], [37], poor requirements [9], [38]–[40], insufficient requirements engineering, poor planning, communication gap among stakeholders [41]-[43], lack of user involvement, lack of planning [13], [44], [45]. It can be argued that vague or inadequate scope definition is a major source of risk for software projects.

Due to the significant role of scope definition in the success of software projects, the literature states various methods and tools that have been established to assist with scope definition. These methods and tools help in identifying the features of scope definition such as Work Breakdown Structure (WBS) [46], controlling scope by dealing with scope changes such as feature break down structure and Feature Transition Charts (FTC) [47], scope verification using WBS [46], [48], and quantifying scope using the given requirements such as Function Points (FP) and Function Size Measurement (FSM) [49], [50]. None of these tools or methods help in defining the scope, although they do help with managing the scope that is already defined. However, among these, is a method called Software Project Scope Rating Index (SPSRI), developed for the software industry in our previous work [51]. SPSRI is the most comprehensive method for defining and evaluating the scope definition of software projects. It provides a list of aspects/elements to assess the completeness and quality of software project scope definition. It uses 45 different aspects/elements of software scope definition in a form of weighted checklist to provide guidance. These aspects were identified through a systematic literature review from 1994 to 2016 and were grouped into 11 categories and further grouped into three main sections (shown in Figure 1 in Appendix-A). Aspects were further defined using the detailed description to generate their clear understanding. Moreover, the aspects were validated through a survey of 20 companies that were mainly engaged in software development projects. SPSRI allows practitioners to define better project scope by evaluating its quality. The list of aspects not only helps to assess overall quality but can also help to identify areas/aspects of scope that need attention or further analysis.

Although SPSRI has been proved to be usable and useful when validated, an assessment of the elements list is still lacking because of the complex characteristics of each element. The elements list can be further examined to have a better understanding of the influencing dynamics of the elements, i.e., the degree of driver-dependence among the elements. This sort of information can be used by the project managers to identify the elements they should concentrate on when defining the scope of the software projects. Moreover, the elements could also be evaluated in terms of contribution to the project scope completeness and quality. Even though all scope elements are significant for planning the project, some have low contribution; this is because these elements may be defined briefly at the planning stage, and then in more detail at later stages of the project's development cycle. It is further noted that the most difficult aspect of developing a software project is precisely deciding what to build, and if done incorrectly cripples the development and the project outcomes. This repeatedly acknowledged observation still holds today [16]. Furthermore, the literature lacks research on software project scope elements interrelationships and influencing dynamics. Therefore, this calls for a wise investigation of the elements. An attempt towards building a framework of scope elements will provide a holistic view to better understand the dynamics of the elements. Thus, there is a need to comprehend the software practitioner's views on the interrelationships among elements which can help improve the scope definition process.

Therefore, based on the above-mentioned research gap, in the present study, we attempt to extend our previous work by developing an ISM model that can serve as a framework for classifying the scope elements in the context of their interdependency. In addition, DEMATEL has been used to analyze and categorize the elements into cause-andeffect groups. The aim is to recognize and scrutinize the interrelationships among the software project scope definition elements. The problem undertaken in this study is of the multi-criteria decision type, hence a combined Interpretive Structural Modeling (ISM) and Decision-Making Trial and Evaluation Laboratory (DEMATEL) approach is used. ISM-DEMATEL seems to be an appropriate methodology because of the similarities in both such as causal relationships can be expressed easily using the driving and dependence power in the ISM and the prominence and relation value in



Basis for Project Decision	occuoits/ Categ	S ₃ Execution Approach
C ₁ Project Conceptualization		C ₉ Procurement Strategy
e1 Future expansions		e35 Contractual Terms and Conditions
e2 Capturing Dependency Relationships amo	ng activities	C10 Project Management and Controlling
es Project Strategy/Project Summary		ess Managing Politics
e4 Project Environment		e37 Monitoring and Feedback
es Market Strategy		e38 Project Plan Review
e6 Deployment Strategy		e39 Effective Change Management
C2 Project Requirements		e40 Control and Information Mechanisms
e7 Project Mission(Clear and Realistic Goals	/Requirements)	e41 Track Progress
es Stakeholder Expectations		C11 Deliverables
e9 Identifying Constraints		e42 Setting Milestones
C ₃ People		e43 Key Deliverables
e10 Capable Team Members		e4 Deliverables Dates
e11 Project Manager's Competence		e45 Client Acceptance
e12 Communication and Coordination		
e13 User/Client Involvement		
e14 Client Consultation		
e15 Top Management Support		
e16 Training		
e17 Project Authority		
eis Responsibilities and commitments		
C4 Project Estimating and Scheduling		
e20 Project Finances		
e21 Project Schedule		
e22 Resource Estimation e23 Initial Cost Estimates		
Cs Technical Information		
e24 Technology e25 Troubleshooting/Testing		
C ₆ Organization		
e26 Capturing Corporate Knowledge		
e27 Company's Strategic Intent		
e28 Organizational Capabilities		
e29 Business plan/Vision		
2 Basis of Design		
C7 Project Design		
e30 Alternative Solutions		
e31 Managing Uncertainties		
e32 Proper Equipment/Tool e33 Operational Concepts		
C ₈ Resource Allocation		
e34 Allocate Sufficient Resources		

FIGURE 1. SPSRI sections, categories, and elements.

DEMATEL. This study draws on the literature of software project scope elements, ISM-DEMATEL for exploring the relationships between elements using expert knowledge and identifying the elements that should be focused on more for achieving a high-quality scope definition. The purpose of this research study is to develop a framework of scope definition elements based on the following objectives:

- to review the elements from the perspective of their interdependency and influence on each other
- to derive interrelationships among the elements
- to classify the elements according to their driving and dependence power, and
- to determine the causal and effect relationships of the elements

The main contribution of this study is to determine the hierarchical and causal relationships among the scope definition elements by employing the ISM-DEMATEL methodology. The significance of our work is that the knowledge from this study does not only supplement the existing information but also a better understanding of the scope definition elements can be sought because of the proposed relationship model and the causal diagram. This study provides advanced guidelines based on the identification of critical elements that enhance the scope definition process. Furthermore, the insights from the hierarchical framework and cause-and-effect relationships would help project managers and team members to identify the critical elements on which they should focus more when defining the scope of software projects, and the management of the trade-offs between elements thereby, increasing the chances of project success.

The rest of this article is structured around the following sections: Section II reviews the literature for the significant role of scope definition and introduces the ISM and DEMATEL techniques. Section III summarizes the method used for this research to identify the important software scope definition elements through literature and explains how ISM-DEMATEL was applied to the software scope elements. Section IV provides the results of the study, followed by discussion and resulting guidelines for the project managers in section V. Section VI concludes the study and gives an outlook on future work.

II. LITERATURE REVIEW

This section explores the role of scope definition in project success and introduces ISM and DEMATEL methodologies. Other frameworks using ISM-DEMATEL are also discussed.

A. ROLE OF SCOPE IN PROJECT SUCCESS

Scope management is regarded as more important to project success than any of the other individual knowledge areas. A complete and well-defined scope definition, including both project and product scope, prior to project execution, is integral to project success. Project scope plays a central role in the smooth implementation of the project. In the early stages of the project, a lot of effort is expected from the project team in determining and establishing the project scope, as this effort makes a project successful [15], [52]-[54]. Scope management is strongly linked to other knowledge areas including cost, time, quality, and risk management. Minor variations and ambiguity in project scope can have costly impacts on the cost, time, and quality of the project affecting project success [15]. Project scope proffers the foundation upon which the project is carried out and is vital to the success of the project [4], [16]. Generally, project success or failure can be traceable to a number of factors that include scope, time, and cost [1], [55]. Among these factors, scope definition is the most significant because if the scope definition is vague and there is no effective control of scope, then the project is a failure [4], [18], [19], [21], [56], [57].

It has been reported that software projects are plagued with high failure rates than other industries' projects [58]-[62]. Even today, the percentage of successful software projects is too low and a large number of software projects experience outright failure [63]. According to the recent Standish Group's CHAOS Report [64], 74% of the software projects studied between 2013 to 2017 failed to meet their scope, schedule, and budget. The survey statistics showed that 21% of the software projects failed, often termed aborted or canceled, and 53% were challenged, completed but with time and budget overrun, also with fewer features and functions. Among several causes of software project failures, an inadequate scope is deemed as one of the biggest contributors [4], [9], [16], [17], [32], [35], [57], [65]. It has been recognized that an inadequate or incomplete scope definition at the early stages of the project life cycle is a common source of difficulties in the implementation process [4], [6], [16], [34], [35], [43].

Numerous studies have stressed the role of scope definition and have shown its correlation to project success [4], [9], [16], [66]. Researchers have realized that defining project scope is an essential practice that processes the projects for execution [4], [9], [15], [67]. Properly defined scope helps in completing the project successfully within time, cost, and the required quality [6], [15]. Moreover, the project scope is considered critical and serves as the basis of every decision [68]. Researchers believe that the scope definition of a project should be made clear at the planning phase and understood by all the stakeholders, involved in important decisions of the project [4], [9], [12]. The development of a high-quality scope is considered as one of the essential practices for achieving better project outcomes [4], [7]. Therefore, it is worth noting that as a well-defined scope plays a critical role in project success, and an inadequate scope definition negatively affects it.

B. EXISTING MODELS/FRAMEWORKS USING INTERPRETIVE STRUCTURAL MODELING (ISM) AND DECISION-MAKING TRIAL AND EVALUATION LABORATORY (DEMATEL)

ISM is a mathematically derived methodology for recognizing and summarizing variables relationships, which are used to define or observe a problem or an issue. In ISM methodology, expert judgment is used for variables interdependences decisions, therefore it is interpretive in nature [69]. ISM process transforms a set of distinct directly and indirectly related variables into a comprehensive model. The model represents the interdependencies among the variables emphasizing potential influences they may have on each other. This approach is intended mainly as a group process, but can also be used individually [70]. Like ISM, DEMATEL is another well-established MCDM technique for identifying interrelationships amongst various factors or elements. It provides a cause-and-effect diagram which effectively demonstrates the relationships between the variables/factors. In MCDM problems, DEMATEL aids in obtaining the direct and indirect influences amongst the criteria using the group knowledge [71]. ISM and DEMATEL are complementing and powerful structural modeling techniques and deemed superior to other interpretive and decision modeling techniques [72], [73]. Integrated ISM-DEMATEL has been successfully utilized in various contexts and in various fields to deal with decision-making problems [72], [74]–[76]. However, limited use of ISM-DEMATEL combination has been undertaken by studies in the software engineering domain, therefore studies using these methodologies separately are discussed.

Many studies show the application of the ISM approach in different industries to classify the interrelationships between variables and to highlight the potential influences among these variables using a digraph model or a framework [77]. ISM has been used in many management-related domains for models/frameworks development [78]-[80]. Literature mentions the use of the ISM approach for software projects as well to discuss how different factors correlate with each other and discover the important factors. ISM has been applied to present a framework to explore the relationships between the risk factors of software projects in public organizations [70]. A risk structure model to identify the risk factors of software projects in e-business has also been presented using ISM [81]. Hughes et al. [82] used ISM and presented a framework to identify the interdependencies and influence of success factors of information system projects. Another study by Hughes et al. [77] has used this approach to present a framework that provides insights into the relationships between the information system project failure factors. Another study was done by Rabani and Talebbeydokhti [83] provided the model of success factors of IT projects using ISM methodology. Chang et al. [84] used ISM to identify the interactive causal relationships between agile factors. Sharma et al. [85] incorporated the ISM approach to establish a model to classify the attributes of agile development methodology. Jain et al. [86] used ISM to present a framework that models and measures agile quality attributes so that practitioners can concentrate their efforts on the most critical attributes and thus achieve higher quality. Qu et al. [87] used the ISM approach to construct a model to recognize potential relationships among risk factors of software projects. Bajwa et al. [88] used this approach to construct a model to extract and present relationships among agile development methodology adoption barriers. Another study used the ISM methodology to develop a hierarchical model for analyzing and modeling software process improvement enablers in software SMEs [89].

Tsai *et al.* [90] combined Analytical Network Process (ANP), DEMATEL, and zero-one goal programming (ZOGP) for a sourcing decision about in-house IT functions or contracting to a third-party provider. Fan *et al.* [91] used the extended DEMATEL method to identify IT outsourcing risk factors. Another study combined fuzzy DEMATEL with adaptive Neuro-fuzzy inference system-based multi-criteria decision making (ANFIS MCDM) and intuitionistic fuzzy-based TODIM (IF-TODIM) approaches for better

assessment of software project risks [92]. Asad et al. [93] used a Grey-based DEMATEL method to study and model the flexibility capabilities in an IT-based supply chain. Chen et al. [94], combined the DEMATEL and ANP to investigate the relationships among project management, project risks, and IT organizational performance. Another study used DEMATEL with Multi-Attribute Utility Theory (MAUT) to analyze the upgradations aspects of software [95]. A study integrated fuzzy DEMATEL and fuzzy TOPSIS approaches to assess the global software development risk factors [76]. To make the web application design more sustainable and secure, Agrawal et al. [96] have integrated Fuzzy Analytic Hierarchy Process (Fuzzy AHP) and Fuzzy Technique for Order of Preference by Similarity to Ideal Solution (Fuzzy TOPSIS) to propose an assessment framework for assessing the sustainable-security of web applications with a focus on design perception. Another study by Agrawal et al. [97] used a hybrid integrated Fuzzy Analytical Hierarchy Process-Technique for Order of Preference by Similarity to Ideal Solution (Fuzzy AHP-TOPSIS) method to evaluate various information security factors of healthcare web applications. Al-Zahrani [98] combined Analytic Network Method (ANP), Fuzzy Sets (FS), and Technique for Order of Preference by Similarity to Ideal Solution (TOPSIS) to test the usability-security of healthcare software. Kumar et al. [99] proposed the measuring methodology using Fuzzy AHP-TOPSIS (Analytic Hierarchy Process-Technique for Order of Preference by Similarity to Ideal Solution) to evaluate the usability-security of web applications. Further, the most prioritized attribute contributing to building usable-security of the web applications is identified in the study. Another study by Kumar et al. [100] used a hybrid approach of Hesitant Fuzzy (HF) sets, Analytic Hierarchy Process (AHP) and, Technique for Order of Preference by Similarity to Ideal Solution (TOPSIS) techniques to estimate the security-durability of web applications. To estimate the usability along with the security of the software, in another study, Kumar et al. [101] proposed a hybrid model using Hierarchy Process (AHP), Hesitant Fuzzy (HF) sets, and Technique for Order of Preference by Similarity to Ideal Solution (TOPSIS) techniques. A study by Alenezi et al. [102] used the hybrid method of Fuzzy AHP-TOPSIS (Analytic Hierarchy Process-Technique for Order Preference by Similarity Ideal Solution) to present a prioritization framework for assessing the security of a software in a tactics perspective. In the healthcare sector, fuzzy-AHP has been utilized, which identified blockchain technique as the most prioritized technique for better data integrity management [103]. A study also used an integrated fuzzy-ANP-TOPSIS method for selecting the most appropriate blockchain model for maintaining breach-free Electronic Healthcare Record (EHR) systems [104].

Most of the above-mentioned studies have utilized ISM or DEMATEL methodology or have combined these with other MCDM techniques to provide insights on the success, failure, outsourcing factors, and risk factors of software projects. However, this present study is different in that it is one of the first that focuses on the scope definition elements of software projects and employs ISM and DEMATEL to recognize the hierarchical and causal relationships among these elements to enhance the scope definition process.

III. RESEARCH METHOD

The purpose of this research is to extend our prior research to develop a hierarchical framework of scope definition elements and to distinguish cause and effect elements by 1) identifying the software projects scope definition aspects from 2016 to 2020, 2) deriving interrelationships among the aspects, 3) classifying the aspects according to their influencing and dependence degree, 4) determining the intensity of cause-and-effect relationships among the elements. For this, a combination of the ISM and DEMATEL methods were utilized. The reasons these two methods were combined for this study are that they have similarities such that they are capable of explaining complex relationships between the elements considered in decision making. Secondly, the causeand-effect relationship can easily be revealed through this combination using driving and dependence power in the ISM and the prominence/relation value in DEMATEL. Thirdly, ISM is a macro-oriented technique whereas DEMATEL balances this with a micro-oriented focus, which helps in understanding and visualizing the level of importance of considered elements through well-described diagrams; hierarchical diagram in ISM and causal diagram in DEMATEL [71].

ISM and DEMATEL use experts' opinions as a base for exhibiting the relationships among the elements therefore, this study seeks inputs from two separate expert groups. An expert group of 8 individuals (5 software development industry professionals and 3 academics) was consulted at the initial stage as an input for the ISM methodology to explore the contextual relationships among the elements. Increasing the number of experts could lead to greater stability and reliability of the results of the proposed method. Therefore, DEMATEL methodology was applied using the views of the second group comprising of 7 experts (4 from the software development industry and 3 from academia). The profiles of experts can be seen in Table 3. This expert group (second of the two groups) was also used for the purpose of the ISM framework result validation and identification of cause-andeffect relationships among the elements. The industry experts considered for this study have several years of work experience in software development projects. Similarly, academics are from the discipline of software engineering. To improve the validity and reliability of the research and to evaluate the findings, data triangulation and investigator triangulation were used in this research. Data from academics and software development industry experts was amassed to give this research the perspectives of those who study the phenomenon and who deal with the issues. Moreover, investigator triangulation with two researchers was used to interpret and analyze the qualitative data at each stage of the research.



FIGURE 2. Research steps.

The steps of the research are shown in Figure 2. Details of the steps are mentioned in the upcoming headings.

A. SYSTEMATIC LITERATURE REVIEW (SLR) FOR IDENTIFICATION OF ASPECTS/ELEMENTS

In previous research, a systematic literature review was conducted from 1994 to 2016 to identify the elements of the software project scope definition (see Figure 1 in Appendix-A). The review protocol was developed by taking into account the guidelines from [105], which are commonly used in software engineering domain [106]. The reader is referred to [51] for more details on software project scope elements identification from literature. In this present research, we decided to extend the previous SLR to add new recent studies from 2016 to 2020 for elements identification. In this present study, a temporal update of the SLR has been performed without any major changes to the original review protocol [107].

1) RESEARCH QUESTIONS AND SEARCH STRINGS

The primary objective of this review was to identify the scope definition aspects of software projects from 2016 to 2020. Since we had already reviewed the literature from 1994 to 2016, therefore this review was guided by the research questions and the search terms used in the previous study. The search terms/strings used to search relevant studies are as follows.

(Scope OR scope management OR project scope OR product scope OR software scope OR scoping OR requirements scoping OR scope determination OR scope definition)

AND

(Scope creep OR de-scop* OR over-scop* OR scope issue* OR scope failure)

TABLE 1. Primary search results.

Electronic Data Sources	Result
Google Scholar	351
IEEE Xplore	86
Springer	72
Elsevier Science Direct	47
ACM Digital Library	22

AND

(Cause* OR root cause* OR reason* OR problem*) AND (Effect* OR consequence*)

AND

(Technique* OR method*)

The search term has been adjusted according to the constraints imposed by literature repositories. For example, for the electronic data sources where search string was not accepted as it is, one search term was picked from each of the AND options, and was applied repeatedly, for example: (Scope AND Scope creep AND Cause AND Effect AND Technique).

2) DATA SOURCES

Following electronic data sources were selected to collect relevant literature: IEEE Xplore, Elseviers Science Direct, ACM Digital Library, Springer, and Google Scholar. All the studies, relevant to the objective of this review, were accumulated based on their titles, abstracts, and keywords. Studies included journal papers, conference proceedings, and book chapters.

3) SEARCH PROCESS

The search process consists of two phases: 1) primary search and 2) secondary search. In the primary search phase, the selected literature repositories were searched comprehensively for the relevant studies using the search terms and strings. From the primary search phase, a total of 578 papers were gathered initially as prospective papers. Table 1. presents the results of the primary search. All research papers were then scrutinized based on their titles, abstracts, and keywords. Duplicate papers were eliminated. This phase resulted in the formation of the primary list of papers. Primary list papers were then passed on to the secondary search phase. In the secondary search phase, referenced work from primary list papers was searched in the same electronic data sources. Relevant papers were selected by studying their titles, abstracts, and keywords. This phase resulted in some more papers; a secondary list, that was added to the primary list of papers.

4) STUDY SELECTION

Following three steps were used for the paper selection: (1) an initial set of 578 studies was filtered out first; (2) potential studies were identified out of these; and (3) quality

assessment of the potential studies. In the first step, the initial set of 578 studies was reduced to 381 on the basis of their titles. Then, based on the abstracts, these 381 studies were further reduced to 96. In the second step, full-text filtration was performed by applying inclusion/exclusion criteria that are mentioned in Appendix-B. This step resulted in 73 potential studies.

5) SCRUTINY AND QUALITY ASSESSMENT

Paper titles were examined, and the contents were briefly explored to find the most relevant ones. Papers that did not express the subject area or study emphasis were disqualified. In addition to that, papers available in English only were included. Relevant papers published between 2016 and 2020 were selected. In the case of duplicates, the most recent, comprehensive, and improved study was included. After scrutiny, 41 papers were selected for quality assessment. All papers were assessed against quality criteria shown in Appendix-C using a scoring method. The outcome of this stage was 16 papers.

6) DATA EXTRACTION AND REVIEW FINDINGS

In the data extraction phase, the selected studies were explored in detail to obtain adequate aspects of the software project scope definition. The identified aspects/elements are listed in Table 2 below.

B. COMPARING AND UPDATING ASPECTS/ELEMENTS

From the selected 16 papers, aspects/elements that are needed in the scope definition of software projects to measure its completeness and quality were identified (shown in Table 1). All these new identified aspects/elements from 2016 to 2020 studies were compared with the previous aspects/elements. It was found out that all the aspects/elements are related. Elements descriptions were also checked for comparison purposes and no new aspects/elements were found. For example, Lampa et al. [15] have highlighted the aspects/elements: "clear and accurate requirements", "stakeholders' needs", "user involvement", "budget", "schedule", "constraints" for the scope definition of software projects. Since all these aspects are already in the list of previous elements in the following categories: "Project Requirements", "People" and "Project Estimating and Scheduling". Therefore, no new aspects/elements were added to the list for the ISM-DEMATEL approach. Furthermore, it is worth noticing here, that "project mission (clear and realistic goals/requirements)", "stakeholder expectations", "capable team members", "project schedule and cost estimates" have the highest ranks in SPSRI, and these aspects have also been accentuated in the newly selected 16 studies in constituting a complete scope definition (see Figure 3).

C. FRAMEWORK DEVELOPMENT USING ISM PROCESS-INTERRELATIONSHIPS OF ELEMENTS

Using ISM methodology, this study aims to build a framework of software scope elements to better understand their

TABLE 2. Scope definition aspects/elements.

Authors	Software projects scope definition aspects/elements
Ajmal et al. [35]	Stakeholders' input, project goals, resources, deliverables, budget, and schedule, customer communications
Lampa et al. [15]	Clear and accurate requirements, stakeholders' needs, user involvement, budget, schedule, constraints
Hughes et al. [77]	Project requirements
Goedeke et al. [108]	Clear requirements
Taghavi and Woo [109]	Project goals, objectives, and business needs
Symons et al. [110]	Requirements, constraints, budget, resources, delivery date
Prasanth et al. [111]	Project objectives, constraints, user needs, requirements, stakeholder preferences and needs, resources, cost
Jarzębowicz and Ślesiński [112]	Business goals, realistic cost and schedule, customer acceptance, changes, stakeholders
Sardjono and Retnowardhani [113]	User needs/demands, clear goals, schedule, cost, expectations of stakeholders
Fellir et al. [114]	Types of requirements, relationships between requirements, and constraints
Aguilar et al. [115]	Software requirements, needs of the customers, and restrictions
Carvalho et al. [116]	User roles/responsibilities and interactions, users and stakeholder needs, and system information
Yesudas et al. [117]	Project manager's skills, budget and time estimation, requirements
Tu et al. [118]	Communication and stakeholders
Wahbeh et al. [119]	User's needs, stakeholders' objectives, and requirements
Asadabadi et al. [120]	Requirements and expectations



FIGURE 3. Software project scope definition aspects/elements with their frequencies.

dynamics and help software practitioners to act proactively and focus on potential elements that may help to achieve a better scope definition for software projects. The ISM methodology suggests that expert perspectives based on various techniques such as brainstorming, structured interviews, Delphi method, or group techniques should be used for developing the contextual relationships among the elements. For this reason, semi-structured interviews and focus group discussions were done with group I which consisted of eight experts, three from academia (from a local university) and five from the software industry (3 from a Sweden based company that specializes in software development and 2 from a local software house based in Islamabad Pakistan). The profiles of these experts are presented in Table 3. Different steps of the ISM methodology are discussed in the upcoming headings.

TABLE 3. Profile of experts.

Group	No. of experts	From	Profiles	Experience (years)
			Project Manager	9
			Project Manager	7
	5	Industry	Project Manager	11
т			Team Lead	8.5
1			Senior Developer	6.5
			Assistant Professor	9
	3	Academia	Lecturer	9.5
			Lecturer	8
			Project Manager	13
	4	Inductor	Project Manager	15
	4	industry	Project Manager	12
II			Senior Developer	10.5
			Associate Professor	15
	3	Academia	Associate Professor	12
			Assistant Professor	11

1) STRUCTURAL SELF-INTERACTION MATRIX

To gather the data about the relationships among aspects/ elements a questionnaire (where an empty SSIM was set out for experts' inputs) was designed which was shared with the experts in semi-structured interviews followed by a focus group discussion to analyze and combine the answers. During the semi-structured interviews, the researcher presented the objective of the study and provided the descriptions of the scope elements to experts for clarifying the meaning of each element. After that, the participants were asked to identify all pairwise relationships between the elements. For investigating the relationships, a contextual relationship of "leads to" was chosen as the focus. This denotes that participants were asked to recognize to what extent one element leads to another, using the SSIM to collect this data.

Four symbols: V, A, X, O are used to denote the type of the relationship that exists between the two elements (i, j) under consideration.

V: means element i leads to element j (element i will influence element j)

A: means element j leads to element i (element i will be influenced by element j)

X: represents a bidirectional relationship (elements i and j will be influenced by each other)

O: no relation between the elements

Separate written opinions of participants were taken, to evade any impact one expert may have on another. In the next step, the answers were analyzed, combined, and discoursed with experts in a focus group discussion to attain a final matrix (see Figure 4) representing the consensus of the experts on their decisions. The following statements clarify the way symbols are used in the SSIM:

- According to experts' opinion, element e₁ (future expansions) and elements: e₄₃ (key deliverables), e₄₄ (deliverables dates) and e₄₅ (client acceptance) are unrelated; therefore, the relationship is denoted by O in SSIM
- Element e₇ (project mission) will help achieve the element e₄₃ (key deliverables), therefore, the relationship is denoted by V in SSIM

- Element e₈ (stakeholder expectations) will influence the element e₁ (future expansions); thus, the relationship is represented by A in SSIM
- Element e₇ (project mission) and element e₈ (stakeholder expectations) are related to each other and hence the relationship is an X in SSIM

2) REACHABILITY MATRIX

In this step, SSIM was transformed into the Initial Reachability Matrix (IRM) and Final Reachability Matrix (FRM). For IRM, the notations in SSIM are converted to binary format pursuant to the following rules:

- every (i, j) entry in SSIM as "V" becomes 1 for (i, j) and 0 for (j, i) pair
- every (i, j) entry in SSIM as "A" becomes 0 for (i, j) and 1 for (j, i) pair
- every (i, j) entry in SSIM as "X" becomes 1 for (i, j) and for (j, i) pair as well
- every (i, j) entry in SSIM as "O" means both the (i, j) and (j, i) entries become 0

Using these rules, IRM was created as shown in Figure 5. The next Figure 6 shows the FRM, which was obtained after incorporating the property of transitivity. Here driving and dependence power of elements was also calculated for MICMAC analysis (discussed in section IV and V).

3) LEVELS PARTITIONING

After the completion of FRM, the reachability and antecedent sets of the elements were developed. The reachability set for a particular element includes the element itself together with other elements that it will support, whereas the antecedent set consists of the element itself and the other elements which will help in achieving it. Afterward, the intersection of these sets is derived for all the elements. The elements with the same reachability and intersection sets are given the top level in the ISM hierarchy. These top-level elements do not help in achieving the other elements in the hierarchy. After identifying the top-level elements, the next step is to remove them from the rest of the elements. This same process is repeated until the level is defined for all the elements. These levels are used in building the ISM framework for visual depiction of the elements and their interrelationships [69]. This partitioning process in the present case was completed in 12 iterations. Figure 7 in Appendix-D shows the elements along with their reachability, antecedent, intersection sets, and levels.

4) DEVELOPMENT OF ISM FRAMEWORK

After the levels partitioning process, the final step of the ISM process is the development of the ISM hierarchical framework. Using the final reachability matrix, an initial digraph was obtained. After removing the transitivity links and adding the levels from Figure 7, this digraph was converted into an ISM hierarchical framework as shown in Figure 8.

e1																																										
÷	0								T																														t			
e3	0	0							T																														Ĺ			
e,	0	0	0					T	T	F																													t			
e,	0	0	0	0				T	T	F																													F			
s,	0	0	0	0	0																																		T			
e7	۷	۲	0	0	0	۲		T	T	F																													F			
e,s	۷	×	0	0	0	۲	×		T	F																													T			
ŝ	>	0	0	0	0	0	>	>	T								F	h	h																			F	t			
e10	0	0	0	0	0	0	۲	۲	×																														F			
e11	0	0	0	0	0	0	۲	۲	×	×																									F				F			
e1:	0	0	0	0	0	0	×	×	×	×	0																									F			t			
e13	0	0	0	0	0	0	×	×	×	×	0	×																											F			
e14	0	0	0	0	0	0	×	×	×	×	0	×	×																										T			
e15	0	0	0	0	0	0	×	×	0	×	0	0	0	0																									t			
e16	0	0	0	0	0	0	>	>	0	¥	0	0	0	0	>																								t			
e17	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0																							F			
e18	0	0	0	0	0	0	>	>	>	>	0	>	0	0	0	0	>																						F			
e19	0	0	0	0	0	0	0	0	0	×	0	0	0	0	0	0	0	0																					T			
e20	0	0	>	0	0	0	۲	۲	۲	۲	0	>	0	0	0	0	>	0	0																				F			
e21	0	0	>	0	0	0	×	×	×	>	0	>	0	0	0	0	0	0	0	×																			F			
e22	0	0	>	0	0	0	×	×	×	×	0	>	0	0	0	0	0	×	0	>	>																		T			
623	0	0	>	0	0	0	×	×	×	>	0	>	0	0	0	0	0	0	0	>	۲	۲																	F			
5	0	0	0	0	0	0	>	>	>	>	0	0	0	0	0	0	0	0	0	>	>	×	>																F			
0.26	0	0	0	0	0	0	>	>	>	>	0	0	0	0	0	0	0	0	0	>	>	×	>	0															Г			
e26	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0														Г			
e27	0	0	0	0	0	0	0	0	0	۲	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	×																
e28	0	0	0	0	0	0	>	>	0	¥	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	×	×															
629	0	0	0	0	0	0	×	×	0	×	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	×	×	×														
e.30	0	0	0	0	0	0	>	>	>	>	0	Þ	0	0	0	0	>	0	0	0	>	×	>	0	0	0	0	0	0													
e31	0	0	0	0	0	0	>	>	>	>	0	Þ	0	0	>	0	>	0	0	0	>	×	0	0	0	0	0	0	0	>												
e 32	0	0	0	0	0	0	>	>	0	>	0	0	0	0	0	0	0	0	0	>	>	×	>	×	0	0	0	0	0	>	>											
e33	0	0	0	0	0	0	>	>	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0										
e34	0	0	0	0	0	0	>	>	>	>	0	>	0	0	0	0	>	×	0	>	>	×	>	0	0	0	0	0	0	>	>	>	0						Γ			
e36	0	0	0	0	0	0	>	>	>	>	0	>	0	0	0	0	>	0	0	0	0	0	0	0	0	0	0	0	0	0	0	>	0	0								
e 36	0	0	0	0	0	0	0	0	0	>	0	0	0	0	0	0	>	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0							
e37	0	0	0	0	0	0	>	>	0	>	0	>	0	0	0	0	>	×	0	0	×	0	0	0	0	0	0	0	0	<	0	>	0	>	0	0			Г			
e.13	0	0	0	0	0	0	>	>	0	0	0	>	0	0	0	0	>	0	0	0	>	0	0	0	0	0	0	0	0	×	۲	>	0	¥	0	0	۲		Γ			
e 39	0	0	0	0	0	0	>	>	0	>	0	>	0	0	0	0	>	×	0	0	>	0	0	0	0	0	0	0	0	×	×	>	0	×	0	0	×	×				
e.40	0	0	0	0	0	0	>	>	>	>	0	>	0	0	0	0	>	×	0	0	>	0	0	0	0	0	0	0	0	0	×	0	0	×	0	0	0	×	×			
e.41	0	0	0	0	0	0	0	0	0	>	0	>	0	0	0	0	>	×	0	0	>	>	0	>	0	0	0	0	0	×	0	>	0	×	0	0	×	×	۷	۲		
;r.	0	0	0	0	0	0	>	>	>	>	>	>	>	>	0	0	>	×	0	0	>	×	>	>	0	0	0	0	0	×	×	0	0	0	0	0	×	×	×	۲	۲	
3	0	0	0	0	0	0	>	>	>	>	0	>	>	>	0	0	>	×	0	0	5	×	5	5	×	0	0	0	0	0	×	5	0	5	5	0	4	×	×	1	1	5

770

e45

5553

es es es es e.45

< X X & & > X

ž	0	0	0	0	0	0	٦	-	-1	-	0	0	-	0	0	0	0	0	0	0	-	0	-1	0	0	0	0	0	0		0	0	0		0		-	0	0	0	0	-	- 0
3	0	0	0	0	0	0	-	-	-1	-	0	-	-	-	0	0	-	-	0	0		-	0	0	-	0	0	0	0	0	-	- (۰.		0	0		-	0	0	-	-	- 0
3		0	0	0	0	0	-	-	-	-	0	-	-	-	0	0	-	-	0	0	-	-	-	-	-	0	0	0	0	0	-	- 0			0	0	-	-	0	0	-	-	- 0
-	-	0	0	0	0	0	-	-		-	-		-			•	_	-	0	0	_	-	-	-	0	0	0	0	0	_						-	-	-	0	0	-		0 0
-		-	~	-	-	~	-	-	-	-	-	-	-	-	-	~		_	-	-	-	-	-	_	~	~	-	~	-	-	-				Ĩ	-	-	-	-	-	-		
<u>ہ</u>	-	0	0	-	0	0	0	0	0	-	0	-	0	0		0	-	-	0	-	-	-	0	-	0	0	0	•	0	-	0		-	- 0	-	-	-	0	0	-	-	-	- 0
ँ	•	0	0	•	0	0	-	-	-	-	•	-	•	0	0	0	-	-	0	•	-	0	•	0	•	0	0	0	0	•	-		•	• •	•	°	-	-	-	-	-	-	- 0
5	•	0	0	•	0	0	-	-	•	-	•	-	•	0	0	•	-	-	0	•	-	0	•	0	•	0	0	•	•	-	-	- (<u>-</u>	• •	•	-	-	٦	-	-	-	-	- 0
ŝ	•	0	•	•	•	•	-	٦	•	0	•	-	•	•	•	•	-	0	•	•	-	0	•	0	•	0	0	•	•	-	•	- (°	0	•	۰	-	٦	-	-	-	-	- 0
ŝ	•	0	•	•	•	•	-	-	0	-	•	-	•	0	•	•	-	-	•	•	-	•	•	•	•	•	0	•	•	0	0	- (- ·	• 0	0	-	-	-	•	-	-	-	- 0
e.36	•	0	0	0	0	0	0	0	0	٦	0	0	0	0	0	0	-	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0		0	-	0	0	0	0	0	0	0	0 0
e ye	0	0	0	0	0	0	٦	-	-1	-	0	-1	0	0	0	0	-	0	0	0	0	0	0	0	0	0	0	0	0	0	0	- 0			0	0	0	0	0	0	0	0	0 0
3	0	0	0	0	0	0	-	-	-1	-	0	-1	0	0	0	0	-	-	0	-1	-	-	-1	0	0	0	0	0	0	-	-1	- 0	.	• 0	0	0		-	-1	-	0	0	0 0
3	•	0	0	0	0	0	-	-	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	•	- <	0	0	0	0	0	0	0	0	0	0 0
3	0	0	0	0	0	0	-	-	0	-	0	0	0	0	0	0	0	0	0	-1	-	-	-	-	0	0	0	0	0	-		- (•	0	0	0	0	0	0	0	0	0	0 0
3	0	0	0	0	0	0	-	-	-	-	0	-	0	0	-	0	-	0	0	0	-	-	0	0	0	0	0	0	0	-	-	0 0		0	0	0	0	-	-	0	-	-	- 0
8		0	0	0	0	0	-	-	-	-	0	-	0	0	0	0	-	0	0	0	-	-	-	0	0	0	0	0	0	-	0	0 0			0	-	-	-	0	-	-	0	0 0
2		0	0	0	0	0	-	-	0	0	0	0	0	0	0	0		0	0	0	0	0	0	0	0	-	-	-	-	0	0						0	0	0	0	0	0	0 0
0 2		-	0	0	0	0	_	_	0	0	0	0	0	-	-	-	-	0	0	-	-	0	0	0	~	-	_	-	_	-	-				-	-	0	0	0	-	-	-	
<mark>ہ</mark> 2		-	-	-	-	-	-	-	-	-	-	-	-		-	-	-	-	-	-	-	-	-	-	-					-					Ĩ	Ē	-	-	-	-	-		
8		-	-	-	-	0	-	-	0	-	0	-	-	-	-	-	-	0	-	-	-	-	-	0	-	-	-	-	-	-	-				-	-	-	-	0	-	-	-	
<u> </u>	-	•	•	•	•	•	•	•	•	•	•	•	•	0	•	•	•	•	•	•	•	•	•	•	•	-	-	-	-	•	0			0	•	•	•	•	0	•	•	0	00
្ម័	•	0	•	•	•	•	-	-	-	-	•	•	•	•	•	•	•	•	•	-	-	-	-	•	-	•	•	•	•	•	0			0	•	°	•	•	•	•	•	-	- 0
<u></u>	•	0	0	•	0	0	-	1	-	-	•	0	•	0	0	0	0	0	0	-	-	-	-	-	•	0	0	0	0	0	•	- (0	0	•	0	0	•	•	•	0	00
ŝ	•	0	٦	0	0	0	-	٦	-	٦	0	-	0	0	0	0	0	0	0	-	•	0	-	0	0	0	0	0	0	0	0	0		0	•	•	0	0	0	0	0	0	- 0
ŝ	•	0	-	•	0	0	-	-	-	-	0	-	0	0	•	•	0	-	0	-1	-	-	-	-	-	0	0	0	0	-	-	- (<u>-</u>	• 0	0	•	0	0	0	•	-	-	- 0
ŝ	•	•	-	•	•	•	-	-	-	-	•	-	•	•	•	•	•	•	•	-	-	•	-	•	•	•	•	•	•	•	•	•	^	°	0	-	•	•	•	•	•	•	- 0
°,	•	0	-	0	0	•	0	0	0	0	0	-	0	0	•	•	-	•	•	-	-	0	0	•	•	•	0	•	0	0	0	•		0	0	0	0	•	0	•	0	•	0 0
e 19	0	0	0	0	0	0	0	0	0	٦	0	0	0	0	0	0	0	0	-	0	0	0	0	0	0	0	0	0	0	0	0	0		0	0	0	0	0	0	0	0	0	0 0
e 15	0	0	0	0	0	0	٦	1	-	1	0	-	0	0	0	0	1	-	0	0	0	1	0	0	0	0	0	0	0	0	0	0		• 0	0	-	0	1	-	-	-	-	1 0
5	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	-	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0		0	0	0	0	0	0	0	0	0	0 0
5	0	0	0	0	0	0	-	-	0	0	0	0	0	0	-	-	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0 0	•	0	0	0	0	0	0	0	0	0	0 0
ä	0	0	0	0	0	0	-	-	0	0	0	0	0	0	-	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	•	0	0	0	0	0	0	0	0	0	0 0
7	•	0	0	0	0	0	-	-	0	-	0	-	-	-	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0		0	0	0	0	0	0	0	0	0	0 0
2	0	0	0	0	0	0	-	1	0	-	0	-	-	-	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0		0	0	0	0	0	0	0	0	0	- 0
a l		0	0	0	0	0	-	-	0	1	0	-1	-	-	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0 0		0	0	0	0	0	0	0	0	0	0 0
-	。	0	0	0	0	0	0	0	0	0	-	0	0	0	。	。	。	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0 0			0	0	0	0	0	0	0	0	0 0
2		0	0	0	0	0	0	0	0	-		-	-	_	_	_		0	-		0		0	0	0	0			-	0	0				0	0	0	0	0	0	0	0	0 0
<u>ہ</u>		-	0	0	0	0	1	_	-		-	-	_				_	0	0	-	-		-	0	0	0	0	0	0	0	-				6	0	0	0	0	0	0	-	0.0
-		~	~	-	~	~		-	-	-		-			_	_	_	~	~		Ē			~	~	~	~	~	-	~					2	ĥ	6	~	~	~	~	~	
<u>,</u>		-	-	-	0	с С	-	-1	-	-	-	-	-					-	-	-	-	-	-	-	-	-	0	0	-1	-	-				-	6	0	0	0	-	-		
<u> </u>	•	0	0	0	0	0	-	-	0	-	-	-	-	-	-	0	0	0	0	-	-	-	-	0	0	0	0	0	-	0	0	0		°	0	0	0	0	0	0	0	0	00
ĕ	•	0	0	0	0	-	-	-	0	0	•	0	0	0	•	•	0	0	0	•	0	0	0	0	0	0	0	0	0	0	0	0		0	0	0	0	0	0	0	0	0	00
ő	0	0	0	0	٦	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0		0	0	0	0	0	0	0	0	0	0 0
ર	•	0	0	-	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0		o	0	0	0	0	0	0	0	0	0 0
ŝ	•	0	-	0	0	0	0	0	0	0	•	0	0	0	•	•	0	0	•	0	0	0	0	0	•	•	0	0	0	0	0	0	•	0	•	۰	0	0	0	0	0	0	0 0
ő	•	-	•	0	0	0	-	-	0	0	•	0	0	•	0	0	•	•	•	•	0	0	•	0	•	0	0	•	0	•	•	0		0	0	•	0	0	0	0	•	0	0 0
ő	-	0	0	0	0	0	٦	-	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0			0	0	0	0	0	0	0	0	0 0
	e,	e2	e3	e,	êε	e 6	e,	ês	e,	e10	e 11	e12	e13	e14	e16	e16	e17	e18	e19	e 20	e21	e 22	e23	e24	e 25	e26	e27	e25	e29	¢30	e.31	e 32	e S	16.31	6.16	e37	e.35	0.39	e.40	6.41	¢.42	e.13	e 41

FIGURE 5. Initial Reachability Matrix (IRM).

	ng						_		~		~	~	~	~			~	_	~		~			10		~		_	~			_	~		~								~	~		1016
	Drivi	1	-	5	-	-	- *	ñ	ñ	ñ	ñ	35	33	3	č	ŝ	5	5	5	-	'n	3	Э,	3	5	5	9	5	5	M	5	5	6		17		- 6	1 =		12	-	8	5	5	-	016/
	e45]	1*	0	-	0	0	0 -	-		-		-	*	-	*_	*_	*_	*	*_	0	*		*	-	*	*	0	-*	*_	*_		*	<u>*</u> ,	0	<u>*</u> -		> -	•	. *	. *.	*	. *	-	*		37 1
I I	e 44	-*	0		0	0	0 -	-	-	-		-	-		-	<u>*</u>	<u>*</u>	-	-	0	<u>*</u>	-	-	<u>*</u>	*	-	0	-	<u>*</u>	-	-	-		0			> *		• -	- *-	*		-		<u>*</u>	37
I I	e43	<u>*</u> -	0	<u>*</u>	0	0	0 -	-	-	-	-	-	-	-		<u>*</u>	<u>*</u> .	-	-	0	<u>*</u>	-	-		-	-	0		<u>*</u>	<u>*</u>	<u>*</u>		-	0			> <u>*</u>		• -	- *-	*		-	-	<u>*</u> .	37
No No<	1 e42	-	0		0	0	0 -	-					-		 	-* -	- -			0	*-	-	-		-		0		*- -				* '	0	*_ <		> -	•	• -	- *-	*	,	0	*	<u>*</u>	35
	10 G4	0 *	•	* 1*	0	0	• *	-	<u>*</u>	<u>*</u>	-	*	-	* *	*	* - *	*- *	-		0	*_ *	-		* 1*		*	•	*	*	*				0	- *		> -	•	· *		-	•	-	-	•	5 35
1 1	39 G4	:1 0	0	1 *	0	0	0 - 0 -	-	_	*	-		-	 *				-	-	0	 *	-					0					-	<u> </u>	0 0	*		> = 	• -			_		-	-	0	5 3(
1 1	38 e	0	0	*	0	0	0 -	_	_	*	*	 *	_	*	*	*	0	_	*	0	 *	_		*		*	0	0	- 0	*	_	*	_	0	* *		> *				-		_	_	- 0	32 3
1 1	e37 e	0	0	1*	0	0	0 -	_	_	*	_	1*		*	-	- *_	*	-	-	0	*	-		1*			0		*	1*				0	- *		- <	, , ,		- *	-				0	35 3
1 1	e36 (0	0	0	0	0	0	-	0	0		1*	*	*	*	*	*		0	0	*	0	*	0	0	0	0	*_	*	*	0	0	0	0	0	> -				0	0	0	0	0	0	4
1 1	e35	-	0	0	0	0	0 -	-	-	-		1*		*	*	<u>*</u>	1*		0	0	<u>*</u>	-*	0	1*	1*	0	0	1*	-*	-*	-*	1*		0	o -					0	C	0	0	0	0	23
	e34	*	0	*	0	0	0 -	-	-	-		-*	-	<u>*</u>	<u>*</u>	<u>*</u>	<u>*</u> _	-	-	0	-	-	-		*_	<u>*</u>	0	<u>*</u>	<u>*</u>	<u>*</u>				0	- 0		> <u>*</u>		•		-	*	<u>*</u>	*	0	35
1 1	e33	0	0	0	0	0	0 -	-	-	0	<u>*</u>	-*	*_	<u>*</u> -	<u>*</u>	<u>*</u>	0	0	0	0	-*	-*	<u>*</u>	<u>*</u>	0	0	0	0	0	-*	0	0	0	-	0					0	С	0	0	0	0	14
1 1	e32	0	0	-	0	0	0 -	-	-	<u>*</u>	-	1*	<u>*</u>	<u>*</u>	~	÷	<u>*</u>	<u>*</u>	-*	0	-	-	-		-	<u>*</u>	0	-*	<u>*</u>	-*			-	0	*_ <		> <u>*</u>		,	> <u>*</u>	*	*	-*	<u>*</u>	0	33
1 1	e31	*	0		0	0	0 -	-	-	-	-		-	<u>*</u>	*		<u>*</u>	-		0	<u>*</u>	-	-		<u>*</u>	<u>*</u>	0	-*	<u>*</u>	<u>*</u>		-	<u>*</u> ,	0	* *		> <u>*</u>	•	• -		*		-	-	0	36
1 1	9 G3(0		0	0	0 -	-	-	-		, 1*		*	*	÷	<u>*</u>	-		0	* 				*	-*	0	÷.	<u>*</u>	0		÷.	0	0	*_ <		> -	•	• -		-	••		*-	0	5 33
1 1	8 e2	0	•	0	•	•	• -	-	-	•	*	*1	*1	*			•	•	0	0	<u>*</u>	*	*	* 1*	•	•	-	-	-	-	•	•	0	•							0	0	•	•	•	4 16
1 1	27 C 2	0	0	0	0	0	0 - 0 *	-	 *	0 0	-	1.	1:	1	0		0	0	0	0	0	-1	1	1	0	0	-	-	-	-	0	0	0										0	0	0	1 17
i i	26 e	0	0	0	0	0	0 - 0 *		 *	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	_	_	_	_	0	0	0	-									0	0	0	5
i i	25 e	*	- 0	*	- 0	-	- -	-	-	-	_	*	*	*	*	*	*	*	*	*	-	-	-	-	*	-	0	*	*	*	*	*	*	- -	* *		> *	*	*	. 0	0	, *	-	-	-	35
I I	224 e	*	0	*	0	0	0 -		_	_			*	*	*	*	*.	*	*	1*			-		-	*	0	*	*_	*	*	*	_	0	*		> *				0	0	*	*	0	31 3
I I	e 23 (*	0	-	0	0	0 -	_	-	-		1*		*	*	<u></u> *	1*	-*	1*	1*	-	-*	-*		1*	-	0	0	<u>*</u>	-*	0	_*	*.	0	* *	_ <		*	,	0	c	. *_	1*		0	31
I I	e 22	*	0		0	0	0 -	-	-	-		1*		<u>*</u>	*	*	*	*		-*							0	*	*	*				0	- *	_ <	> *	*		. *.	*				0	37
F1 C2 C3 C4 C5 C5<	e ₂₁	-*	0		0	0	0 -	-	-	-		1*		<u>*</u>	*	<u>*</u>	-*	*	-*	1*			*		-*	*	0	-*	<u>*</u>	-*	0	-*	<u>*</u> ,	0	* *	_ <	> -	. *		. 0	C	, *	.*	-	0	34
F1 C2 C3 C4 C5 C5<	e 20	0	0	-	0	0	0 *	-	<u>*</u>	-*	-*	0	-	-*	*	0	0	-	0	0	-		0	0	-*	0	0	0	0	0	0	0	0	0	0		> *	, c		0	C	0	0	0	0	13
I I	e19	0	0	0	0	0	0		•	0	-	-*	*_	<u>*</u>	*	<u>*</u>	*_	0	0	-	*-	0	<u>*</u>	0	0	0	0	0	0	0	0	0	0	0	0					0	C	0	0	0	0	10
1 1	r e 18		0	-*	0	0	0 -	-	-	-	-	1*	-	<u>*</u> -	*		-	-	-	1*	-	-	-	-*	-	-*	0	0	0	0	0		* '	0			> -	• *-	• -		-	•	-		0	32
ci ci<	6 e13	0	0	0	0	0	0		0	0	0	0	0	0	0	0	0	-	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0					0	0	0	0	0	0	
e1 c2 c3 c4 c5 c5<	5 e1	0	•	0	0	0	o -	-	-	•	*	* 1*	*1		*		-	0	0	0	*	*	*	* 1*	•	0	•	0	•	0 *	0	0	0	•)	C	0	0	0	•	3 13
61 62 63 64<	14 e1	0	0	0	0	0	0 - 0 -		-	0 0		-1 *				*	0 *	0	0	0 *	 *				0	0	0	0	0		0	0	0	0									0	0	• *	6 1:
FI C2 C3 C4 C5 C4 C4 <thc4< th=""> C4 C4 C4<!--</th--><th>13 e</th><th>0</th><th>- 0</th><th>0</th><th>0</th><th>0</th><th>0 -</th><th>_</th><th>_</th><th>*</th><th>_</th><th></th><th>_</th><th>_</th><th>_</th><th>*</th><th>*</th><th>0</th><th>0</th><th>1</th><th>*</th><th>*</th><th>*</th><th>*</th><th>0 0</th><th>0</th><th>0</th><th>0</th><th>*</th><th>*</th><th>*</th><th>0</th><th>0</th><th>-</th><th>o *</th><th></th><th></th><th>*</th><th>, c</th><th></th><th></th><th></th><th>*</th><th>0</th><th>-</th><th>3 1</th></thc4<>	13 e	0	- 0	0	0	0	0 -	_	_	*	_		_	_	_	*	*	0	0	1	*	*	*	*	0 0	0	0	0	*	*	*	0	0	-	o *			*	, c				*	0	-	3 1
et et<	312 e	0	0	0	0	0	0 -	_	_	-		*				*	*	0	0	*	*	*	*	*	0	0	0	0	*	*	- 0	0	0	0	0						0	0	0	0	*	17 2
cicicicicicicicicici100000000ci100000000ci000000000ci000000000ci000000000ci000000000ci100000000ci100000111ci10000111ci10000111ci10000111ci10000111ci10000111ci10000111ci100000111ci100000011ci1111111ci1111111ci1111	611 (0	0	0	0	0	0	-	0	0	0	1	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0					0	C	0	0	0	0	-
ei ci ci ci ci ci ci ci ci ei 1 0 0 0 0 0 0 0 0 ei 1 0 0 1 0 <t< th=""><th>e10</th><th>0</th><th>0</th><th>-*</th><th>0</th><th>0</th><th>0</th><th>-</th><th>*</th><th>*</th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th>*</th><th>*</th><th></th><th></th><th>*</th><th></th><th>*</th><th>*</th><th>*</th><th>0</th><th>0</th><th></th><th></th><th>0</th><th>*</th><th>*</th><th>0</th><th>*</th><th></th><th></th><th></th><th></th><th>0</th><th>C</th><th></th><th>*</th><th>*</th><th>*</th><th>29</th></t<>	e10	0	0	-*	0	0	0	-	*	*								*	*			*		*	*	*	0	0			0	*	*	0	*					0	C		*	*	*	29
e1 e2 e3 e4 e5 e6 e7 e8 e2 1 0 <td< th=""><th>e9</th><th>-</th><th>0</th><th>1*</th><th>0</th><th>0</th><th>0 -</th><th>-</th><th></th><th>-</th><th></th><th></th><th></th><th>-</th><th>-</th><th>*</th><th>-*</th><th>0</th><th>*</th><th>0</th><th></th><th></th><th></th><th>-</th><th>-</th><th>-*</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>-*</th><th><u>*</u></th><th>0</th><th>*</th><th></th><th>> <u>*</u></th><th>, c</th><th></th><th>0</th><th>c</th><th>*</th><th>*</th><th>*</th><th>*</th><th>27</th></td<>	e9	-	0	1*	0	0	0 -	-		-				-	-	*	-*	0	*	0				-	-	-*	0	0	0	0	0	-*	<u>*</u>	0	*		> <u>*</u>	, c		0	c	*	*	*	*	27
e1 c2 c3 c4 c5	e8	0	0	-	0	0	0 -	-	-	*	-			-	-	-	-*	0	*	0	-	-	-	-	-*	*	-*	-*	-*	-	*	-*	<u>*</u>	0	*		> *	. 0		0	C	*	-*	*	<u>*</u>	31
ei	еı	0	0	-*	0	0	0 -	-	-		-	-	-	-	-	-	<u>*</u>	0	-*	0	-	-	-	-	-*	<u>*</u> -	-*	-*	-*	-	<u>*</u>	-*	<u>*</u> ,	•	*		> *	. 0		0	C	*	*	*	<u>*</u>	31
e1 e2 e3 e4 e3 e3 0 0 0 0 0 e4 1 0 0 0 0 0 e5 0 1 0 0 0 0 0 e5 0 0 0 0 0 0 0 0 e7 0 <t< th=""><th>. е₆</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th></th><th>-</th><th>-</th><th>0</th><th></th><th>-</th><th>-*</th><th>*</th><th>-*</th><th>-*</th><th>0</th><th>0</th><th>0</th><th>0</th><th>-*</th><th></th><th></th><th>-*</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th></th><th></th><th></th><th></th><th>0</th><th>C</th><th>0</th><th>0</th><th>0</th><th>0</th><th>13</th></t<>	. е ₆	0	0	0	0	0		-	-	0		-	-*	*	-*	-*	0	0	0	0	-*			-*	0	0	0	0	0	0	0	0	0	0	0					0	C	0	0	0	0	13
ei ci ci <thci< th=""> ci ci ci<</thci<>	4 es	0	0	0	0	-	0		0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0						C	0	0	0	0	-
e1 e2 e3 e2 1 0 1 e3 0 1 0 1 e4 1 0 0 1 1 e5 0 0 0 1 1 1 e5 0 0 0 0 0 1 1 e5 0 0 0 0 0 1 <td< th=""><th>3 e4</th><th>0</th><th>0</th><th>0</th><th>-</th><th>0</th><th>0</th><th></th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th></th><th></th><th></th><th></th><th></th><th></th><th>0</th><th>0</th><th>0</th><th>0</th><th>0</th><th>-</th></td<>	3 e4	0	0	0	-	0	0		0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0	0							0	0	0	0	0	-
e_1 e_1 e_1 e_1 e_1 e_2 e_3 0 e_4 0 e_4 0 e_6 0 e_6 0 e_1 1 e_1	2 e	0 (0	1	0	0	0	، د _	- -	0	° *	0 *	0 *	0 *	0 *	0 *	0	0	0	0	0 *	0 *	0 *	0 *	0	0	0	0	0	0	0	0	0								0		0	0	0	3 1
e1 e1 e2 e3 e3 e3 e4 e3 e5 e3 e6 e3 e7 e3 e6 e3 e7 e3 e7 e3 e7 e3 e1 e1 e2 e1 e1 e1 e1 e1 e2 e1 e2 e1 e3 e1 e3 e2 e3 <	а I,	Ĕ)) (~		-		_	 *	 *	 *	 *	 *	 *)) () (-	 *	 *	 *	 *)) (~	-	_	-	-	~	_								0		-	- -	~	3 1
e1 e2 e3 e3 e6 e1 e1 e1 e1 e1 e1 e1 e1 e2 e2 e2 e2 e2 e2 e2 e2 e2 e2 e2 e2 e2	e	F	-	-	-	-		-		-		-		-	-	-	-	-	-	-	1	-	-	-	5	_	-	-	-	-	-	-	_						+		-		F	Ĕ	-	e 1
		e1	e2	e3	e4	es	e6	e7	e8	69	e10	e ₁₁	e12	e13	e14	e15	e16	e17	e ₁₈	e19	e ₂₀	e ₂₁	e22	e ₂₃	e24	e 25	e ₂₆	e27	e28	e29	e30	e ₃₁	e32	e33	e34	e35	C36	6.9	5.5	C39	641	e41 642	C 43	e44	e45	Dependenc

FIGURE 6. Final Reachability Matrix (FRM) with driving and dependence power of elements.

D. CAUSAL-EFFECT ANALYSIS USING DEMATEL

DEMATEL is a mathematical method, suitable for the study of complex causal relationships among the elements. In the multi-criteria decision problem, it helps to obtain direct and indirect influences between criteria. DEMATEL methodology can assist in computing the relationship and strength among the involved elements. Using DEMATEL, elements can be easily distinguished into cause and effect groups [71]. The second part of the study categorizes the software project scope definition elements into cause-andeffect groups. Before this exercise, a group of experts (second of the two groups) was engaged in the ISM model validation to ensure model integrity. This group who took part in the ISM model validation and DEMATEL exercise constituted of seven experts, four of whom were from the software industry (two local software houses based in Islamabad Pakistan) with a broad knowledge of software project development and management and three active academicians in an Islamabad university. The group experts' profiles have been summarized in Table 3. Following the procedure of the DEMATEL method, the experts were asked to determine the intensity of the influence between the elements through the use of scale and pairwise comparisons using a questionnaire survey and focus group discussions. The steps involved in the DEMA-TEL methodology are discussed in the upcoming headings.

1) DIRECT RELATION MATRIX

In the DEMATEL method, the starting point is direct relation matrix A. This matrix is generated using pairwise comparisons, where the experts estimate the degree of direct influence between the elements. Within A, a_{ij} is denoted as the influence of the element i on element j, and all principal diagonal elements of A are set to zero. In this first step, to generate A, a questionnaire survey and focus group discussion was undertaken. For each expert, a questionnaire in the form of an $n \times n$ matrix (n equals the number of elements) is designed. A 5-point Likert scale with 1 to 5 influence levels representing "No influence," "Low influence," "Equal influence," "High influence," and "Very high influence," respectively, were designed for the pairwise comparisons. Figure 9. shows the average direct relationship matrix A, which is constructed from the aggregation of the arithmetic means of responses given by the seven experts on each pair of comparisons on the DEMATEL survey.

2) NORMALIZED DIRECT RELATION MATRIX

In this step, the normalization of the direct relation matrix is done. Based on the direct relation matrix A, the normalized direct relation matrix can thus be obtained through the following equations (1) and (2):

$$Y = \frac{1}{s}A\tag{1}$$

$$S = \max\left\{\max_{1 \le i \le n} \sum_{j=1}^{n} \alpha_{ij}, \max_{1 \le j \le n} \sum_{i=1}^{n} \alpha_{ij}\right\}$$
(2)

The normalized direct relation matrix of elements is shown in Figure 10.

3) TOTAL RELATION MATRIX

In step 3 of DEMATEL methodology, the total relation matrix T is computed, using the equation (3), which is shown in Figure 11.

$$T = Y(I - Y)^{-1} \quad where \quad \lim_{d \to \infty} = [0]_{n \times n} \tag{3}$$

Here I refer to the identity matrix.

4) SUM OF ROW AND COLUMNS FROM MATRIX T

In the next step, the summation of rows and columns of the elements were computed. Vectors D and R, are used to represent the sum of rows and sum of columns, respectively in the total relation matrix T. D_i (sum of the ith row in matrix T) value indicates the total given both direct and indirect effects, that element *i* has on other elements and R_j (sum of the jth column in matrix T) signifies the total received both direct and indirect effects, that all other elements have on element j. These values are shown in Table 4.

5) CAUSE-EFFECT DIAGRAM

After calculating the D and R vectors, in the next step, (D+R) and (D-R) values were calculated which are tabulated in Table 4. The importance order of elements is obtained through the (D+R) dataset, which represents the "centrality degree" where $(D_i + R_j)$, wherein j = i, indicates the strength of interrelatedness that an element i has with the other elements. Furthermore, "causality degree" is represented by (D-R), which separates the elements into a cause group and an effect group. If (D-R) comes out to be positive, the element is categorized into the cause group and if a negative value is obtained, the element is placed into the effect group [71]. Figure 12. shows the cause-and-effect relationship diagram of the elements.

IV. RESULTS

This section summarizes the important results obtained from the two analyses. Sub-section C presents the first of the two analyses that used expert group I. Data about the relationships among elements using ISM methodology was gathered using semi-structured interviews and focus group discussions with five software industry and three academic experts. Further, the Cross-Impact Matrix Multiplication Applied to Classification (MICMAC) analysis was adopted to examine the influencing dynamics among the elements. MICMAC is not a formal step in ISM methodology, but an additional step for the visual representation of the driving and dependency power of elements using a structured quadrant separated matrix [69]. Based on the driver-dependence degree of the elements (see Figure 6), the strength of the relationships of elements was examined. After that, the elements were classified into the following four clusters: autonomous, linkage, dependent, and independent presented in Figure 13. Elements with weak

Element	Reachability set: R (Pi)	Antecedent set: A (Pi)	Intersection R (Pi)∩A (Pi)	Level
e1	e ₁ ,e ₉ , e ₁₈ , e ₂₁ , e ₂₂ , e ₂₃ , e ₂₄ e ₂₅ , e ₃₀ , e ₃₁ , e ₃₄ , e ₃₅ , e ₄₀ , e ₄₂ e ₄₃ , e ₄₄ , e ₄₅	$e_1, e_7, e_8, e_{10}, e_{11}, e_{12}, e_{13}, e_{14}, e_{15}, e_{20}, e_{21}, e_{22}, e_{23}$	$e_1, e_{21}, e_{22}, e_{23}$	VIII
e ₂	e ₂	e ₂ , e ₇ , e ₈ , e ₁₀ , e ₁₁ , e ₁₂ , e ₁₃ , e ₁₄ , e ₁₅ , e ₂₀ , e ₂₁ , e ₂₂ , e ₂₃	e ₂	I
e3	e ₃ , e ₇ , e ₈ , e ₉ , e ₁₀ , e ₁₈ , e ₂₀ , e ₂₁ , e ₂₂ , e ₂₃ , e ₂₄ , e ₂₅ , e ₃₀ , e ₃₁ , e ₃₂ , e ₃₄ , e ₃₇ , e ₃₈ , e ₃₉ , e ₄₀ , e ₄₁ , e ₄₂ , e ₄₃ , e ₄₄ , e ₄₅	e ₃	e ₃	хп
e4	e4	e4	e4	I
e5	e ₅	e5	e ₅	1
e 6	e ₆	e ₆ , e ₇ , e ₈ , e ₁₀ , e ₁₁ , e ₁₂ , e ₁₃ , e ₁₄ , e ₁₅ , e ₂₀ , e ₂₁ , e ₂₂ , e ₂₃	e ₆	I
e 7	e ₁ , e ₂ , e ₆ , e ₇ , e ₈ , e ₉ , e ₁₀ , e ₁₂ e ₁₃ , e ₁₄ , e ₁₅ , e ₁₆ , e ₁₈ , e ₂₀ , e ₂₁ , e ₂₂ , e ₂₃ , e ₂₄ , e ₂₅ , e ₂₆ , e ₂₇ , e ₂₈ , e ₂₉ , e ₃₀ , e ₃₁ , e ₃₂ , e ₃₃ , e ₃₄ , e ₃₅ , e ₃₇ , e ₃₈ , e ₃₉ , e ₄₀ , e ₄₁ , e ₄₂ , e ₄₃ , e ₄₄ , e ₄₅	e ₃ , e ₇ , e ₈ , e ₉ , e ₁₀ , e ₁₁ , e ₁₂ , e ₁₃ , e ₁₄ , e ₁₅ , e ₁₆ , s e ₁₇ , es ₁₈ , e ₁₉ , e ₂₀ , e ₂₁ , e ₂₂ , e ₂₃ , e ₂₄ , e ₂₅ , e ₂₆ , e ₂₇ , e ₂₈ , e ₂₉ , e ₃₁ , e ₃₂ , e ₃₅ , e ₃₇ , e ₄₂ , e ₄₃ , e ₄₄ , e ₄₅	e7, e8, e9, e10, e11, e12, e13 e14, e15, e16, e18, e20, e21, e22, e23, e24, e25, e26 e27, e28, e29, e31, e32,e35, e37, e42, e43, e44, e45	IX
es	e ₁ , e ₂ , e ₆ , e ₇ , e ₈ , e ₉ , e ₁₀ , e ₁₂ , e ₁₃ , e ₁₄ , e ₁₅ , e ₁₆ ,e ₁₈ , e ₂₀ , e ₂₁ , e ₂₂ , e ₂₃ , e ₂₄ , e ₂₅ , e ₂₆ , e ₂₇ , e ₂₈ , e ₂₉ , e ₃₀ , e ₃₁ , e ₃₂ , e ₃₃ , e ₃₄ , e ₃₅ , e ₃₇ , e ₃₈ , e ₃₉ , e ₄₀ , e ₄₁ , e ₄₂ , e ₄₃ , e ₄₄ , e ₄₅	e ₃ , e ₇ , e ₈ , e ₉ , e ₁₀ , e ₁₁ , e ₁₂ , e ₁₃ , e ₁₄ , e ₁₅ , e ₁₆ , e ₁₇ , e ₁₈ , e ₁₉ , e ₂₀ , e ₂₁ , e ₂₂ , e ₂₃ , e ₂₄ , e ₂₅ , e ₂₆ , e ₂₇ , e ₂₈ , e ₂₉ , e ₃₀ , e ₃₁ , e ₃₂ , e ₃₄ , e ₃₇ , e ₄₁ , e ₄₂ , e ₄₃ , e ₄₄ , e ₄₅	e7, e8, e9, e10, e12, e13, e14, e15, e16, e18, e20, e21 e22, e23, e24, e25, e26, e27, e28, e29, e30, e31, e32, e34, e37, e41, e42, e43, e44, e45	іх
e9	e7, e8, e9, e10, e13, e18, e20, e21,e22 e23, e24, e25, e30, e31, e32, e34, e35, e37, e38, e39, e40, e41, e42, e43, e44, e45	e ₁ , e ₃ , e ₇ , e ₈ , e ₉ , e ₁₀ , e ₁₁ e ₁₂ , e ₁₃ , e ₁₄ , e ₁₅ , e ₁₆ , e ₁₇ ,e ₁₈ e ₁₉ , e ₂₀ , e ₂₁ , e ₂₂ , e ₂₃ , e ₂₄ , e ₂₅ , e ₂₇ , e ₂₈ , e ₂₉ , e ₃₁ , e ₃₂ , e ₃₄ , e ₃₇ , e ₄₂ , e ₄₃ , e ₄₄ , e ₄₅	e7, e8, e9, e10, e13,e18, e20, e21, e22, e23, e24, e25, e31, e32, e34, e37, e42, e43, e44, e45	VII
e 10	e ₁ , e ₂ , e ₆ , e ₇ ,e ₈ ,e ₉ ,e ₁₀ , e ₁₂ , e ₁₃ , e ₁₄ , e ₁₅ , e ₁₆ ,e ₁₈ , e ₁₉ ,e ₂₀ ,e ₂₁ ,e ₂₂ , e ₂₃ e ₂₄ e ₂₅ , e ₂₈ , e ₂₉ , e ₃₀ , e ₃₁ , e ₃₂ , e ₃₃ , e ₃₄ e ₃₅ ,e ₃₆ ,e ₃₇ ,e ₃₈ , e ₃₉ , e ₄₀ , e ₄₁ ,e ₄₂ e ₄₃ e ₄₄ , e ₄₅	e3, e7, e8, eS9, e10, e11, e12, e13, e14, e15, e16, e17, e18, e19, e20, e21, e22, e23, e24, e25, e26, e27, e28, e29 e31,e32, e34, e42, e43, e44, e45	e7, e8, e9, e10, e12, e13, e14, e15, e16, e18, e19, e20, e21, e22, e23, e24, e25, e28, e29 e31, e32, e34, e42, e43, e44, e45	іх
e 11	e ₁ , e ₂ , e ₆ , e ₇ , e ₈ , e ₉ , e ₁₀ , e ₁₁ e ₁₂ e ₁₃ , e ₁₄ , e ₁₅ , e ₁₆ , e ₁₈ , e ₁₉ , e ₂₁ , e ₂₂ e ₂₃ , e ₂₄ , e ₂₅ , e ₂₈ , e ₂₉ , e ₃₀ , e ₃₁ , e ₃₂ e ₃₃ , e ₃₄ , e ₃₅ , e ₃₆ ,e ₃₇ , e ₃₈ , e ₃₉ e ₄₀ , e ₄₁ , e ₄₂ , e ₄₃ , e ₄₄ , e ₄₅	e ₁₁	e ₁₁	XI
e 12	e1, e2, e6, e7, e8, e9, e10, e12, e13, e14 e15, e16, e18, e19, e20, e21, e22, e23 e24, e25, e28, e29, e30, e31, e32, e33 e34, e35, e36, e37, e38, e39, e40, e41, e42, e43, e44, e45	e7, e8, e10, e11, e12, e13, e14, e15, e16, e19, e20, e21, e22, e23,e27,e28, e29, e45	e7, e8, e10, e12, e13, e14, e15, e16, e19, e20, e21, e22, e23, e28, e29, e45	IX
e13	e ₁ ,e ₂ ,e ₆ , e ₇ , e ₈ , e ₉ , e ₁₀ , e ₁₂ , e ₁₃ , e ₁₄ e ₁₅ , e ₁₆ , e ₁₈ , e ₁₉ , e ₂₀ , e ₂₁ e ₂₂ e ₂₃ e ₂₄ , e ₂₅ ,e ₂₈ , e ₂₉ , e ₃₀ , e ₃₁ , e ₃₂ e ₃₃ e ₃₄ , e ₃₅ , e ₃₆ , e ₃₇ , e ₃₈ , e ₃₉ e ₄₀ e ₄₁ e ₄₂ , e ₄₃ , e ₄₄ , e ₄₅	e ₇ ,e ₈ , e ₉ , e ₁₀ , e ₁₁ , e ₁₂ , e ₁₃ , e ₁₄ , e ₁₅ e ₁₆ , e ₁₉ , e ₂₀ e ₂₁ , e ₂₂ , e ₂₃ , e ₂₇ , e ₂₈ , e ₂₉ , e ₃₀ , e ₃₅ , e ₃₇ , e ₃₈ , e ₄₃ , e ₄₅	e7,e8, e9, e10, e12, e13 e14, e15, e16, e19, e20, e21 e22, e23,e28, e29, e30, e35 e37, e38, e43, e45	іх

FIGURE 7. Level partitioning results of elements.



Element	Reachability set: R (Pi)	Antecedent set: A (Pi)	Intersection R (Pi)∩A (Pi)	Level
e 14	e ₁ , e ₂ , e ₆ , e ₇ , e ₈ , e ₉ , e ₁₀ , e ₁₂ , e ₁₃ e ₁₄ , e ₁₅ ,e ₁₆ , e ₁₈ , e ₁₉ ,e ₂₀ , e ₂₁ , e ₂₂ e ₂₃ ,e ₂₄ e ₂₅ , e ₂₉ e ₃₀ , e ₃₁ , e ₃₂ , e ₃₃ e ₃₄ , e ₃₅ e ₃₆ , e ₃₇ ,e ₃₈ , e ₃₉ , e ₄₀ e ₄₁ e ₄₂ , e ₄₃ , e ₄₄ , e ₄₅	e7, e8, e10, e11, e12, e13, e14, e15, e16, e19, e20, e21 e22, e23, e27,e28, e29, e45	e7, e8, e10,e12, e13, e14 e15, e16, e19, e20, e21, e22, e23, e29, e45	IX
e 15	e ₁ , e ₂ , e ₆ , e ₇ ,e ₈ ,e ₉ , e ₁₀ , e ₁₂ ,e ₁₃ e ₁₄ , e ₁₅ e ₁₆ , e ₂₈ , e ₁₉ , e ₂₁ ,e ₂₂ e ₂₃ e ₂₄ ,e ₂₅ , e ₂₈ , e ₂₉ , e ₃₀ , e ₃₁ ,e ₃₂ e ₃₃ e ₃₄ , e ₃₅ ,e ₃₆ , e ₃₇ , e ₃₈ , e ₃₉ , e ₄₀ e ₄₁ e ₄₂ , e ₄₃ , e ₄₄ , e ₄₅	e7, e8, e10, e11, e12, e13, e14, e15, e20, e21, e22, e23, e29	e7, e8, e10,e12, e13, e14 e15, e21, e22, e23, e29	x
e 16	e7, e8,e9, e10 e12, e13,e14,e16, e18 e19, e21, e22, e23, e24, e25, e30 e31 e32, e34, e35, e36, e37, e39, e40 e41 e42, e43, e44, e45	e7, e8, e10, e11, e12, e13 e14, e15, e16, e20, e21, e22, e23, e29	e_7 , e_8 , e_{10} , e_{12} , e_{13} , e_{14} , e_{16} , e_{21} , e_{22} , e_{23}	IX
e 17	e ₁₀ , e ₁₇ , e ₁₈ , e ₂₀ , e ₂₁ , e ₂₂ , e ₂₃ , e ₂₄ e ₂₅ ,e ₃₀ , e ₃₁ , e ₃₂ , e ₃₄ , e ₃₅ , e ₃₆ , e ₃₇ e ₃₈ , e ₃₉ , e ₄₀ , e ₄₁ , e ₄₂ , e ₄₃ , e ₄₄ , e ₄₅	e ₁₇	e ₁₇	XII
e18	e7, e8, e9, e10, e18, e21, e22, e23 e24, e25, e30, e31, e32, e34, e37, e38 e39,e40, e41,e42, e43, e44, e45	$e_{1},e_{3},e_{7}, e_{8}, e_{9}, e_{10}, e_{11} e_{12}, e_{13},e_{14} e_{15}, e_{16}, e_{17} e_{18}, e_{19} e_{20}, e_{21}, e_{22}, e_{23} e_{24}, e_{25} e_{27}, e_{28}, e_{29} e_{31}, e_{32} e_{34}, e_{35} e_{37}, e_{38}, e_{39}, e_{40}, e_{41}, e_{42}, e_{43}, e_{44}$	e7, e8, e9, e10, e18, e21, e22, e23, e24, e25,e31, e32, e34, e37, e38, e39,e40, e41,e42, e43, e44	VII
e 19	e ₁₀ ,e ₁₂ ,e ₁₃ ,e ₁₄ ,e ₁₈ ,e ₁₉ ,e ₂₁ ,e ₂₂ ,e ₂₃ ,e ₂₄ , e ₂₅ e ₃₀ , e ₃₁ ,e ₃₂ ,e ₃₄ ,e ₃₅ e ₃₆ ,e ₃₇ e ₃₉ , e ₄₀ , e ₄₁ , e ₄₂ , e ₄₃ , e ₄₄ e ₄₅	e ₁₀ , e ₁₁ , e ₁₂ , e ₁₃ , e ₁₄ , e ₁₅ e ₁₆ , e ₁₉ , e ₂₀ , e ₂₂ , e ₂₇ ,e ₂₈ , e ₂₉	e ₁₀ , e ₁₂ , e ₁₃ , e ₁₄ , e ₁₉ , e ₂₂	VIII
e20	e ₁ , e ₂ , e ₆ ,e ₇ ,e ₈ ,e ₉ ,e ₁₀ , e ₁₂ , e ₁₃ e ₁₄ , e ₁₅ e ₁₆ , e ₁₈ , e ₁₉ , e ₂₀ , e ₂₁ ,e ₂₂ e ₂₃ ,e ₂₄ e ₂₅ e ₂₉ ,e ₃₀ ,e ₃₁ ,e ₃₂ , e ₃₃ e ₃₄ , e ₃₅ , e ₃₆ , e ₃₇ , e ₃₈ , e ₃₉ , e ₄₀ , e ₄₁ e ₄₂ , e ₄₃ , e ₄₄ , e ₄₅	e ₃ , e ₇ e ₈ e ₉ e ₁₀ e ₁₂ e ₁₃ e ₁₄ , e ₁₇ , e ₂₀ , e ₂₁ , e ₂₄ , e ₃₇	e ₇ e ₈ e ₉ e ₁₀ e ₁₂ e ₁₃ e ₁₄ , e ₂₀ , e ₂₁ , e ₂₄ , e ₃₇	XI
e 21	e ₁ , e ₂ , e ₆ , e ₇ , e ₈ , e ₉ , e ₁₀ , e ₁₂ , e ₁₃ , e ₁₄ e ₁₅ , e ₁₆ , e ₁₈ , e ₂₀ , e ₂₁ , e ₂₂ , e ₂₃ e ₂₄ , e ₂₅ e ₂₈ , e ₂₉ , e ₃₀ , e ₃₁ , e ₃₂ e ₃₃ e ₃₄ , e ₃₅ , e ₃₇ e ₃₈ , e ₃₉ , e ₄₀ , e ₄₁ , e ₄₂ e ₄₃ ,e ₄₄ ,e ₄₅	e1,e3, e7, e8, e9, e10,e11 e12, e13 e14 e15, e16, e17 e18, e19, e20, e21 e22 ,e23 e24, e25, e27, e28, e29, e31, e32, e34, e35 e37, e38, e39 e42,e43, e44	e ₁ , e ₇ , e ₈ , e ₉ , e ₁₀ ,e ₁₂ e ₁₃ e ₁₄ , e ₁₅ e ₁₆ , e ₁₈ ,e ₂₀ e ₂₁ e ₂₂ ,e ₂₃ , e ₂₄ , e ₂₅ e ₂₈ , e ₂₉ e ₃₁ , e ₃₂ , e ₃₄ , e ₃₅ , e ₃₇ e ₃₈ e ₃₉ , e ₄₂ , e ₄₃ , e ₄₄	VII
e 22	e ₁ , e ₂ , e ₆ , e ₇ , e ₈ , e ₉ , e ₁₀ , e ₁₂ ,e ₁₃ e ₁₄ e ₁₅ , e ₁₆ e ₁₈ , e ₁₉ , e ₂₁ , e ₂₂ , e ₂₃ e ₂₄ , e ₂₅ e ₂₈ , e ₂₉ , e ₃₀ , e ₃₁ , e ₃₂ ,e ₃₃ e ₃₄ , e ₃₆ , e ₃₇ e ₃₈ , e ₃₉ , e ₄₀ e ₄₁ , e ₄₂ e ₄₃ , e ₄₄ , e ₄₅	e ₁ , e ₃ , e ₇ , e ₈ ,e ₉ , e ₁₀ ,e ₁₁ ,e ₁₂ , e ₁₃ , e ₁₄ e ₁₅ , e ₁₆ , e ₁₇ e ₁₈ , e ₁₉ , e ₂₀ , e ₂₁ ,e ₂₂ , e ₂₃ e ₂₄ , e ₂₅ , e ₂₇ e ₂₈ , e ₂₉ , e ₃₀ e ₃₁ , e ₃₂ , e ₃₄ e ₃₅ , e ₃₇ , e ₃₈ e ₃₉ , e ₄₀ , e ₄₁ , e ₄₂ , e ₄₃ , e ₄₄	e ₁ , e ₇ , e ₈ , e ₉ , e ₁₀ , e ₁₂ e ₁₃ , e ₁₄ ,e ₁₅ e ₁₆ , e ₁₈ , e ₁₉ e ₂₁ ,e ₂₂ , e ₂₃ , e ₂₄ , e ₂₅ , e ₂₈ e ₂₉ , e ₃₀ , e ₃₁ e ₃₂ e ₃₄ , e ₃₇ e ₃₈ e ₃₉ e ₄₀ e ₄₁ e ₄₂ e ₄₃ e ₄₄	п
e 23	e ₁ , e ₂ , e ₆ , e ₇ , e ₈ , e ₉ , e ₁₀ ,e ₁₂ , e ₁₃ e ₁₄ , e ₁₅ , e ₁₆ , e ₁₈ , e ₂₁ ,e ₂₂ ,e ₂₃ ,e ₂₄ e ₂₅ ,e ₂₈ ,e ₂₉ ,e ₃₀ , e ₃₁ , e ₃₂ , e ₃₃ , e ₃₄ e ₃₅ , e ₃₇ , e ₃₈ e ₃₉ ,e ₄₀ , e ₄₁ , e ₄₂ , e ₄₃ , e ₄₄ , e ₄₅	e ₁ , e ₃ , e ₇ , e ₈ , e ₉ , e ₁₀ , e ₁₁ e ₁₂ e ₁₃ ,e ₁₄ e ₁₅ , e ₁₆ , e ₁₇ , e ₁₈ , e ₁₉ e ₂₀ , e ₂₁ , e ₂₂ , e ₂₃ e ₂₄ , e ₂₅ , e ₂₇ , e ₂₈ , e ₂₉ , e ₃₁ , e ₃₂ , e ₃₄ , e ₃₅ e ₃₈ , e ₄₂ , e ₄₃ , e ₄₄	e ₁ , e ₇ , e ₈ , e ₉ , e ₁₀ , e ₁₂ e ₁₃ , e ₁₄ , e ₁₅ e ₁₆ , e ₁₈ , e ₂₁ ,e ₂₂ , e ₂₃ , e ₂₄ , e ₂₅ , e ₂₈ e ₂₉ , e ₃₁ , e ₃₂ , e ₃₄ , e ₃₅ , e ₃₈ , e ₄₂ , e ₄₃ , e ₄₄	VII

FIGURE 7. (Continued.) Level partitioning results of elements.

Element	Reachability set: R (Pi)	Antecedent set: A (Pi)	Intersection R (Pi)∩A (Pi)	Level
e 24	e7, e8, e9, e10, e18, e20, e21,e22, e23 e24 e25, e30, e31, e32, e34, e35, e37, e38 e39, e40, e41, eS42, e43, e44, e45	e ₁ , e ₃ , e ₇ , e ₈ ,e ₉ , e ₁₀ , e ₁₁ , e ₁₂ , e ₁₃ e ₁₄ e ₁₅ e ₁₆ e ₁₇ e ₁₈ e ₁₉ e ₂₀ e ₂₁ , e ₂₂ e ₂₃ , e ₂₄ ,e ₂₅ , e ₂₇ e ₂₈ e ₂₉ e ₃₀ ,e ₃₁ e ₃₂ e ₃₄ , e ₃₇ , e ₄₃ , e ₄₄	e7, e8, e9, e10, e18, e20, e21 e22 e23, e24, e25, e30, e31, e32, e34 e37, e43, e44	v
e 25	e ₇ , e ₈ , e ₉ , e ₁₀ , e ₁₈ , e ₂₁ , e ₂₂ , e ₂₃ , e ₂₄ , e ₂₅ , e ₃₀ , e ₃₁ , e ₃₂ , e ₃₄ , e ₃₇ , e ₃₈ , e ₃₉ , e ₄₀ , e ₄₁ , e ₄₂ , e ₄₃ , e ₄₄ , e ₄₅	e ₁ ,e ₃ , e ₇ , e ₈ , e ₉ e ₁₀ , e ₁₁ , e ₁₂ , e ₁₃ e ₁₄ , e ₁₅ ,e ₁₆ ,e ₁₇ ,e ₁₈ ,e ₁₉ ,e ₂₀ , e ₂₁ , e ₂₂ e ₂₃ ,e ₂₄ , e ₂₅ , e ₂₇ ,e ₂₈ e ₂₉ e ₃₀ e ₃₁ e ₃₂ e ₃₄ , e ₃₅ , e ₃₇ e ₃₈ e ₃₉ e ₄₂ e ₄₃ e ₄₄	e7, e8,e9,e10,e18,e21,e22, e23, e24, e25, e30, e31, e32, e34, e37, e38, e39, e42, e43, e44	ш
e ₂₆	e7, e8, e26, e27, e28, e29	$e_7, e_8, e_{26}, e_{27}, e_{28}, e_{29}$	e7, e8, e26, e27, e28, e29	I
e27	e7, e8, e24, e25, e26, e27, e28, e29, e30 e31, e32,e34, e35, e36, e37, e39, e40, e41 e42, e43, e44, e45	e7, e8, e26, e27, e28, e29	e7, e8, e26, e27, e28, e29	VII
e 28	e7, e8, e10, e12, e13, e21, e22, e23, e24, e25,e26, e27,e28, e29,e30, e31, e32, e34, e35, e36, e37, e39, e40, e41, e42, e43,e44 e45	e7, e8, e10, e11 e12, e13, e15, e21, e22, e23, e26, e27, e28, e29	e7, e8, e10,e12, e13, e21, e22, e23, e26, e27, e28,e29	VII
e 29	e7, e8, e10, e12,S e13, e14, e15, e21, e22, e23, e24, e25,e26, e27,e28, e29, e31,e32, e33,e34, e35, e36, e37, e38, e39, e40, e41, e42, e43, e44, e45	e ₇ , e ₈ , e ₁₀ , e ₁₁ , e ₁₂ , e ₁₃ , e ₁₄ e ₁₅ e ₂₀ , e ₂₁ e ₂₂ e ₂₃ e ₂₆ e ₂₇ e ₂₈ e ₂₉	e7, e8, e10, e12, e13, e14 e15, e21, e22, e23, e26 e27 e28 e29	VI
e 30	e7, e8, e13, e22, e24, e25,e30, e31, e32, e34, e35, e37, e38,e39, e40,e41, e42, e43, e44, e45	e ₁ , e ₃ , e ₇ ,e ₈ , e ₉ , e ₁₀ ,e ₁₁ ,e ₁₂ , e ₁₃ ,e ₁₄ e ₁₅ , e ₁₆ ,e ₁₇ , e ₁₈ , e ₁₉ , e ₂₀ , e ₂₁ e ₂₂ e ₂₃ , e ₂₄ , e ₂₅ e ₂₇ , e ₂₈ , e ₃₀ , e ₃₁ , e ₃₄ , e ₃₇ , e ₃₈ , e ₃₉ , e ₄₀ ,e ₄₁ , e ₄₂ , e ₄₃ , e ₄₄	e7, e8, e13, e22, e24, e25 e30, e31 e34, e37,e38,e39, e40,e41, e42, e43, e44	VI
e 31	e7, e8, e9, e10,e18,e21, e22, e23, e24 e25, e30,e31, e32, e34, e35, e37, e38 e39, e40, e41,e42, e43, e44,e45	e ₁ , e ₂ , e ₇ , e ₈ , e ₉ , e ₁₀ , e ₁₁ , e ₁₂ , e ₁₃ , e ₁₄ e ₁₅ , e ₁₆ , e ₁₇ ,e ₁₈ , e ₁₉ e ₂₀ e ₂₁ e ₂₂ , e ₂₃ , e ₂₄ , e ₂₅ , e ₂₇ ,e ₂₈ , e ₂₉ , e ₃₀ , e ₃₁ , e ₃₂ , e ₃₄ , e ₃₅ e ₃₇ e ₃₈ , e ₃₉ e ₄₀ , e ₄₁ , e ₄₂ , e ₄₃ , e ₄₄	e7, e8, e9, e10, e18, e21, e22, e23, e24, e25, e30,e31, e32, e34, e35, e37,e38, e39, e40, e41, e42, e43, e44	п
e ₃₂	e7, e8, e9,e10, e18, e21, e22, e23, e24, e25, e31, e32, e34, e35,e37, e38, e39, e40, e41, e42, e43, e44, e45	e ₃ , e ₇ , e ₈ , e ₉ , e ₁₀ , e ₁₁ , e ₁₂ , e ₁₃ , e ₁₄ , e ₁₅ e ₁₆ e ₁₇ e ₁₈ e ₁₉ e ₂₀ e ₂₁ , e ₂₂ e ₂₃ e ₂₄ e ₂₅ e ₂₇ e ₂₈ e ₂₉ e ₃₀ , e ₃₁ e ₃₂ e ₃₄ , e ₃₇ , e ₃₈ , e ₄₀ , e ₄₁ , e ₄₂ , e ₄₃ , e ₄₄	e7, e8, e9, e10,e18, e21, e22, e23, e24,e25, e31, e32, e34, e37, e38, e40,e41,e42, e43, e44	v
e 33	e ₃₃	e ₇ , e ₈ , e ₁₀ ,e ₁₁ ,e ₁₂ , e ₁₃ , e ₁₄ e ₁₅ , e ₂₀ e ₂₁ , e ₂₂ , e ₂₃ , e ₂₉ , e ₃₃	e ₃₃	I
e 34	e ₇ , e ₈ , e ₉ , e ₁₀ , e ₁₈ , e ₂₁ ,e ₂₂ ,e ₂₃ , e ₂₄ , e ₂₅ , e ₃₀ , e ₃₁ ,e ₃₂ , e ₃₄ , e ₃₇ , e ₃₈ ,e ₃₉ , e ₄₀ , e ₄₁ , e ₄₂ , e ₄₃ , e ₄₄ , e ₄₅	e ₁ , e ₃ , e ₇ , e ₈ , e ₉ , e ₁₀ , e ₁₁ , e ₁₂ ,e ₁₃ , e ₁₄ , e ₁₅ , e ₁₆ e ₁₇ e ₁₈ e ₁₉ ,e ₂₀ e ₂₁ e ₂₂ e ₂₃ e ₂₄ e ₂₅ e ₂₇ ,e ₂₈ e ₂₉ e ₃₀ e ₃₁ e ₃₂ , e ₃₄ , e ₃₇ e ₃₈ e ₃₉ e ₄₀ e ₄₁ e ₄₂ e ₄₃ e ₄₄	e7, e8, e9, e10, e18, e21, e22, e23, e24, e25, e30, e31, e32, e34, e37, e38, e39, e40, e41, e42, e43, e44	п

FIGURE 7. (Co	ontinued.) Level	partitioning	results of	f elements.
---------------	------------------	--------------	------------	-------------



Element	Reachability set: R (Pi)	Antecedent set: A (Pi)	Intersection R (Pi)∩A (Pi)	Level
e 35	e ₁₃ , e ₂₁ , e ₂₂ , e ₂₃ , e ₂₅ , e ₃₁ , e ₃₅ , e ₃₇ , e ₃₈ , e ₃₉ , e ₄₀ , e ₄₁ , e ₄₃ ,e ₄₄ , e ₄₅	e _{1 e⁷} , e ₈ e ₉ e ₁₀ e ₁₁ e ₁₂ e ₁₃ e ₁₄ , e ₁₅ , e ₁₆ e ₁₇ e ₁₉ e ₂₀ e ₂₁ e ₂₃ , e ₂₄ , e ₂₇ e ₂₈ e ₂₉ e ₃₀ e ₃₁ e ₃₂ e ₃₅	e ₁₃ , e ₂₁ , e ₂₃ , e ₃₁ , e ₃₅	IV
e 36	e ₃₆	$e_{10} e_{11} e_{12} e_{13} e_{14} e_{15} e_{16} e_{17} e_{19}$ $e_{20}, e_{22}, e_{27} e_{28} e_{29}, e_{36}$	e ₃₆	I
e 37	e7, e8, e9, e13, e18, e20, e21, e22,e24 e25, e30, e31, e32, e34, e37, e38 e39 e40, e41, e42, e43, e44,e45	e ₃ , e ₇ e ₈ e ₉ e ₁₀ e ₁₁ e ₁₂ e ₁₃ e ₁₄ e ₁₅ e ₁₆ e ₁₇ e ₁₈ e ₁₉ e ₂₀ e ₂₁ e ₂₂ e ₂₃ e ₂₄ e ₂₅ , e ₂₇ , e ₂₈ e ₂₉ e ₃₀ e ₃₁ e ₃₂ , e ₃₄ , e ₃₅ , e ₃₇ e ₃₈ e ₃₉ e ₄₀ e ₄₁ e ₄₂ e ₄₃ e ₄₄	e7, e8, e9, e13,e18, e20, e21, e22, e24, e25, e30, e31, e32, e34, e37, e38, e39, e40, e41, e42, e43, e44	п
e38	e ₁₃ , e ₁₈ , e ₂₁ , e ₂₂ ,e ₂₃ , e ₂₅ , e ₃₀ , e ₃₁ , e ₃₂ , e ₃₄ , e ₃₇ , e ₃₈ , e ₃₉ , e ₄₀ , e ₄₁ , e ₄₂ , e ₄₃ , e ₄₄ , e ₄₅	e ₃ , e ₇ e ₈ e ₉ e ₁₀ e ₁₁ e ₁₂ e ₁₃ e ₁₄ e ₁₅ , e ₁₇ , e ₁₈ , e ₂₀ e ₂₁ e ₂₂ e ₂₃ e ₂₄ e ₂₅ e ₂₉ e ₃₀ e ₃₁ e ₃₂ , e ₃₄ , e ₃₅ , e ₃₇ e ₃₈ e ₃₉ e ₄₀ e ₄₁ e ₄₂ e ₄₃ e ₄₄	e ₁₃ , e ₁₈ , e ₂₁ , e ₂₂ , e ₂₃ , e ₂₅ , e ₃₀ , e ₃₁ , e ₃₂ , e ₃₄ , e ₃₇ , e ₃₈ , e ₃₉ , e ₄₀ , e ₄₁ , e ₄₂ , e ₄₃ , e ₄₄	п
e 39	e ₁₈ , e ₂₁ ,e ₂₂ , e ₂₅ , e ₃₀ , e ₃₁ , e ₃₄ , e ₃₇ , e ₃₈ , e ₃₉ , e ₄₀ , e ₄₁ , e ₄₂ , e ₄₃ ,e ₄₄ , e ₄₅	$\begin{array}{c} e_3, e_7, e_8 e_9 e_{10} e_{11} e_{12} e_{13} e_{14} e_{15} \\ e_{16} e_{17} e_{18} e_{19} e_{20} e_{21} e_{22} e_{23} e_{24} \\ e_{25} e_{27} e_{28} e_{29} e_{30} e_{31} e_{32} e_{34} e_{35} , \\ e_{37} e_{38} e_{39} e_{40} e_{41} e_{42} e_{43} e_{44} \end{array}$	e ₁₈ , e ₂₁ , e ₂₂ , e ₂₅ , e ₃₀ , e ₃₁ , e ₃₄ , e ₃₇ , e ₃₈ ,e ₃₉ , e ₄₀ , e ₄₁ , e ₄₂ , e ₄₃ , e ₄₄	Ш
e 40	e ₁₈ , e ₂₂ , e ₃₀ , e ₃₁ , eS ₃₂ , e ₃₄ , e ₃₇ , e ₃₈ , e ₃₉ , e ₄₀ , e ₄₁ , e ₄₂ , e ₄₃ , e ₄₄ , e ₄₅	e ₁ , e ₃ , e ₇ e ₈ e ₉ e ₁₀ e ₁₁ e ₁₂ e ₁₃ e ₁₄ e ₁₅ e ₁₆ e ₁₇ e ₁₈ e ₁₉ e ₂₀ e ₂₁ e ₂₂ ,e ₂₃ , e ₂₄ , e ₂₅ , e ₂₇ e ₂₈ e ₂₉ e ₃₀ e ₃₁ e ₃₂ , e ₃₄ e ₃₅ , e ₃₇ e ₃₈ e ₃₉ e ₄₀ e ₄₁ e ₄₂ e ₄₃ e ₄₄	e ₁₈ , e ₂₂ ,e ₃₀ , e ₃₁ , e ₃₂ , e ₃₄ , e ₃₇ ,e ₃₈ , e ₃₉ , e ₄₀ , e ₄₁ , e ₄₂ , e ₄₃ ,e ₄₄	п
e 41	e ₁₈ , e ₂₂ , e ₃₀ , e ₃₁ , eS ₃₂ , e ₃₄ , e ₃₇ , e ₃₈ , e ₃₉ , e ₄₀ , e ₄₁ , e ₄₂ , e ₄₃ , e ₄₄ , e ₄₅	e ₃ ,e ₇ e ₈ e ₉ e ₁₀ e ₁₁ e ₁₂ e ₁₃ e ₁₄ e ₁₅ e ₁₆ e ₁₇ e ₁₈ e ₁₉ e ₂₀ e ₂₁ e ₂₂ e ₂₃ ,e ₂₄ , e ₂₅ , e ₂₇ e ₂₈ e ₂₉ e ₃₀ e ₃₁ e ₃₂ , e ₃₄ , e ₃₅ , e ₃₇ e ₃₈ e ₃₉ ,e ₄₀ e ₄₁ e ₄₂ e ₄₃ e ₄₄	e ₁₈ , e ₂₂ , e ₃₀ , e ₃₁ , eS ₃₂ , e ₃₄ , e ₃₇ , e ₃₈ , e ₃₉ , e ₄₀ , e ₄₁ , e ₄₂ , e ₄₃ , e ₄₄ , e ₄₅	п
e 42	e7, e8, e9, e18, e21, e22, e23, e25, e30, e31,e32, e34,e37, e38, e39, e40, e41, e42, e43, e44,e45	$e_{1},e_{3},e_{7},e_{8},e_{9},e_{10},e_{11},e_{12},e_{13},e_{14},e_{15}$ $e_{16},e_{17},e_{18},e_{19},e_{20},e_{21},e_{22},e_{23},e_{24}$ $e_{25},e_{27},e_{28},e_{29},e_{30},e_{31},e_{32},e_{34},e_{37},e_{38}$ $e_{39},e_{40},e_{41},e_{42},e_{43},e_{44},e_{45}$	e7, e8, e9, e18, e21, e22, e23, e25, e30, e31, e32, e34, e37, e38, e39, e40, e41,e42, e43, e44, e45	I
e 43	e7, e8, e9, e10, e13, e18, e21, e22, e23, e24, e25, e30, e31, e32, e34, e37, e38, e39, e40, e41, e43, e44, e45	e1, e3 e7 e8 e9 e10 e11 e12 e13 e14 e15 e16 e17 e18 e19 e20 e21 e22 e23 e24 e25 e27 e28 e29 e30 e31 e32 e34 e35 e37 e38 e39 e40 e41 e42 e43 e44	e7, e8, e9, e10, e13, e18, e21, e22, e23, e24, e25, e30, e31, e32, e34, e37, e38, e39, e40, e41, e43, e44, e45	I
e 44	e7, e8, e9, e10,e18, e21, e22, e23, e24, e25 e30,e31, e32, e34, e37, e38, e39, e40, e41, e42, e43, e44, e45	e_1 , $e_3 e_7 e_8 e_9 e_{10} e_{11} e_{12} e_{13} e_{14}$ $e_{15} e_{16} e_{17} e_{18} e_{19} e_{20} e_{21} e_{22} e_{23} e_{24}$ $e_{25} e_{27} e_{28} e_{29} e_{30} e_{31} e_{32} e_{34} e_{35} e_{37}$ $e_{38} e_{39} e_{40} e_{41} e_{42} e_{43} e_{44} e_{45}$	e7, e8, e9, e10,e18, e21, e22, e23, e24, e25 e30,e31, e32, e34, e37, e38, e39, e40, e41, e42, e43, e44, e45	I
e 45	e7, e8, e9, e10, e12, e13, e14, e42, e43, e44, e45	e ₁ ,e ₃ ,e ₇ , e ₈ ,e ₉ , e ₁₀ , e ₁₁ , e ₁₂ , e ₁₃ , e ₁₄ , e ₁₅ ,s e ₁₆ , e ₁₇ , e ₁₈ , e ₁₉ , e ₂₀ ,e ₂₁ e ₂₂ , e ₂₃ , e ₂₄ , e ₂₅ , e ₂₇ , e ₂₈ , e ₂₉ , e ₃₀ , e ₃₁ , e ₃₂ , e ₃₄ , e ₃₅ , e ₃₇ , e ₃₈ , e ₃₉ , e ₄₀ , e ₄₁ , e ₄₂ , e ₄₃ e ₄₄ , e ₄₅	e7, e8, e9, e10, e12, e13, e14, e42, e43, e44, e45	I

FIGURE 7. (Continued.) Level partitioning results of elements.



FIGURE 8. ISM hierarchical framework.

driver and dependence power are included in the first cluster (autonomous) of the matrix, such elements are relatively detached and have a few links. Elements that are located within this cluster are: e_1 (future expansions), e_2 (capturing dependency relationships among activities), e_4 (project environment), e_5 (market strategy), e_6 (deployment strategy), e_{19} (building trust), e_{26} (capturing corporate knowledge), e_{33} (operational concepts), e_{36} (managing politics). In the second cluster (dependent), elements having weak driving, but strong dependence power are included. This cluster includes the elements such as e_{45} (client acceptance), e_{42} (setting milestones), e_{41} (track progress), e_{40} (control and information

E1 E2 E3 E4 E5 E4 E5 E4 E5 E6 E7 E8 E9 E10 E11 E12 E18 E14 E15 E16 E15 E16 E17 E18 E19 E19 E20 E21 E21 E21 E21 E21 E24 E25 E24 E25 E26 E27 E28 E29 E30 E31 E32 E34 E35 E36 E37 E38 E39 E40 E41 E42 E43 E44	e44 e45
e 100 100 100 100 100 100 100 100 100 10	1.00 1.00
e 100 100 100 100 100 100 100 100 100 10	1.00 1.00
es 4:14 1.00 0.00 4.00 3.71 4.29 4.24 1.71 1.14 4.29 1.29 1.29 1.29 1.29 1.20 1.00 1.00 1.00 1.00 1.00 1.00 1.00	3.43 3.00
e 100 100 100 100 100 100 100 100 100 10	1.00 1.00
es 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.0	1.00 1.00
e 100 100 100 100 100 100 100 100 100 10	1.00 1.00
ex 4.29 4.71 1.29 4.43 4.43 2.71 0.00 3.00 4.43 1.29 1.14 2.14 3.29 1.57 1.00 4.00 1.00 4.29 1.00 3.00 4.71 4.71 5.00 3.50 4.57 5.00 3.50 4.57 5.50 3.55 1.54 1.50 1.00 1.00 1.00 1.00 1.00 1.00 2.11 1.55 5.50 3.55 1.55 5.50 3.55 5.50 5.55 5.5	3.86 4.86
es 3.00 3.00 1.14 3.43 3.00 2.00 4.57 0.00 3.29 1.00 1.29 2.00 1.29 2.00 1.29 2.00 1.20 2.86 1.00 3.57 1.00 4.00 4.43 4.00 5.00 2.57 1.14 1.00 1.00 1.00 1.00 2.29 1.57 2.43 3.71 1.00 2.29 1.14 1.00 1.00 1.00 1.00 1.00 1.00 4.29 5.00 3.86 1.00 4.00 4.00 4.00 4.00 4.00 4.00 4.00	3.86 5.00
ex 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.0	2.29 1.14
eta 1.00 3.86 1.29 1.14 2.29 2.43 3.29 3.29 4.29 0.00 1.00 1.14 1.29 1.43 1.00 1.00 2.14 1.43 1.86 3.86 3.86 3.86 3.86 3.86 3.86 3.86 3	3.29 1.57
e1 2:14 3:86 4.71 3:33 3.71 3:29 4.71 4.29 4.71 3:29 4.71 3.29 4.71 3.29 4.71 3.29 4.71 3.29 3.00 1.00 1.00 1.00 1.00 1.00 1.00 1.00	4.57 4.57
ers 1.00 1.00 1.05 1.00 1.14 2.14 4.57 4.57 4.57 4.57 1.29 1.00 0.00 1.00 1.04 1.00 1.43 1.14 1.00 1.43 2.25 2.57 2.71 1.00 2.00 1.00 1.00 1.00 1.00 1.01 4.13 2.9 1.14 1.00 1.00 2.00 1.00 1.00 4.14 4.33 4.00 1.00 1.01 4.01 4.00 1.00 1.00 4.14 4.33 4.00 4.00 4.00 4.00 4.00 4.00 4.0	4.00 4.57
ers 1.00 1.00 2.00 1.00 2.00 1.00 1.00 1.00	4.00 4.43
eta [2:29] 1.86 1.14 1.00 2.13 2.00 4.57 4.43 4.29 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.0	4.71 5.00
ets 1.29 1.00 1.29 1.00 2.8 1.14 3.7 3.7 13.14 3.14 2.86 3.84 3.14 2.86 3.84 2.00 4.14 2.00 2.14 2.30 3.86 3.57 1.14 1.24 1.71 1.71 1.71 1.71 1.71 1.71 3.43 2.71 1.71 1.71 1.71 1.71 1.71 1.71 1.71	1.29 1.43
eta [2:29 2:57 3:29 2:57 1:86 1:00 2:00 1:86 4:29 4:00 4:00 2:00 1:86 3:57 2:29 1:14 1:20 2:00 1:29 1:14 1:29 2:00 2:20 2:27 3:29 2:71 1:14 1:29 1:10 2:34 3:34 3:34 3:34 1:57 1:86 2:86 1:14 2:14 1:14 2:29 3:57 3:29 2:71 1:14 1:29 1:00 2:00 1:20 1:20 1:20 1:20 1:20 1:20	2.71 1.86
er 1.27 1.20 1.00 1.01 1.01 1.01 1.02 1.22 2.22 1.00 1.00	2.29 1.43
eta 1.00 1.14 1.43 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.43 2.23 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.0	2.00 1.57
e1 100 1.00 1.00 1.00 1.00 1.00 1.00 1.0	1.00 1.29
exa 3.57 1.00 3.85 3.57 4.00 3.57 1.00 1.02 1.14 1.28 3.57 1.57 1.14 1.29 1.00 2 4.3 1.00 1.00 1.00 3.85 3.86 4.00 4.14 4.00 1.00 1.00 1.00 1.00 1.00	1.14 1.14
e ¹ 1.00 1.14 1.43 1.00 4.14 2.86 4.71 4.14 3.71 3.43 1.142 1.44 1.86 1.43 1.00 4.14 1.00 3.43 0.00 4.14 4.00 2.14 3.86 1.00 1.00 1.00 1.00 3.86 2.29 1.00 3.71 1.57 1.00 1.43 1.14 3.43 1.43 2.00 4.03 2.29 4.43	4.43 5.00
ez 1100 1.71 1.29 1.29 1.29 1.29 1.29 1.29 1.29 1.2	2.57 1.86
era 5:20 1:29 3:14 4:00 4:14 4:43 4:57 4:00 1:57 3:86 2:29 1:18 2:57 1:29 1:00 3:14 1:00 1:71 1:00 3:57 5:00 5:00 5:00 5:00 5:00 5:00 4:29 4:86 1:14 1:14 2:43 3:14 4:43 1:43 1:40 1:29 1:57 3:57 2:57 2:57 2:43 4:43 4:74 4:00 1:51 3:57 3:57 2:57 2:57 2:57 2:57 2:57 2:57 2:57 2	4.00 4.71
exa 3.71 1.00 2.29 2.14 1.00 1.86 1.43 1.43 1.48 1.00 1.00 1.00 1.00 1.00 1.00 1.00 4.00 1.00 4.29 1.29 4.43 5.00 0.00 4.71 1.00 1.00 1.14 1.14 2.11 2.9 2.29 1.00 1.00 1.00 1.00 1.01 1.41 1.29 1.00 1.14 1.29 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.0	1.14 1.71
ex 1 14 1.00 1.00 2.29 1.00 1.27 1.14 2.86 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.0	2.00 4.57
ex 1.14 1.00 1.00 1.00 1.00 1.00 1.00 1.00	1.00 1.00
ex7 1.29 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.0	1.00 1.00
ex 1.57 2.00 1.57 1.57 3.57 3.57 2.14 2.86 2.00 1.14 1.14 1.14 1.14 1.14 1.10 1.00 1.0	3.29 1.86
es 177 100 143 233 214 257 400 243 100 1.00 1.00 1.00 1.00 1.00 1.00 1.10 1.00 1.	1.00 1.00
eval 1.14 1.43 2.57 2.14 3.00 2.86 4.57 4.29 2.86 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.0	1.00 1.00
ex 1 100 1.00 1.00 1.00 1.00 1.00 1.00 1.	1.00 1.00
ex 100 100 100 100 100 100 100 100 100 10	1.00 1.14
es 1:00 1:00 1:00 1:00 1:00 1:00 1:00 1:0	1.00 1.00
ea 1.00 1.14 1.57 1.86 1.86 1.86 1.86 1.87 3.71 3.57 1.86 1.29 1.00 1.101 1.41 0.00 1.00 1.00 1.00 1.0	3.57 2.71
ex 1.57 1.00 1.57 1.14 1.14 1.14 1.14 1.00 1.00 1.00 1.00	1.00 2.00
exe 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.0	1.00 1.00
ex7 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.0	1.00 1.14
ex 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.0	1.00 1.00
ex 1.00 1.10 1.00 1.00 1.00 1.00 1.00 1.0	1.00 1.00
evel 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.0	1.00 1.00
e1 100 1.00 1.00 1.00 1.00 1.00 1.00 1.0	2.43 1.00
ex 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.0	1.86 2.14
era 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.0	1.00 4.86
en 1.00 1.00 1.00 1.00 1.00 1.00 1.00 1.0	0.00 5.00
es 1.14 1.00 1.00 1.00 1.00 1.00 1.00 1.00	1.00 0.00

ł			ļ				ĺ	ļ	Ì	Ì	Ì	İ	İ	Ì	Ì	Ì	Ì		ľ	ŀ				-				ļ	l	ļ	ļ	ļ				Ī	Ì	ľ	ł	ļ		
్	67 67	3 C4	e5	e,6	e,	es	e9	e ₁₀	e II	312 e	313 e	314 e	315	9 ¹ 9	e17 6	18 e	19 e	30	e2	2 C23	C 24	e25	e ₂₆	e27	e ₂₈	e ₂₉	e ₃₀	e ₃₁	e32	e33	e34	e35	e ₃₆	e37	e ₃₈	e ₃₉	e40	41 e.	e 43	e.44	e45	
-	0000 0.0061 0	1.0061 0.00	161 0.006	51 0.0061	1 0.0061	1 0.0061	0.0061	0.0061	0.0061	0.0061 (0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 0	0 1900	0061 0.0	0.061 0.0	0.0	0.00	0.00	61 0.00	51 0.006	61 0.006	1 0.006	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 0	0061 0.0	361 0.00	61 0.006	10
	0 0000 0 1200	000	200.0	1700 0 13	17000	7000	17000	1000	17000	1,000	1,000,0	1,000	17000	17000	17000	10000	1000	1200	0	1700	1700	00 170	100	000	000	000	2000	2000	17000	12000	17000	1200.0	17000	1200.0	1700 0	17000	17000	0 1700 0	00 1000	000	0000	1 5
<u></u>	1 0000.0 1000	10.0 1000.0)00'0 IQ	900'0 10	90000	10000	1900.0	1900.0	1000.0	. 1900.0	1900.0	1900.0	1900'0	1900'0	1900.0	1900'0	1900'0	0 1900.0	0 1900.	1000	70 1900	D'D 190	10'n 190	0070 190	0070	0.00	/00/0 10	0000 6	1900.0	1900.0	1900.0	1900.0	1900.0	1900.0	1900:0	1900'0	1900.0	0 1900'0	170 I900	10:0 100	900.0	-
ي د	0254 0.0061 (0.0000 0.02	246 0.022	28 0.026.	3 0.027.	2 0.0105	0.0070	0.0263	0.0193	0.0263 (0.0079 (0.0123	0.0061	0.0096	0.0061	0.0061	0.0061	0.0175 0	.0272 0.	0272 0.0	0272 0.0	281 0.0	289 0.00	0.00	61 0.00	51 0.006	61 0.029	8 0.0290	0.0307	0.0070	0.0132	0.0061	0.0070	0.0088	0.0105	0.0070	0.0096	0.0061 0	0281 0.0	281 0.0	11 0.018	쳤는
*	0061 0.0061 (0.0061 0.00	000 0:001	700.0	9 0.006.	1 0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 (0.0061 t	0.0061	0.0088	0.0061	0.0070	0.0061	0.0061 0	0.0061 0.	0061 0.0	0.0 0.0	070 0.0	079 0.00	0.00	61 0.00	51 0.006	61 0.006	0.006	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 0	0061 0.0	061 0.00	61 0.006	10
35	0061 0.0061 0	0.0061 0.00	0.000	00 0.006	0.006	1 0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 (0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 0	0079 0.	0061 0.0	0070 0.0	061 0.0	0.00	0.00	61 0.00	51 0.006	61 0.006	0.006	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 0	0061 0.0	961 0.00	61 0.006	10
_ع و (0061 0.0061 0	0.0061 0.00	0.006	51 0.0000	0 0.006	1 0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 (0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 0	0 1900	0061 0.0	0.0 0.0	0.0	0.0	0.00	61 0.00	51 0.006	61 0.006	0.006	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 0	0061 0.0	961 0.00	61 0.006	15
34	0263 0.0289 (0.0079 0.02	272 0.025	72 0.016.	7 0.000	0.0184	0.0272	0.0079	0.0070	0.0132	0.0202	0.0096	0.0061	0.0246	0.0061	0.0263	0.0061	0.0184 0	0289 0.	0289 0.0	0307 0.0	281 0.0	132 0.00	0.00	61 0.00	51 0.006	61 0.014	0 0.007	0.0237	0.0114	0.0289	0.0228	0.0061	0.0246	0.0167	0.0114	0.0096	0.0158 0	0281 0.0	307 0.02	37 0.029	~~~
) 8	0184 0.0184 (0.0070 0.02	211 0.018	34 0.012.	3 0.028	1 0.0000	0.0202	0.0061	0.0061	0.0079 (0.0123 (0.0070	0.0061	0.0175	0.0061	0.0219 (0.0061	0.0246 0	0.0272 0.	0246 0.0	0307 0.0	158 0.0	070 0.00	0.00	61 0.00	51 0.006	61 0.012	3 0.006	0.0140	0.0096	0.0149	0.0228	0.0061	0.0140	0.0070	0.0061	0.0061	0.0061 0	0263 0.0	307 0.02	37 0.030	5
) (0061 0.0061 0	0.0061 0.00	161 0.006	51 0.006	1 0.008	8 0.0088	0.0000	0.0061	0.0061	0.0079 (0.0070	0.0061 (0.0061	0.0070	0.0061	0.0061	0.0061	0.0088 0	0237 0.	0175 0.0	0237 0.0	061 0.0	0.0	0.00	61 0.00	51 0.006	61 0.007	9 0.018	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0070	0.0061	0.0061 0	0132 0.0	140 0.01	40 0.007	2
) ⁰¹ ;	0061 0.0237 0	0.0079 0.00	70 0.014	40 0.0145	9 0.0202	2 0.0202	0.0263	0.0000	0.0061	0.0070	0.0079 (0.0088 (0.0061	0.0061	0.0132	0.0088 (0.0114 0	0.0237 0	0237 0.	0184 0.0	0202 0.0	228 0.0	202 0.00	0.00	61 0.00	51 0.006	61 0.020	2 0.022	0.0061	0.0061	0.0088	0.0079	0.0132	0.0140	0.0123	0.0202	0.0088	0.0132 0	0254 0.0	228 0.02	02 0.005	- 2
1	0132 0.0237 (0.0289 0.02	111 0.022	28 0.0202	2 0.0285	9 0.0263	0.0272	0.0272	0.0000	0.0281	0.0246 (0.0237 (0.0193	0.0246	0.0211	0.0298	0.0254 (0.0193 0	0.0272 0.	0289 0.0	0289 0.0	202 0.0	184 0.01	23 0.00	61 0.00	51 0.006	61 0.028	1 0.0307	0.0202	0.0070	0.0289	0.0289	0.0211	0.0263	0.0211	0.0272	0.0105	0.0281 0	0281 0.0	281 0.02	81 0.028	1.22
11	0061 0.0061 (0.0114 0.00	161 0.007	70 0.0132	2 0.028	1 0.0281	0.0281	0.0079	0.0061	0.0000	0.0061	0.0070	0.0061	0.0070	0.0061	0.0088 0	0.0070	0.0061 0	0140 0.	0158 0.0	0167 0.0	0.0 10.0	123 0.00	0.00	61 0.00	51 0.006	61 0.016	7 0.016	0.0070	0.0061	0.0070	0.0202	0.0070	0.0061	0.0061	0.0140	0.0061	0.0061 0	0254 0.0	272 0.02	46 0.028	- 22
13	0061 0.0061 (0.00	161 0.012	23 0.0132	2 0.0272	2 0.0263	0.0070	0.0061	0.0061	0.0088 (0.0000	0.0158	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 0	0070 0.	0070 0.0	0070 0.0	0.0	0.00	0.00	61 0.00	51 0.006	61 0.007	0 0.007(0.0061	0.0061	0.0061	0.0061	0.0061	0.0096	0.0061	0.0061	0.0061	0.0061 0	0202 0.0	246 0.02	46 0.027	21
1	0140 0.0114 0	0.0070 0.00	0.013	31 0.0125	3 0.028	1 0.0272	0.0263	0.0061	0.0061	0.0254 (0.0272 (0.0000 (0.0061	0.0061	0.0061	0.0061	0.0061	0.0158 0	0.0263 0.	0263 0.0	0263 0.0	061 0.0	0.00	0.00	61 0.00	51 0.006	61 0.016	7 0.016	0.0096	0.0061	0.0061	0.0246	0.0070	0.0070	0.0070	0.0070	0.0061	0.0140 0	0289 0.0	289 0.02	89 0.030	5
) SI	0079 0.0061 (0.0079 0.00	161 0.007	79 0.0075	9 0.0215	9 0.0228	0.0193	0.0193	0.0175	0.0237	0.0193 (0.0149 (0.0000	0.0254	0.0123	0.0132 (0.0202 (0.0237 0	.0219 0.	0193 0.0	0184 0.0	149 0.0	167 0.01	05 0.00	70 0.00	70 0.010	0.021	9 0.021	0.0175	0.0061	0.0175	0.0070	0.0202	0.0131	0.0123	0.0219	0.0070	0.0088 0	0211 0.0	193 0.00	79 0.008	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~
91;	0140 0.0158 (0.0114 0.00	0.012	23 0.011-	4 0.026	3 0.0246	0.0246	0.0123	0.0114	0.0219	0.0140	0.0070	0.0061	0.0000	0.0061	0.0079 (0.0070	0.0079 0	0123 0.	0140 0.0	0158 0.0	202 0.0	167 0.00	0.00	79 0.00	70 0.006	61 0.021	1 0.021	0.0202	0.0061	0.0132	0.0096	0.0114	0.0175	0.0070	0.0132	0.0070	0.0140 0	0219 0.0	211 0.01	67 0.011	4
, 17	0105 0.0061 (0.0105 0.00	900.0 960	96 0.012	3 0.014(0 0.0140	0.0061	0.0061	0.0079	0.0070	0.0088 (0.0114 (0.0061	0.0061	0.0000	0.0061	0.0061	0.0149 0	0149 0.	0167 0.0	0167 0.0	0.0 0.0	0.0	0.00	61 0.00	51 0.006	61 0.007	0 0.007	0.0132	0.0061	0.0149	0.0123	0.0070	0.0061	0.0061	0.0088	0.0061	0.0088 0	0105 0.0	140 0.01	40 0.008	~~~~
318 (0061 0.0070 0	0.088 0.00)61 0.006	51 0.006	1 0.007(0.0070	0.0061	0.0088	0.0140	0.0061	0.0061 (0.0061 (0.0061	0.0061	0.0061	0.0000	0.0061	0.0061 0	0.0061 0.	0061 0.0	0.0 0.0	0.0 100	0.0	0.00	61 0.00	51 0.006	61 0.006	1 0.006	0.0061	0.0061	0.0132	0.0070	0.0061	0.0114	0.0061	0.0061	0.0061	0.0088 0	0061 0.0	132 0.01	23 0.005	19
) ₆₁	0061 0.0061 (0.0061 0.00	0.006	51 0.006	0.009	6 0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 (0.0061	0.0061	0.0061	0.0061	0.0061	0000 (0.0061 0	0 1900	0061 0.0	0.0 0.0	0.0	0.00	0.00	61 0.00	51 0.006	61 0.006	1 0.006	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 0	0061 0.0	361 0.00	61 0.007	2
0 50	0219 0.0061 (0.0237 0.02	19 0.024	46 0.0202	2 0.022	8 0.0211	0.0070	0.0175	0.0219	0.0096	0.0070	0.0079	0.0061	0.0149	0.0061	0.0061 (0.0061	0.0000	0237 0	0237 0.0	0246 0.0	254 0.0	246 0.00	0.00	61 0.00	51 0.006	61 0.007	0 0.018	0.0219	0.0061	0.0219	0.0061	0.0070	0.0070	0.0070	0.0149	0.0088	0.0132 0	0114 0.0	237 0.00	70 0.007	2
5	0061 0.0070 0	0.0088 0.00	161 0.025	54 0.0175	5 0.0285	9 0.0254	0.0228	0.0211	0.0070	0.0132 (0.0114 (0.0088 (0.0061	0.0261	0.0061	0.0070	0.0061	0.0211 0	0000	0254 0.0	0246 0.0	132 0.0	237 0.00	0.00	61 0.00	51 0.006	61 0.023	7 0.014(0.0140	0.0061	0.0228	0.0096	0.0061	0.0088	0.0070	0.0211	0.0088	0.0123 0	0246 0.0	202 0.02	72 0.030	5
10	0061 0.0105 0	10167 0.00	700.0	79 0.0070	0 0.0263	3 0.0246	0.0079	0.0202	0.0061	0.0070	0.0070	0.0061	0.0061	0.0061	0.0061	0.0254 (0.0061	0.0219 0	0298 0.	0000	0289 0.0	061 0.0	237 0.00	0.00	61 0.00	51 0.006	61 0.010	5 0.0075	0.0298	0.0061	0.0307	0.0211	0.0061	0.0105	0.0105	0.0140	0.0070	0.0132 0	0132 0.0	228 0.01	58 0.011	4
	0319 0.0079 6	0193 0.02	46 0.025	54 0.0275	2 0.0281	1 0.0246	0.0096	0.0237	0.0140	0.0070	0.0158	0.0079	0.0061	0.0193	0.0061	0.0105	0061 0	0 0100	0307 0	0307 0.0	0000	0.0 890	0.0 862	0.0 0.00	70 0.00	51 0.000	0 0.014	9 0.0195	0.0272	0.0088	0.0289	0.0289	0.0061	0.0079	0.0096	0.0219	0.0158	0.0149_0	0.0 220	789 0.02	46 0.025	1 8
3 3	0.778 0.0061 0	10140 0.01	30 0 06	10.0112	4 0.0085	2 0.0088	0.0114	0.0061	0.0061	0.0061	19000	1900 0	1900.0	0.0246	0.0061	0.0070	0.1900	0.63.0	0 0200	0 020	0307 0.0	0.0 0.00	0.0 0.80	161 0.00	000	00.0	0.016	1001	0.0175	0.0070	0.0070	0.0140	0.0061	0.0061	0.0061	0.0070	0.0070	0 0061 0	0070 0.0	10.0	70 0 010	S 1 12
 	10000 0200	0.0 0 010.0	200 0 00.	11000 10	0000 U	00000	3210.0	10000	10000	10000	100000	10000	17000	1700 0	10000	13000	10000	0 3210						0000 100				1100 C	2100	00000	77000	0110.0	100010	10000	100000	0,000,0	0200.0	0 1700 0	00 1000	100 100	2000 0/	3 5
52		70:0 10001	300 0 177	00010 10	-500 0	1/00/0	C/ 10'0	1000 0	1000.0	1000.0	1000.0	100010	1000.0	Tonnin	1000'0	10000	10001	0 0/10/	0 2070	10 6/00	YN 6870	0.0 0.0		10/10/10/		1000 IO	710.0	+INU 0	10000	0.0061	0+7010	0+10:0	1000.0	110.0	0/00/0	0/00/0	0/00/0		170 T000	0.0 0.0	270.0	Z 2
				2000 0 F.		10000 I	1000.0	1000.0	1000.0	100000	100000	100000	10000	100000	10000	10000	10000										0000 U		10000	10000	1000.0	10000	100010	1000-0	100010	10000	1000.0	0 10000				5 5
21	1 1000.0 2000	10.0 1000.0	100.0 100	000.0	0000	100000	10000	1900.0	10000	1900.0	1900.0	1900.0	1900.0	1900.0	1900.0	1000.0	1900.0	1 1900.0	0 1000	1000	1000	0.0 100	10.0 100	40 0.0	00 000	0.014	0000 64	0000	1900.0	0.000	0.001	0.000	0.000	1900.0	1900.0	1900.0	1000.0	0 1000.0	1001 0.0	1010	0.000	7 3
628	1 0710.0 0600	10:0 0600.0	710.0 880	600.0 62	170'0 9	6170'0 6	701010	C/ 10:0	\$710.0	. 0/00.0	. 0/00.0	0/00/0	0.0184	1610.0	1900.0	1900.0	1900.0	0/007	0 6710.	111 01	10 2710	0.0	10 190	nnn 19	0070	10.0	110.0 0	4 0.011	1900.0	0.0088	0.0149	960010	0.0152	0.0088	9600.0	0600.0	0.0114	0 0600.0	170 SCI0	184 0.0.	110:0 20	4
P29 (0105 0.0061 (0.0088 0.01	149 0.013	32 0.015.	8 0.024	6 0.0149	0.0061	0.0061	0.0061	0.0070	0.0061	0.0061	0.0070	0.0096	0.0131	0.0061	00001	0.0158 0	0070 0.	0061 0.0	0.00	0.0	0.0 100	81 0.02	98 0.02	81 0.000	00.006	90070	0.0061	0.0070	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 0	0061 0.0	061 0.00	61 0.006	5
30 (0070 0.0088 (0.0158 0.0.	131 0.018	84 0.017.	5 0.028.	1 0.0263	0.0175	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	00061	0.0132 0	0140 0	0132 0.0	0158 0.0	158 0.0	184 0.00	961 0.00	61 0.00	0.000	0000	0 0.007	0.0088	0.0061	0.0070	0.0061	0.0061	0.0061	0.0061	0.0070	0.0061	0.0061 0	0061 0.0	061 0.00	61 0.006	5
16	0061 0.0061 (0.0061 0.00	0.006	61 0.006	1 0.007	0.0070	0.0070	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0079 0	0079 0	0088 0.0	0079 0.0	0.0	0.00	0.00	61 0.00	51 0.006	61 0.007	0 0.000	0.0061	0.0061	0.0070	0.0061	0.0061	0.0070	0.0070	0.0079	0.0061	0.0061 0	0061 0.0	061 0.00	61 0.006	5
32 (0061 0.0061 (0.0061 0.00	0.006	61 0.006	1 0.006.	1 0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 (0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0132 0	0123 0.	0079 0.0	0167 0.0	158 0.0	175 0.00	0.00	61 0.00	51 0.006	61 0.007	0 0.007	0.0000	0.0061	0.0079	0.0061	0.0061	0.0079	0.0061	0.0061	0.0061	0.0061 0	0070 0.0	061 0.00	61 0.007	2
33	0061 0.0061 (0.0061 0.00	0.006	61 0.006	61 0.013.	1 0.0079	0.0070	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0070	0.0061	0.0061	0.0061	0.0131 0	0070 0.	0070 0.0	0131 0.0	123 0.0	131 0.00	0.00	61 0.00	51 0.006	0.006	900.0 1:	0.0061	0.0000	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 0	0061 0.0	061 0.00	61 0.006	12
534	0061 0.0070 (0.0096 0.01	114 0.011	14 0.009	6 0.022	8 0.0219	0.0114	0.0079	0.0061	0.0070	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0193 0	0202 0	0237 0.0	0193 0.0	061 0.0	184 0.00	0.00	61 0.00	51 0.011	4 0.006	1 0.007	0.0061	0.0061	0.0000	0.0061	0.0061	0.0114	0.0131	0.0184	0.0061	0.0131 0	0228 0.0	219 0.0	19 0.016	5
35 (0096 0.0061 (0.0096 0.00	0.007	70 0.007	0 0.007	0 0.0070	0.0061	0.0061	0.0061	0.0061 (0.0061 (0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 0	0.0061 0.	0061 0.0	0061 0.0	061 0.0	0.00	0.00	61 0.00	51 0.006	61 0.006	1 0.006	0.0061	0.0061	0.0061	0.0000	0.0061	0.0061	0.0061	0.0061	0.0061	0.0096 0	0061 0.0	114 0.00	61 0.012	8
e36 (0061 0.0061 (0.0061 0.00	0.006	61 0.006	1 0.006.	1 0.0061	0.0061	0.0061	0.0061	0.0061 (0.0061 (0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 0.	0061 0.0	0061 0.0	061 0.0	0.00	0.00	61 0.00	51 0.006	61 0.006	1 0.006	0.0061	0.0061	0.0061	0.0061	0.0000	0.0061	0.0061	0.0061	0.0061	0.0061 0	0061 0.0	061 0.00	61 0.006	5
e37 (0061 0.0061 (0.0061 0.00	0.006	61 0.006	1 0.007	0 0.0070	0.0061	0.0061	0.0061	0.0061 (0.0061 (0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 0	0.0061 0.	0061 0.0	0061 0.0	061 0.0	0.00	0.00	61 0.00	51 0.006	61 0.006	0.006	0.0061	0.0061	0.0061	0.0061	0.0061	0.0000	0.0061	0.0211	0.0061	0.0175 0	0061 0.0	061 0.00	61 0.007	21
38	0061 0.0061 (0.0114 0.00	0.006	61 0.006.	0.006.	1 0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 (0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 0	0 1900	0061 0.0	0.0	0.0	0.0	0.00	61 0.00	51 0.006	61 0.006	900'0 1	0.0061	0.0061	0.0061	0.0061	0.0061	0.0096	0.0000	0.0123	0.0061	0.0061 0	0061 0.0	061 0.00	61 0.006	12
39	0061 0.0070 (0.0061 0.00	0.006	51 0.006.	1 0.008	8 0.0088	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 (0.0061	0.0061	0.0061	0.0061 (0.0061	0.0061 0	0 1900	0061 0.0	0.0 0.0	0.0	0.00	0.00	61 0.00	51 0.000	61 0.006	0.006	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0000	0.0061	0.0096 0	0061 0.0	061 0.00	61 0.006	12
97	0061 0.0061 (0.0061 0.00	0.006	51 0.006.	0.006,	1 0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 (0.0061 (0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 0	0.0061 0.	0061 0.0	0061 0.0	0.0	0.00	0.00	61 0.00	51 0.006	61 0.006	0.006	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0167	0.0000	0.0061 0	0061 0.0	061 0.00	61 0.006	5
541 (0061 0.0061 (0.0061 0.00	0.006	51 0.006.	0.006	1 0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 (0.0061	0.0061	0.0061	0.0061	0.0061	0.0070	0070 0.	0070 0.0	0070 0.0	0.0	0.00	0.00	61 0.00	51 0.006	61 0.006	0.006	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0096	0.0114	0.0000 0	0061 0.0	061 0.01	49 0.006	12
342	0061 0.0061 (0.0061 0.00	0.006	51 0.006.	0.006.	1 0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 (0.0061	0.0061	0.0061	0.0061	0.0061	0.0140 0	0140 0.	0131 0.0	0158 0.0	0.0	0.00	0.00	61 0.00	51 0.006	61 0.006	0.006	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 0	0000 0.0	10:0 960	14 0.013	2
543 (0061 0.0061 (0.0061 0.00	0.006	51 0.006.	0.015	8 0.0158	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0141 0	.0149 0.	0149 0.0	0149 0.0	175 0.0	149 0.00	0.00	61 0.00	51 0.006	61 0.006	0.006	0.0061	0.0061	0.0096	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 0	0061 0.0	000 0.00	61 0.029	8
4	0061 0.0061 (0.0061 0.00	0.006	61 0.006.	0.006.	1 0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 (0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0246 0	0.0228 0.	0211 0.0	0211 0.0	061 0.0	0.00	0.00	61 0.00	51 0.000	61 0.006	0.006	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 0	0061 0.0	061 0.00	00 0.030	5
19	0070 0.0061 (0.00	0.006	61 0.006	0.006	1 0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 0	0.0061 0	0.0061 0	.0061 0	0061 0.0	0061 0.0	061 0.0	0.00	0.00	61 0.00	51 0.006	1 0.006	1 0.006	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061	0.0061 0	0061 0.0	061 0.00	61 0.000	8

	12
4 0004 0012 0015 0016 0012 0013 0016 0013 0013 0013 0014 0006 0009 0019 0019 0005 0009 0009 0019 0012 0012 0012 0019 0012 0013 0013 0019 0009 0014 0011 0011 0011 0014 0003 0019 0019 0009 0014 0012 0012 0012 0012 0012 0012	0.0128
4 0010 0004 0016 0010 0011 0011 0012 0015 0015 0015 0015	0.0129
4 0 0055 0015 0015 0015 0015 0015 0015 0	0.0375
	0.0130
4 0 0019 0009 0009 0009 0009 0005 0011 0013 0015 0015 0015 0005 0009 0019 0005 0009 0009	0.0130
4 0 0108 60102 00105 00105 00105 00105 00105 00105 00105 00105 00105 00009 00105 00009 00105 00009 00105 00005 00105 000000	0.0128
• 0.0054 0.0056 0.0056 0.0052 0.0052 0.0052 0.0051 0.0056 0.0174 0.0256 0.0174 0.0154 0.0152 0.0152 0.0152 0.0152 0.0152 0.0152 0.0152 0.0152 0.0152 0.0152 0.0152 0.0152 0.0152 0.0154 0.0264 0.0154 0.0154 0.0154 0.0154 0.0154	0.0489
4 0 0.0256 0.0276 0.023 0.0347 0.023 0.0440 0.015 0.0151 0.0150 0.0154 0.0217 0.0152 0.0134 0.0255 0.0154 0.0255 0.0154 0.0256 0.0157 0.0156 0.0156 0.0156 0.0156 0.0156 0.0154 0.0256 0.0154 0.0156 0.0154 0.0156 0.0154 0.0156 0.0154 0.0156 0.0154 0.0156 0.0154 0.0156 0.0154 0	0.0469
60 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	0.0172
4m 00170 0023 00171 00179 00179 00179 00179 00125 00250 00251 00133 00145 00189 00149 00189 00169 00169 00179 00141 00174 00201 00179 00194 0024 00251 00201 00170 00141 00174 00204 00251 00201 00024 00251 00201 00171 00174	0.0250
1 00266 00382 00473 00382 00461 00381 00566 00461 00381 00462 00460 04510 00321 00412 00321 00421 00411 0025 00411 0025 00411 0025 00411 0025 00411 0025 00411 0025 00411 0025 00411 0025 00411 0025 00411 0025 00411 0025 00411 0025 00411 0025 00411 0025 00411 0025 0041 0025 00411 0025 00411 0025 00411 0025 00411 0025 00411 0025 00411 0025 00411 0025 00411 0025 00411 0025 0041 0041 0041 0041 0041 0041 0041 004	0.0532
41 00155 00145 00145 00145 00170 00254 00411 00234 00411 00397 00150 00135 00131 00109 00137 00121 00131 00131 00131 00131 00140 00230 00230 0023 00132 00123 00123 00123 00133 00143 00133 00143 00137 00142 00137 00142 00137 00142 00137 00142 00137 00142 00137 00142 00141 00144 00141 00144 00142 00142 00142 00142 00142 00142 00142 00142 00142 00142 00142 00142 00142 00142 00142 00142 00142 00142 00142 00141 00144 00141 00144 00141 00144 0014	0.0418
42 0 00140 00139 00105 00105 00105 00105 00205 00205 00125 00150 00125 00101 00125 00112 00150 00112 00151 00112 00148 00148 00148 00143 00113 00113 00113 00145 00148 00148 00148 00149 00149 00105 00125 00201	0.0388
4 0 0251 0209 0013 0013 0024 0023 0013 0024 0023 0024 0024 0020 0034 0014 0020 0034 0014 0013 0014 0013 0014 0013 0014 0013 0014 0013 0014 0013 0013	0.0477
42 0 00199 00168 00193 00159 00159 00159 00159 00159 00251 00259 00253 00259 00259 00259 00259 00251 00252 00259 00251 0).0263
er. 00245 0245 0249 00169 00156 0023 00413 0185 0021 02413 0245 0257 0244 025 0214 025 0214 0215 0214 0215 0251 0259 0244 0215 0251 0244 0214 0215 0244 0215 0244 0215 0245 0215 0245 0215 0251 0245 0215 0245 0215 0245 0215 0245 0215 0245 0215 0245 0215 0245 0215 0245 0215 0245 0215 0245 0215 0245 0215 0245 0245 0245 0245 0245 0245 0245 024	0.0268
	0199
etw 0017 0012 0012 0012 0013 0014 0015 0012 0014 0015 0014 0015 0014 0015 0010 0010	0179
4re 0 0110 0104 0010 0010 0010 0011 0011 0	0148
	0.0237
21 0018 0015 0015 0015 0015 0015 0015 001	0.0476
	0.0275
4 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	0.0490
210 0018 0015 0015 0024 0027 0027 0020 0200 0200 0200 0200	0.0227
2x 0015 0015 0015 0015 0015 0015 0015 001	0.0394
2x 0012 0010 0013 0013 0013 0013 0013 0013	0.0134
27 00150 00106 00119 00113 00116 00115 00118 00116 00119 00111 00102 00109 00106 00109 00106 00109 00105 00125 00125 00125 00125 00125 00124 00126 0004 00156 00115 00113 00119 00109 00107 00107 00109 00105 00114 00099 00105 00124 0012	0.0134
cm 00166 0025 00179 0022 00189 00222 00189 00222 00189 00252 00139 00254 00258 00190 00151 00149 00155 00124 00154 00054 000054 00054 00054 00054 00054 00054 00054 0005	0.0245
cm 00181 0012 00124 00125 00124 00131 00130 00124 00141 00131 00130 00135 00112 00131 00131 00131 00131 00130 00132 00143 00140 00131 00131 00135 00143 00140 00131 001300000000	0.0163
cm 00155 00158 00259 00277 00279 00279 00279 00259 00269 00269 00156 00159 00131 00130 00119 00115 00141 00141 00149 00114 00249 00259 00269 00254 00259 00269 00254 00259 00269 00254 00259 00269 00254 00259 00259 00259 00259 00259 00259 00159 00139 001	0.0178
420 0 0112 0109 0112 0114 0114 0114 0114 0114 0114 0114	0.0133
42 0012 0011 0017 0012 0010 0012 0012 001	0.0155
	0.0143
41 0015 0014 0017 0024 0019 0015 0014 0019 0015 0011 0014 0018 0021 0014 0013 0014 0011 0014 0011 0014 0011 0014 0013 0010 0014 0013	0:0301
	0.0195
	0.0128
421 0012 0016 0016 0016 0016 0016 0016 00	0141
Cw 0 0112 0014 0115 0116 0115 0116 0115 0115 0115 0115	0.0132
Cyse 0 0111 00113 00107 00111 00113 00107 00112 00112 00102 00103 00107 00100 00105 00104 00090 00094 00109 00095 00104 00105 00105 00103 00105 0009 00104 00105 0009 00104 00105 0009 00104 00105 00104 00105 0009 00104 00105 0009 00104 00105 0009 00104 00105 0009	0.0131
4a 0110 010 010 010 010 010 011 011 011 0	0.0130
4 0 0111 0015 0016 0016 0011 0016 0011 0016 0011 0016 0013 0015 0017 0019 0017 0019 0010 0006 0010 0009 0005 0009 0013 0014 0014 0014 0014 0014 0014 0011 0009 0008 0010 0008 0010 0018 0013 0013 0013	0.0134
212 0012 0010 0016 0019 0015 0012 0015 0015 0014 00140 0014 0014 0014 001	0.0214
20 0013 0013 0013 0013 0013 0013 0013 00	0.0394
24 0 0015 0015 0015 0015 0015 0015 0015 0	0.0398
24 0017 0002 0009 0013 0010 0012 0010 0012 0010 0012 0010 0000 0008 0010 0000 000	00067

TABLE 4. Degree of influence.

Elements	D	R	D+R	D-R
e ₁	0.48333	0.77678	1.26011	-0.29345
e ₂	0.48645	0.67424	1.16069	-0.18780
e ₃	1.30265	0.72494	2.02759	0.57771
e_4	0.50007	0.77735	1.27742	-0.27728
e ₅	0.48924	0.84333	1.33256	-0.35409
e ₆	0.48333	0.80600	1.28933	-0.32267
e ₇	1.37553	1.15955	2.53508	0.21599
e_8	1.13275	1.04391	2.17666	0.08885
e ₉	0.67547	0.87323	1.54870	-0.19776
e_{10}	1.07284	0.73535	1.80819	0.33748
e ₁₁	1.77909	0.60970	2.38879	1.16940
e ₁₂	0.92767	0.69639	1.62406	0.23128
e ₁₃	0.76414	0.67097	1.43511	0.09317
e ₁₄	1.11477	0.59024	1.70501	0.52453
e ₁₅	1.19979	0.52461	1.72441	0.67518
e ₁₆	1.05442	0.75778	1.81220	0.29664
e ₁₇	0.75176	0.53933	1.29108	0.21243
e_{18}	0.57525	0.67535	1.25060	-0.10010
e ₁₉	0.49418	0.54753	1.04171	-0.05335
e ₂₀	1.12160	0.98894	2.11054	0.13266
e ₂₁	1.17837	1.12030	2.29868	0.05807
e ₂₂	1.05978	1.08372	2.14351	-0.02394
e ₂₃	1.41782	1.18437	2.60220	0.23345
e ₂₄	0.87244	0.85053	1.72298	0.02191
e ₂₅	0.82269	0.93241	1.75510	-0.10972
e ₂₆	0.53681	0.55051	1.08732	-0.01370
e ₂₇	0.53322	0.55559	1.08880	-0.02237
e ₂₈	0.90760	0.54976	1.45736	0.35783
e ₂₉	0.75313	0.54258	1.29571	0.21055
e ₃₀	0.78753	0.81514	1.60266	-0.02761
e ₃₁	0.51418	0.81772	1.33190	-0.30354
e ₃₂	0.59182	0.86318	1.45499	-0.27136
e ₃₃	0.56613	0.52238	1.08851	0.04375
e ₃₄	0.90016	0.89178	1.79194	0.00839
e ₃₅	0.52735	0.78639	1.31373	-0.25904
e ₃₆	0.48333	0.56789	1.05122	-0.08456
e ₃₇	0.52792	0.69045	1.21836	-0.16253
e ₃₈	0.50990	0.60385	1.11375	-0.09396
e ₃₉	0.50167	0.80355	1.30523	-0.30188

TABLE 4. (Continued.) Degree of influence.

e_{40}	0.49904	0.56662	1.06566	-0.06758
e_{41}	0.51840	0.70050	1.21890	-0.18209
e ₄₂	0.57948	0.99798	1.57746	-0.41850
e ₄₃	0.67998	1.09052	1.77050	-0.41054
e ₄₄	0.66086	0.99324	1.65410	-0.33238
e ₄₅	0.48460	1.10205	1.58665	-0.61745

mechanisms), e₃₉ (effective change management), e₃₈ (project plan review), e₃₅ (contractual terms and conditions), e₃₀ (alternative solutions). These elements form the top-level in the ISM framework and depend on other lowerlevel elements. The third cluster (linkage) consists of the elements with strong driver and dependence power. These elements have high influencing dynamics such that any action on these elements will affect others and they will also get easily affected by other elements. The linkage elements are: e7 (clear requirements and goals), e8 (stakeholder expectations), e9 (identifying constraints), e10 (capable team members), e13 (user involvement), e18 (responsibilities and commitments), e_{21} (project schedule), e_{22} (resource estimation), e₂₃ (initial cost estimates), e₂₄ (technology), e25 (troubleshooting/testing), e31 (managing uncertainties), e₃₂ (proper equipment/tool), e₃₄ (allocate sufficient resources), e₃₇ (monitoring and feedback), e₄₃ (key deliverables), e_{44} (deliverables dates). These elements come in the middle of the ISM framework and link the independent and dependent elements. A closer review of linkage cluster highlights two clusters within this cluster; one with high driving power, including elements: e7 (clear requirements and goals), e_8 (stakeholder expectations), e_{10} (capable team members), e13 (user involvement), e21 (project schedule), e22 (resource estimation), e₂₃ (initial cost estimates), and the other one with low driving power, having the remaining elements of linkage cluster. Elements having strong driving power but weak dependence power fall into the fourth cluster (independent). The elements e₃ (project strategy/summary), e_{11} (project manager competence), e_{12} (communication and coordination), e14 (client consultation), e15 (top management support) e₁₆ (training), e₁₇ (project authority), e₂₀ (project finances), e27 (company's strategic intent), e28 (organizational capabilities), and e₂₉ (business plan/vision) are located within the independent cluster. These elements demonstrate high levels of influence on connected elements at higher levels in the digraph. It may be noted that elements with a very strong driving power called the "primary elements" fall into the linkage and independent clusters.

Sub-section D, on the other hand, provides the details of the second of the two analysis that used expert group II. From the DEMATEL analysis, it can be seen

FIGURE 12. Cause and effect diagram for elements.

FIGURE 13. Elements classification.

from Table 4, that eight elements have large influencing degrees including, e_{11} (project manager competence), e_{23} (initial cost estimates), e_7 (clear requirements and goals), e_3 (project strategy/summary), e_{15} (top management support), e_{21} (project schedule), e_8 (stakeholder expectations), e_{20} (project finances), e_{14} (client consultation),

e₁₀ (capable team members), e₁₆ (training). Moreover, based on the (D+R) values, the elements: e23 (initial cost estimates), e7 (clear requirements and goals), e₁₁ (project manager competence), e₂₁ (project schedule), e_8 (stakeholder expectations), e_{22} (resource estimation), e₃ (project strategy/summary), are the topmost among the important elements. Furthermore, on the basis of (D-R) dataset, it is observed that nineteen elements: e₃ (project strategy/summary), e7 (clear requirements and goals), e8 (stakeholder expectations), e_{10} (capable team members), e₁₁ (project manager competence), e₁₂ (communication and coordination), e13 (user involvement), e14 (client consultation), e_{15} (top management support), e_{16} (training), e17 (project authority), e20 (project finances), e21 (project schedule), e23 (initial cost estimates), e24 (technology), e₂₈ (organizational capabilities), e₂₉ (business plan/vision), e₃₃ (operational concepts), e₃₄ (allocate sufficient resources), belong to the cause group. These cause group elements may be understood as independent elements, which have high priority and direct impact on the other elements, indicating that these elements are the most influential ones and should be addressed first to figure out the elements under the effect group. Elements who fell into the effect group having negative (D-R) value include: e1 (future expansions), e2 (capturing dependency relationships among activities), e₄ (project environment), e5 (market strategy), e6 (deployment strategy), e9 (identifying constraints), e18 (responsibilities and commitments), e19 (building trust), e22 (resource estimation), e25 (troubleshooting/testing), e26 (capturing corporate knowledge), e27 (company's strategic intent), e30 (alternative solutions), e₃₁ (managing uncertainties), e₃₂ (proper equipment/tool), e35 (contractual terms and conditions), e36 (managing politics), e₃₇ (monitoring and feedback), e₃₈ (project plan review), e₃₉ (effective change management), e₄₀ (control and information mechanisms), e₄₁ (track progress), e₄₂ (setting milestones), e43 (key deliverables), e44 (deliverables dates), e₄₅ (client acceptance). These elements tend to be easily influenced by the other elements.

V. DISCUSSIONS

The present study presents the interrelationships among the software scope definition elements which are modeled using the ISM and DEMATEL methodology. ISM and DEMATEL share some similar characteristics as both investigate the relationships among multiple criteria so analysis of the interdependence of software scope definition elements is carried out with the help of ISM and DEMATEL analysis. The MICMAC analysis and the ISM framework diagrammatically illustrate the strong interdependency among the elements and the possible impact of some elements on others in the framework. The elements in the autonomous cluster are considered important too, but they possess weak interaction in terms of driving and dependence power, such that these elements do not help achieve or derive other elements. Therefore, we can say that these elements such as e1 (future expansions), e2 (capturing dependency relationships among activities), e4 (project environment), e₅ (market strategy), e₆ (deployment strategy), e₁₉ (building trust), e₂₆ (capturing corporate knowledge), e₃₃ (operational concepts), e₃₆ (managing politics) are relatively disconnected with other elements and can be selected as not applicable at the planning stage, as they also have low weights in SPSRI method and can be defined at the later stages of the project. The second cluster includes the elements which are strongly dependent on others, such as e_{42} (setting milestones) and e45 (client acceptance) are dependent on the requirements and stakeholder expectations of the project and therefore, should be defined after defining the other elements. Furthermore, the improvement of the elements in this cluster depends on other elements, namely e7 (project mission/clear requirements and goals), e8 (stakeholder expectations), and so on. The elements: e₃ (project strategy/summary), e7 (project mission/clear requirements and goals), e_8 (stakeholder expectations), e_{10} (capable team members), e₁₁ (project manager's competence), e₁₂ (communication and coordination), e13 (user/client involvement), e_{14} (client consultation), e_{15} (top management support), e_{20} (project finances), e₂₁ (project schedule), e₂₂ (resource estimation), and e23 (initial cost estimates) exhibit maximum driving powers and are at the bottom levels of the ISM framework. The characteristics of these elements are that they have higher weights in the SPSRI method, meaning that they have been mentioned as the critical factors for measuring the completeness and quality of the scope definition of software projects [51]. Moreover, these elements can also be found in the recent literature review studies (see Figure 3) [15], [35], [111], [116], [119]–[121]. These studies have emphasized these elements for the software project scope definition. This research work extends these findings by identifying the criticality of these elements and their influence on the other elements. Furthermore, the interdependency of these elements suggests that software project managers/developers or organizations need to concentrate on these elements for driving a high-quality scope definition. In literature, element e7 (project mission/clear and realistic requirements and goals) is cited as the most critical element of the software project scope definition [15], [119]-[121]. This finding is also supported by this study since this element demonstrates the highest levels of driving and dependence power in MICMAC analysis and ISM framework emphasizing its influence on other elements. This is because when defining the other elements such as project cost estimates or schedule estimates, requirements of the projects are considered. Besides, this element is clustered in the linkage quadrant and also closely associated with other elements, to name a few: e_8 (stakeholder expectations), e_{21} (project schedule), e22 (resource estimation), and e23 (initial cost estimates), highlighting the significant role that these closely linked elements play in the remaining linkages within the framework. The elements with higher dependency powers located at the top level of the ISM framework such as e_{42} (setting milestones), e₄₃ (key deliverables), e₄₄ (deliverable dates), and e₄₅ (client acceptance), demonstrate how the

interconnected elements below in the framework can influence these elements which include the project requirements as well. The elements in the fourth cluster include e_{11} (project manager competence), e₁₂ (communication and coordination), e14 (client consultation), e15 (top management support) e_{16} (training). These elements influence other elements such as project manager competence, communication, and coordination and client consultation will result in better elicitation of the project requirements. Thus, the organizations should focus on these elements as well to improve the scope definition. The primary contributors to scope definition completeness and quality are the elements in the linkage cluster as they have high influencing dynamics and substantially interact with other elements. The results indicate that by enhancing these elements, other elements can also be improved which will help in achieving a better scope definition. Therefore, project managers should define these elements properly in the scope definition at the planning stage of the project development life cycle, recognizing the influence they have on other elements in the framework.

From the DEMATEL analysis, it is observed that, e3 (project strategy/summary), e7 (clear requirements and goals), e8 (stakeholder expectations), e10 (capable team members), e11 (project manager competence), e12 (communication and coordination), e₁₃ (user involvement), e₁₄ (client consultation), e₁₅ (top management support), e₁₆ (training), e17 (project authority), e20 (project finances), e21 (project schedule), e₂₃ (initial cost estimates), e₂₄ (technology), e₂₈ (organizational capabilities), e₂₉ (business plan/vision), e₃₃ (operational concepts), e₃₄ (allocate sufficient resources), come under the cause group and are considered as the most influential elements. Among the cause group elements, element e₁₁ (project manager competence), has the highest value of causality (D-R). Furthermore, it also has the 3rd highest value of (D+R) and the highest value of influencing degree (D) of 1.77909, confirming that it has the highest influence on the other elements while accepting very little impact. Furthermore, this element has also been recognized as the independent element in the MICMAC analysis with high driving power. Therefore, based on the results of both methods, element e_{11} can be recognized as the core element. This could be supported by the research of [117], who argued that the project manager's skills as a scope definition element. Moreover, e_{15} (top management support) has the second highest (D-R) value and possesses a high (D+R) value, which implies that it has significant importance. The next element in the cause group with a high (D-R) value is e₃ (project strategy/summary) with the fourth-highest value of D as well. It suggests that this element also greatly influence the scope definition. Similarly, based on the MICMAC analysis, these cause group elements are also in the independent cluster having strong driving powers (Figure 13). Although elements e₇, e8, e10, e13, e21, e23, e24, e34 are classified as linkage elements in ISM analysis, these elements are in the cause group in the case of DEMATEL analysis because of their high influencing values. These facts suggest that the primary elements

with high driving powers fall into the linkage and independent clusters in ISM analysis. Element e₃₃ (operational concepts), which is located in the autonomous cluster in MICMAC analysis, is in the cause group. Although its (D-R) value is positive, its influencing degree (D) is not high enough, therefore, we can say that it is an independent element and does not have a notable impact on other elements. Elements in the autonomous cluster have the least strength of the influence on other elements in the DEMATEL analysis. Moreover, the elements with the higher dependency powers in the MICMAC analysis are categorized into the effect group. Among all the effect group elements, e45 (client acceptance) has the least (D-R) score of -0.61745 which means it receives the highest impact. However, its (D+ R) value of 1.58665 indicates that it is still important in the software project scope definition. Moreover, all the cause elements can enhance this element, such as well-written requirements will help in driving this element directly. The next element in the effect group is e_{42} (setting milestone) with a (D-R) of -0.4185. However, owing to its high (D+R) and D values, it is considered important. The element with a (D-R) of -0.41054 and high D and (D+R) values of 0.67998 and 1.770503 respectively signify that it places third in the effect group. Considering the results, one can state that the majority of the results obtained from the ISM and DEMATEL are consistent to some extent. Therefore, a very similar interpretation to that used in the ISM is applied in DEMATEL; elements from the cause group are of great importance for the software project scope definition and should be addressed at an early stage and more carefully by the project managers as these elements might be treated as the origin of all the other effect group elements. Further, effect group elements are significantly driven by constructing efforts on the cause group elements, thus for a high-quality scope definition, cause group elements should be tackled on a high priority basis.

This study has some implications for project managers. This research offers a base to extend the understanding of important software project scope definition elements. Hierarchic and cause-effect structures of the elements were clarified based on structural modeling methodologies. First, a comprehensive understanding of the relationships among the elements was revealed using the ISM methodology. Then, DEMATEL was employed to quantitatively calculate the importance, intensity, and effect in the interaction among the elements. The above discussions highlight the critical elements that can enhance the scope definition of software projects. The results of the combined ISM and DEMATEL approach used in this work are very relevant and useful for the managers. The findings of the study offer project managers and team members a more focused approach to segregate the important elements in order to achieve a high-quality scope definition. The important elements having higher driving power are required to be treated at an early stage, so as there may be few other dependent elements influenced by them. These elements include: e_{23} (initial cost estimates), e7 (project mission/clear requirements and goals), e11 (project

TABLE 5. The same criteria used in the previous study were applied in this study (except for the years).

Inclusion criteria	Exclusion criteria
IC1. The main criterion is that the papers should focus on software scope, its issues, the causes of these issues and their effects, and techniques or methods to mitigate them AND	EC1. Papers not addressing the software project scope and its issues, papers outside the scope of our study OR
IC2. All papers must be written in the English language as either journal papers, conference papers, or book chapters AND	EC2. Papers not written in the English language OR
IC3. Related papers published between 2016 & 2020 AND	EC3. Papers published before 2016 OR
IC4. Papers must be available in full-text format.	EC4 . Duplicate papers (except the recent and improved ones) Gray papers (without bibliographic information such as publication date/type, volume, and issue numbers) OR
	EC5. Dissertations, presentations, and review papers. Moreover, very short papers such as editorials, position papers, abstracts, and keynotes were also excluded.

TABLE 6. We adapted the previously used quality assessment criteria and the scoring process for this study.

Sr.	QA question	Scoring
1	Are the research questions clearly described?	Yes = 1, Partially = 0.5 , No = 0
2	Is the study supported by enough literature review?	Yes = 1, Partially = 0.5 , No = 0
3	Is the scope problem clearly explained and supported by sufficient data?	Yes = 1, Partially = 0.5 , No = 0
4	Are the causes and effects of the scope problem clearly explained and supported by sufficient data?	Yes = 1, Partially = 0.5 , No = 0
5	Do the authors discuss the credibility and limitations of their study?	Yes = 1, Partially = 0.5 , No = 0

manager's competence), e21 (project schedule), e8 (stakeholder expectations), e₂₂ (resource estimation), e₂₀ (project finances), e₃ (project strategy/summary), e₁₆ (training), and e₁₀ (capable team members). These important elements are identified based on their higher degree of centrality (Table 4), and they are also closely related to other elements. Furthermore, the remaining cause group elements such as e12 (communication and coordination), e13 (user/client involvement), e₁₄ (client consultation), e₁₅ (top management support), e₂₈ (organizational capabilities), e₂₉ (business plan/vision) which also have high driving power and high influencing degree are also significant as they may help to attain other elements appearing at the top and middle of the ISM model. These elements require continuous attention from the project managers, therefore paying attention to these elements will facilitate the achievement of a high-quality scope definition.

26864

VI. CONCLUSION AND FUTURE WORK

To formulate the scope definition of software projects, elements that constitute a complete scope definition were previously identified and ranked. However, it is important for practitioners to understand the characteristics and interrelationships of these elements. Considering the highly important role of scope definition in the success of software projects, this study has extended the existing research and understanding of the scope definition elements. Based on the combination of the structural modeling methodologies ISM and DEMATEL, a framework has been proposed, to analyze the relationship structure and identify the most influencing elements. ISM methodology was employed to explore the interrelationships and hierarchy among the elements that yielded further insights into the interdependency of the elements. Further, the MICMAC analysis helped in examining the influencing dynamics of the elements, which also offered

Papers	Q-1	Q-2	Q-3	Q-4	Q-5	Total Score
Pace [133]	1	0	0.5	0	0	1.5
Khanfar et al. [33]	0	1	0	0	0	1
Al-Rubaiei et al. [66]	0	0	0.5	0.5	0	1
Ajmal et al. [35]	0	1	0.5	0.5	0.5	2.5
Hughes et al. [134]	1	0	0	0	0.5	1.5
Ibraigheeth and Fadzli [20]	0	0	0	0	0.5	0.5
Lampa et al. [15]	0	0	1	1	0	2
Adam and Danaparamita [65]	0	1	0.5	0	0	1.5
Alrumaih et al. [135]	0	0.5	0.5	0	0	1
Zahid et al. [41]	0	0.5	0	0	0	0.5
Hidding and Nicholas [136]	0	1	0	0	0.5	1.5
Wachnik [137]	0	0.5	0	0	0	0.5
Bezdrob et al. [138]	0.5	0.5	0	0	0.5	1.5
Sudhakar [139]	0	1	0	0	0	1
Taherdoost and Keshavarzsaleh [140]	0	1	0	0	0	1
Aranyossy et al. [141]	1	0.5	0	0	0	1.5
Chesterman Jr.et al. [142]	0	0.5	0	0	0	0.5
Hughes et al. [77]	0	1	0.5	0	1	2.5
Sarif et al. [143]	0.5	0.5	0	0	0	1
Hughes et al. [144]	0	1	0	0	0.5	1.5
Goedeke et al. [108]	0	1	0	1	0	2
Yost et al. [145]	0	1	0	0	0	1
Taghavi and Woo [109]	0	1	0	1	0	2
Atoum [146]	0	0.5	0.5	0	0.5	1.5
Surbakti et al. [147]	0	1	0	0	0	1
Saini et al. [148]	0	0.5	0.5	0	0.5	1.5
Majerowicz and Shinn [149]	0	1	0	0	0	1
Symons et al. [110]	0	1	1	0	0	2
Haoues et al. [150]	0	1	0.5	0	0	1.5
Prasanth et al. [111]	0	0.5	0.5	0.5	0.5	2
Jarzębowicz and Ślesiński [112]	1	1	0.5	0.5	0.5	3.5
Maalej et al. [151]	0	0.5	0	0	0	0.5
Sardjono and Retnowardhani [113]	1	0	0.5	0	0.5	2
Fellir et al. [114]	0	1	0.5	0	0.5	2
Aguilar et al. [115]	0	0.5	0.5	0	1	2
Carvalho et al. [116]	0	1	0.5	0.5	0.5	2.5
Yesudas et al. [117]	0	1	0.5	0.5	0	2
Frączkowski and Gładysz [152]	0	0.5	0	0	0	0.5
Tu et al. [118]	1	0.5	0.5	0.5	1	3.5
Wahbeh et al. [119]	0	1	0.5	0.5	1	3
Asadabadi et al. [120]	0	1	0.5	0.5	0.5	2.5

 TABLE 7. Based on the above QA questions, studies that provided complete information were given 1 score, 0.5 for partial information, and 0 for no information against any question. The papers which had greater than or equal to 2 score were finally selected.

useful insights into the relative significance of the elements and their interdependencies (autonomous, dependent, linkage, and independent). Moreover, the DEMATEL analysis output gave a further detailed depiction of the cause-effect relationship of the elements, including the assignment of numerical values which corresponded to the strength each element exerted on the others and to the group of elements as a whole.

The findings of this study demonstrate the important interconnectedness of several scope definition elements. The results call attention to the elements with maximum driving power to be considered properly when defining the scope of software projects as these elements influence many other elements in the framework. The most influential elements with high driving power in the proposed framework are: e7 (project mission/clear requirements and goals), e8 (stakeholder expectations), e₁₀ (capable team members), e₁₁ (project manager's competence), e_{12} (communication and coordination), e13 (user/client involvement), e14 (client consultation), e15 (top management support), e₂₀ (project finances), e₂₁ (project schedule). These elements help drive other elements and are also dependent on each other, such as project requirements help in driving the project cost and schedule estimate, but these three factors are connected in such a way that any change to one of them has an effect on other as well. Moreover, the DEMATEL approach uncovered the influenced and influential interactions (causal interactions) among the elements which corresponded to nineteen cause group elements. The combination of ISM and DEMATEL proved suitable to better understand elements' structure and identify the most influential elements, which corresponded to the following elements: e23 (initial cost estimates), e7 (project mission/clear requirements and goals), e₁₁ (project manager's competence), e₂₁ (project schedule), e₈ (stakeholder expectations), e₂₂ (resource estimation), e₂₀ (project finances), e₃ (project strategy/summary), e_{16} (training), and e_{10} (capable team members). Therefore, this interrelationship analysis of elements revealed the most governing elements and have provided a precise way of identifying the significant elements which should be considered by the project managers and team members for defining the scope of software projects. These elements with high driving power and high influencing degree should receive extra attention for defining high-quality project scope.

This study has contributed to the literature by proposing a relationship model (hierarchical and cause-effect relationship) of the scope definition elements to guide the practitioners with a deeper understanding of these elements when defining the scope of projects. By adopting ISM and DEMA-TEL methodology, this research study provided new insights over the previous research in recognizing the significance and influence of specific elements and how these contribute to scope definition. Advanced guidelines based on the critical elements are provided for software practitioners to foster improvements in achieving a high-quality scope definition. The proposed framework can serve as a step forward towards improving the success rate of software projects by defining better scope statements at the planning stage. However, this study has some limitations as experts' opinions have been sought to recognize the driving and dependence power of the elements and for the development of the cause-effect diagram. Therefore, the developed framework is dependent on these opinions and has some element of bias. Moreover, the number of experts was limited to fifteen, which could have also caused bias in the study due to their specific knowledge and experience. Thus, expanding the number of experts in the future study is also suggested. Another limitation of this research is the exclusion of some digital repositories e.g., Scopus during the systematic literature review. However, we have selected well-known scientific repositories i.e., IEEE Xplore, Springer, Elsevier Science Direct, and ACM Digital Library. An immense number of good quality studies are provided by these repositories. Moreover, to not miss any relevant studies, backward snowballing technique has also been applied by tracking the references of the selected studies. Furthermore, software project scope definition elements have been searched manually and using the SLR from 1994 to 2020 in well-renowned repositories, therefore according to some existing recent SLR studies, the used electronic data sources are sufficient to generalize the results of the study [122]–[132]. Further, the framework has not been statistically validated therefore, using Structural Equation Modeling (SEM) for statistically validating the framework using a larger number of experts would be fruitful for future research.

APPENDIX A

See Figure 1.

APPENDIX B INCLUSION/EXCLUSION CRITERIA

See Table 5.

APPENDIX C QUALITY ASSESSMENT CRITERIA AND ASSESSMENT TABLE

See Table 6 and 7.

APPENDIX D LEVELS PARTITIONING

See Figure 7.

REFERENCES

- J. R. Turner and Y. Xue, "On the success of megaprojects," *Int. J. Manag. Projects Bus.*, vol. 11, no. 3, pp. 783–805, Jul. 2018, doi: 10.1108/ijmpb-06-2017-0062.
- [2] A Guide to the Project Management Body of Knowledge (PMBOK Guide).
 6th ed. Newtown Square, PA, USA: Project Management Institute, 2017.
- [3] S. Naeem, B. Khanzada, T. Mubashir, and H. Sohail, "Impact of project planning on project success with mediating role of risk management and moderating role of organizational culture," *Int. J. Bus. Social Sci.*, vol. 9, no. 1, pp. 88–98, 2018.
- [4] M. N. Mirza, Z. Pourzolfaghar, and M. Shahnazari, "Significance of scope in project success," *Procedia Technol.*, vol. 9, pp. 722–729, 2013, doi: 10.1016/j.protcy.2013.12.080.

- [5] H. Taherdoost and A. Keshavarzsaleh, "Managing successful IT project; marketing perspective," *Recent Adv. Comput. Sci. Appl.*, vol. 28, no. 11, pp. 144–153, Aug. 2018.
- [6] A. O. Ogunberu, T. O. Olaposi, and S. O. Akintelu, "Factors affecting the choice of project scope management practices among telecommunication organizations," *Global J. Manage. Bus. Res.*, vol. 16, no. 3, pp. 13–20, 2016.
- [7] K. H. Rose, A Guide to the Project Management Body of Knowledge (PMBOK Guide), 5th ed. Newton Square, PA, USA: Project Management Institute, 2013.
- [8] E. S. Mtsweni, S. A. University of South Africa, T. Hörne, and J. A. V. D. Poll, "Soft skills for software project team members," *Int. J. Comput. Theory Eng.*, vol. 8, no. 2, pp. 150–155, 2016, doi: 10.7763/ijcte.2016.V8.1035.
- [9] J. M. McQuighan and R. J. H. Ii, "Scope as a leading indicator for managing software development," in *Proc. 9th Int. Conf. Softw. Eng. Res., Manage. Appl.*, Aug. 2011, pp. 235–241.
- [10] M. K. Fageha and A. A. Aibinu, "Prioritising project scope definition elements in public building project," *Australas. J. Construct. Econ. Building*, vol. 14, no. 3, p. 18, 2014.
- [11] O. J. Ameh and A. B. Ogunyemi, "Challenges of scope definition for projects using conventional construction procurement method in nigeria," *J. Building Perform.*, vol. 6, no. 1, pp. 77–82, 2015.
- [12] N. Agarwal and U. Rathod, "Defining 'success' for software projects: An exploratory revelation," *Int. J. Project Manage.*, vol. 24, no. 4, pp. 358–370, 2006, doi: 10.1016/j.ijproman.2005.11.009.
- [13] N. Kumari and S. A. Pillai, "A study on project scope as a requirements elicitation issue," presented at the Int. Conf. Comput. Sustain. Global Develop. (INDIACom), 2014.
- [14] J. A. G. Correa, S. L. S. Castañeda, D. A. V. Quintero, and G. E. Giraldo, "Identification and analysis of project management success factors in information technology SMEs," *Int. J. Inf. Technol. Project Manage.*, vol. 9, no. 4, pp. 73–90, Oct. 2018, doi: 10.4018/ijitpm.2018100105.
- [15] I. L. Lampa, A. de Godoi Contessoto, A. R. Amorim, G. F. D. Zafalon, C. R. Valêncio, and R. C. G. de Souza, "Project scope management: A strategy oriented to the requirements engineering," presented at the Int. Conf. Enterprise Inf. Syst., 2017.
- [16] O. Shmueli and B. Ronen, "Excessive software development: Practices and penalties," *Int. J. Project Manage.*, vol. 35, no. 1, pp. 13–27, Jan. 2017, doi: 10.1016/j.ijproman.2016.10.002.
- [17] B. Komal, U. I. Janjua, F. Anwar, T. M. Madni, M. F. Cheema, M. N. Malik, and A. R. Shahid, "The impact of scope creep on project success: An empirical investigation," *IEEE Access*, vol. 8, pp. 125755–125775, 2020, doi: 10.1109/access.2020.3007098.
- [18] R. T. Hans and E. Mnkandla, "To what extent do South African software organizations lack project success factors?" presented at the Int. Conf. Contemporary Comput. Inform. (IC3I), 2019.
- [19] R. Ahmad, C. Abdul Rehman, and M. Ilyas, "Linking the project scope with project success: The moderating role of managerial expertise in IT," *Research*, vol. 20, no. 4, pp. 815–824, 2019.
- [20] M. Ibraigheeth and S. A. Fadzli, "Core factors for software projects success," JOIV: Int. J. Informat. Vis., vol. 3, no. 1, pp. 69–74, Jan. 2019.
- [21] K. L. Madhuri, V. Suma, and U. M. Mokashi, "A triangular perception of scope creep influencing the project success," *Int. J. Bus. Inf. Syst.*, vol. 27, no. 1, p. 69, 2018.
- [22] F. Valdes-Souto, "Earned scope management: Scope performance evaluation for software projects considering people and effort as resources," presented at the 7th Int. Conf. Softw. Eng. Res. Innov. (CONISOFT), 2019.
- [23] A. H. Yousef, A. Gamal, A. Warda, and M. Mahmoud, "Software projects success factors identification using data mining," presented at the Int. Conf. Comput. Eng. Syst., 2006.
- [24] R. M. Adzmi and Z. Hassan, "A theoretical framework of critical success factors on information technology project management during project planning," *Int. J. Eng. Technol.*, vol. 7, no. 4, pp. 650–655, 2018.
- [25] T. A. Kurniawan, D. S. Rusdianto, A. H. Brata, F. Amalia, A. Santoso, and D. I. Raharjo, "An exploratory study of requirements engineering practices in Indonesia–Part 2: Efforts, processes and techniques," *J. Inf. Technol. Comput. Sci.*, vol. 5, no. 1, pp. 65–74, 2020.
- [26] K. Wnuk, T. Gorschek, D. Callele, E.-A. Karlsson, E. Ahlin, and B. Regnell, "Supporting scope tracking and visualization for very largescale requirements engineering-utilizing FSC+, decision patterns, and atomic decision visualizations," *IEEE Trans. Softw. Eng.*, vol. 42, no. 1, pp. 47–74, Jan. 2016, doi: 10.1109/tse.2015.2445347.

VOLUME 9, 2021

- [27] R. Stoica and P. Brouse, "IT project failure: A proposed four-phased adaptive multi-method approach," *Procedia Comput. Sci.*, vol. 16, pp. 728–736, 2013.
- [28] M. A. Ellatif, M. S. Farhan, M. K. Elmahy, and K. A. Mohamed, "Using ontology-based concept maps to reduce requirement engineering errors," *Int. J. Comput. Sci. Inf. Secur. (IJCSIS)*, vol. 15, no. 6, pp. 292–303, 2017.
- [29] I. Jebreen, R. Wellington, and S. G. MacDonell, "Understanding feasibility study approach for packaged software implementation by SMEs," in *Information System Development*. Cham, Switzerland: Springer, 2014, pp. 29–43.
- [30] M. Rose and W. Jotham, "Intelligent Decision Support (IDS) in software risk management based on data mining, rough sets and decision theory," *Afr. J. Sci., Technol. Eng. (AJSTE*, vol. 1, no. 1, pp. 82–96, 2020.
- [31] E. Mkoba and C. Marnewick, "IT project success," presented at the Annu. Conf. South Afr. Inst. Comput. Scientists Inf. Technologists (SAICSIT), 2016.
- [32] S. Tariq, N. Ahmad, M. U. Ashraf, A. M. Alghamdi, and A. S. Alfakeeh, "Measuring the impact of scope changes on project plan using EVM," *IEEE Access*, vol. 8, pp. 154589–154613, 2020, doi: 10.1109/access.2020.3018169.
- [33] A. A. Khanfar, R. K. Mavi, and F. Jie, "Prioritizing critical failure factors of IT projects with fuzzy analytic hierarchy process," presented at the AIP Conf., 2018.
- [34] K. S. Khan, M. K. Shahid, and S. F. Ahmad, "Critical success factors for IT projects in the telecom sector," *Ind. Eng. Lett.*, vol. 5, no. 11, pp. 1–14, 2015.
- [35] M. Ajmal, M. Khan, and H. Al-Yafei, "Exploring factors behind project scope creep-stakeholders' perspective," *Int. J. Manag. Projects Bus.*, vol. 13, no. 3, pp. 483–504, Nov. 2019.
- [36] A. Hussain, E. O. C. Mkpojiogu, and F. M. Kamal, "The role of requirements in the success or failure of software projects," *Int. Rev. Manage. Marketing*, vol. 6, no. S7, pp. 306–311, 2016.
- [37] A. Shende, I. College of Commerce, Science, and J. Pawar, "Study and analysis of factors responsible for failures of software project," *Int. J. Comput. Trends Technol.*, vol. 41, no. 2, pp. 81–86, Nov. 2016.
- [38] A. S. Pillai, "A study on the software requirements elicitation issues-its causes and effects," presented at the 3rd World Congr. Inf. Commun. Technol. (WICT), 2013.
- [39] A. Hussain and E. O. Mkpojiogu, "Requirements: Towards an understanding on why software projects fail," in *Proc. AIP Conf.*, 2016, vol. 1761, no. 1, Art. no. 020046.
- [40] A. Shah, M. Ali Alasow, F. Sajjad, and J. Javed Akbar Baig, "An evaluation of software requirements tools," in *Proc. 8th Int. Conf. Intell. Comput. Inf. Syst. (ICICIS)*, Dec. 2017, pp. 278–283.
- [41] A. Hussain Afzal Zahid, M. Waji Haider, M. Shoaib Farooq, A. Abid, and A. Ali, "A critical analysis of software failure causes from project management perspectives," VFAST Trans. Softw. Eng., pp. 113–119, Nov. 2018.
- [42] P. Mohagheghi and M. Jorgensen, "What contributes to the success of IT projects? Success factors, challenges and lessons learned from an empirical study of software projects in the Norwegian public sector," presented at the IEEE/ACM 39th Int. Conf. Softw. Eng. Companion (ICSE-C), 2017.
- [43] J. S. Busch, "Project delivery problems: Why can't we make them go away?" presented at the PICMET, 2008.
- [44] E. Bjarnason, K. Wnuk, and B. Regnell, "Overscoping: Reasons and consequences-A case study on decision making in software product management," presented at the 4th Int. Workshop Softw. Product Manage., 2010.
- [45] E. Bjarnason, K. Wnuk, and B. Regnell, "Are you biting off more than you can chew? A case study on causes and effects of overscoping in large-scale software engineering," *Inf. Softw. Technol.*, vol. 54, no. 10, pp. 1107–1124, Oct. 2012.
- [46] R. T. Hans, "Work breakdown structure: A tool for software project scope verification," 2013, arXiv:1308.2876. [Online]. Available: http://arxiv.org/abs/1308.2876
- [47] A. B. Badiru, STEP Project Management: Guide for Science, Technology, and Engineering Projects. Boca Raton, FL, USA: CRC Press, 2009.
- [48] S. R. Schach and S. R. Schach, Object-Oriented and Classical Software Engineering. New York, NY, USA: McGraw-Hill, 2007.
- [49] M. Bhardwaj, A. Rana, and N. K. Sharma, "How software size influence productivity and project duration," *Int. J. Electr. Comput. Eng. (IJECE)*, vol. 9, no. 3, p. 2006, Jun. 2019, doi: 10.11591/ijece.v9i3.pp2006-2017.
- [50] Information Technology-Software Measurement-Functional Size Measurement Part 6: Guide for Use of Series and Related International Standards, Standard ISO/IEC 14143 2012.

- [51] I. U. Hassan, N. Ahmad, and B. Zuhaira, "Calculating completeness of software project scope definition," *Inf. Softw. Technol.*, vol. 94, pp. 208–233, Feb. 2018, doi: 10.1016/j.infsof.2017.10.010.
- [52] P. Pospieszny, B. Czarnacka-Chrobot, and A. Kobyliński, "Application of function points and data mining techniques for software estimationa combined approach," in *Software Measurement* (Lecture Notes in Business Information Processing). Cham, Switzerland: Springer, 2015, ch. 7, pp. 96–113.
- [53] C. P. Rao, P. S. Kumar, S. R. Sree, and J. Devi, "An agile effort estimation based on story points using machine learning techniques," presented at the 2nd Int. Conf. Comput. Intell. Inform. Singapore: Springer, 2018, Ch. 20.
- [54] S. R. Sree and C. P. Rao, "A study on application of soft computing techniques for software effort estimation," presented at the Journey Towards Bio-Inspired Techn. Softw. Eng. Cham, Switzerland: Springer, 2020.
- [55] J. Pollack, J. Helm, and D. Adler, "What is the iron triangle, and how has it changed?" Int. J. Manag. Projects Bus., vol. 11, no. 2, pp. 527–547, May 2018, doi: 10.1108/ijmpb-09-2017-0107.
- [56] S. M. A. Suliman and G. Kadoda, "Factors that influence software project cost and schedule estimation," presented at the Sudan Conf. Comput. Sci. Inf. Technol. (SCCSIT), 2017.
- [57] R. N. Kasayu, A. N. Hidayanto, and P. I. Sandhyaduhita, "Critical success factors of software development projects using analytic hierarchy process: A case of Indonesia," *Int. J. Innov. Learn.*, vol. 22, no. 1, pp. 1–22, 2017.
- [58] H. Taherdoost and A. Keshavarzsaleh, "A theoretical review on IT project success/failure factors and evaluating the associated risks," *Math. Comput. Methods Elect. Eng.*, pp. 80–88, Aug. 2015.
 [59] T. O. Lehtinen, M. V. Mäntylä, J. Vanhanen, J. Itkonen, and
- [59] T. O. Lehtinen, M. V. Mäntylä, J. Vanhanen, J. Itkonen, and C. Lassenius, "Perceived causes of software project failures–An analysis of their relationships," *Inf. Softw. Technol.*, vol. 56, no. 6, pp. 623–643, 2014.
- [60] P. Savolainen, J. J. Ahonen, and I. Richardson, "Software development project success and failure from the supplier's perspective: A systematic literature review," *Inf. Softw. Technol.*, vol. 30, no. 4, pp. 458–469, May 2012.
- [61] V. Garousi, A. Tarhan, D. Pfahl, A. Coşkunçay, and O. Demirörs, "Correlation of critical success factors with success of software projects: An empirical investigation," *Softw. Qual. J.*, vol. 27, no. 1, pp. 429–493, Mar. 2019.
- [62] K. M. Whitney and C. B. Daniels, "The root cause of failure in complex IT projects: Complexity itself," *Procedia Comput. Sci.*, vol. 20, pp. 325–330, 2013.
- [63] D. Carlton and K. Peszynski, "Situational Incompetence: The failure of governance in the management of large scale IT projects," presented at the Int. Work. Conf. Transf. Diffusion IT, 2018.
- [64] A. Mersino. (2019). Agile Project Success Rates 2x Higher Than Traditional Projects. Vitality Chicago. [Online]. Available: https://vitalitychicago.com/blog/agile-projects-are-more-successfultraditional-projects/
- [65] A. R. Adam and M. Danaparamita, "Understanding the influence of poor scope management affecting the successful of an IT project," presented at the Int. Conf. Inf. Manage. Technol. (ICIMTech), 2016.
- [66] Q. H. S. Al-Rubaiei, F. A. A. Nifa, and S. Musa, "Project scope management through multiple perspectives: A critical review of concepts," in *Proc. AIP Conf.*, 2018, vol. 2016, no. 1, Art. no. 020025.
- [67] T. Williams, H. Vo, K. Samset, and A. Edkins, "The front-end of projects: A systematic literature review and structuring," *Prod. Planning Control*, vol. 30, no. 14, pp. 1137–1169, Oct. 2019.
- [68] K. Schwalbe, Information Technology Project Management. Boston, MA, USA: Cengage Learning, 2018.
- [69] R. Attri, N. Dev, and V. Sharma, "Interpretive structural modelling (ISM) approach: An overview," *Res. J. Manage. Sci.*, vol. 2, no. 2, pp. 3–8, 2013.
- [70] E. Ziemba and I. Kolasa, "Risk factors relationships for information systems projects-insight from polish public organizations," in *Information Technology for Management*. Heidelberg, Germany: Springer, 2016, pp. 55–76.
- [71] G.-H. Tzeng and J.-J. Huang, Multiple Attribute Decision Making: Methods and Applications. Boca Raton, FL, USA: CRC Press, 2011.
- [72] S. K. Mangla, S. Luthra, N. Rich, D. Kumar, N. P. Rana, and Y. K. Dwivedi, "Enablers to implement sustainable initiatives in agri-food supply chains," *Int. J. Prod. Econ.*, vol. 203, pp. 379–393, Sep. 2018, doi: 10.1016/j.ijpe.2018.07.012.
- [73] H. Shakeri and M. Khalilzadeh, "Analysis of factors affecting project communications with a hybrid DEMATEL-ISM approach (A case study in Iran)," *Heliyon*, vol. 6, no. 8, Aug. 2020, Art. no. e04430, doi: 10.1016/j.heliyon.2020.e04430.

- [74] D.-Q. Zhou, L. Zhang, and H.-W. Li, "A study of the system's hierarchical structure through integration of DEMATEL and ISM," presented at the 5th Int. Conf. Mach. Learn. Cybern., Dalian, China, 2006.
- [75] Chauhan, Singh, and Jharkharia, "An interpretive structural modeling (ISM) and decision-making trail and evaluation laboratory (DEMATEL) method approach for the analysis of barriers of waste recycling in India," *J. Air Waste Manage. Assoc.*, vol. 68, no. 2, pp. 100–110, Feb. 2018, doi: 10.1080/10962247.2016.1249441.
- [76] A. K. Sangaiah, O. W. Samuel, X. Li, M. Abdel-Basset, and H. Wang, "Towards an efficient risk assessment in software projects– fuzzy reinforcement paradigm," *Comput. Elect. Eng.*, vol. 71, pp. 833–846, Oct. 2018.
- [77] D. L. Hughes, Y. K. Dwivedi, N. P. Rana, and A. C. Simintiras, "Information systems project failure-analysis of causal links using interpretive structural modelling," *Prod. Planning Control*, vol. 27, no. 16, pp. 1313–1333, 2016, doi: 10.1080/09537287.2016.1217571.
- [78] A. Barve, A. Kanda, and R. Shankar, "Analysis of interaction among the barriers of third party logistics," *Int. J. Agile Syst. Manage.*, vol. 2, no. 1, pp. 109–129, 2007.
- [79] S. Luthra, D. Garg, and A. Haleem, "An analysis of interactions among critical success factors to implement green supply chain management towards sustainability: An Indian perspective," *Resour. Policy*, vol. 46, pp. 37–50, Dec. 2015.
- [80] D. K. Yadav and A. Barve, "Analysis of critical success factors of humanitarian supply chain: An application of interpretive structural modeling," *Int. J. Disaster Risk Reduction*, vol. 12, pp. 213–225, Jun. 2015.
- [81] J. Wan and Z. Wang, "Case study on E-business v corp. Software project risk management with interpretive structural modeling," *Open J. Social Sci.*, vol. 3, no. 4, pp. 1–7, 2015, doi: 10.4236/jss.2015.34001.
- [82] D. L. Hughes, N. P. Rana, and Y. K. Dwivedi, "Elucidation of IS project success factors: An interpretive structural modelling approach," Ann. Oper. Res., vol. 285, nos. 1–2, pp. 35–66, Feb. 2020, doi: 10.1007/s10479-019-03146-w.
- [83] S. M. Rabani and A. Talebbeydokhti, "Factors affecting the success of IT projects using interpretive structural approach (case study: State tax administration of Kohgiluyeh)," *Int. Academic J. Accounting Financial Manage.*, vol. 3, no. 2, pp. 70–85, 2016.
- [84] A.-Y. Chang, K.-J. Hu, and Y.-L. Hong, "An ISM-ANP approach to identifying key agile factors in launching a new product into mass production," *Int. J. Prod. Res.*, vol. 51, no. 2, pp. 582–597, Jan. 2013, doi: 10.1080/00207543.2012.657804.
- [85] S. Sharma, N. Hasteer, S. P. Mishra, and J.-P. Van Belle, "Identifying the contextual relationship among the Agile adoption factors through interpretive structural modeling," presented at the Int. Conf. Inf. Technol. (InCITe)-Next Gener. IT Summit Theme-Internet Things: Connect Your Worlds, 2016.
- [86] P. Jain, A. Sharma, and L. Ahuja, "ISM Based Identification of Quality Attributes for Agile Development," presented at the 5th Int. Conf. Rel., Infocom Technol. Optim. (Trends and Future Directions)(ICRITO), 2016.
- [87] Y. Qu, M.-J. Yuan, and F. Liu, "The risk factor analysis for software project based on the interpretative structural modelling method," in *Proc. Int. Conf. Mach. Learn. Cybern.*, vol. 3, Jul. 2012, pp. 1019–1024.
- [88] J. K. Bajwa, K. Singh, and N. Sharma, "Modelling the key challenges in agile adoption: An Indian perspective," *Int. J. Innov. Eng. Technol.* (*IJIET*), vol. 7, no. 1, 2016.
- [89] P. Sharma and A. L. Sangal, "Building a hierarchical structure model of enablers that affect the software process improvement in software SMEs—A mixed method approach," *Comput. Standards Interfaces*, vol. 66, Oct. 2019, Art. no. 103350, doi: 10.1016/j.csi.2019.04.009.
- [90] W.-H. Tsai, J.-D. Leu, J.-Y. Liu, S.-J. Lin, and M. J. Shaw, "A MCDM approach for sourcing strategy mix decision in IT projects," *Expert Syst. Appl.*, vol. 37, no. 5, pp. 3870–3886, May 2010, doi: 10.1016/j.eswa.2009.11.031.
- [91] Z.-P. Fan, W.-L. Suo, and B. Feng, "Identifying risk factors of IT outsourcing using interdependent information: An extended DEMATEL method," *Expert Syst. Appl.*, vol. 39, no. 3, pp. 3832–3840, Feb. 2012, doi: 10.1016/j.eswa.2011.09.092.
- [92] K. Suresh and R. Dillibabu, "A novel fuzzy mechanism for risk assessment in software projects," *Soft Comput.*, vol. 24, no. 3, pp. 1683–1705, Feb. 2020, doi: 10.1007/s00500-019-03997-2.
- [93] M. M. Asad, V. Mohammadi, and M. Shirani, "Modeling flexibility capabilities of IT-based supply chain, using a grey-based DEMATEL method," *Procedia Econ. Finance*, vol. 36, pp. 220–231, 2016, doi: 10.1016/s2212-5671(16)30033-8.

- [94] Y.-S. Chen, H.-M. Chuang, A. K. Sangaiah, C.-K. Lin, and W.-B. Huang, "A study for project risk management using an advanced MCDM-based DEMATEL-ANP approach," *J. Ambient Intell. Humanized Comput.*, vol. 10, no. 7, pp. 2669–2681, Jul. 2019, doi: 10.1007/s12652-018-0973-2
- [95] R. Majumdar, P. K. Kapur, and S. K. Khatri, "Assessing software upgradation attributes and optimal release planning using DEMATEL and MAUT," *Int. J. Ind. Syst. Eng.*, vol. 31, no. 1, p. 70, 2019.
- [96] A. Agrawal, M. Alenezi, R. Kumar, and R. A. Khan, "Measuring the sustainable-security of Web applications through a fuzzy-based integrated approach of AHP and TOPSIS," *IEEE Access*, vol. 7, pp. 153936–153951, 2019, doi: 10.1109/ACCESS.2019.2946776.
- [97] A. Agrawal, A. K. Pandey, A. Baz, H. Alhakami, W. Alhakami, R. Kumar, and R. A. Khan, "Evaluating the security impact of healthcare Web applications through fuzzy based hybrid approach of multi-criteria decision-making analysis," *IEEE Access*, vol. 8, pp. 135770–135783, 2020, doi: 10.1109/access.2020.3010729.
- [98] F. A. Al-Zahrani, "Evaluating the usable-security of healthcare software through unified technique of fuzzy logic, ANP and TOPSIS," *IEEE Access*, vol. 8, pp. 109905–109916, 2020, doi: 10.1109/access.2020.3001996.
- [99] R. Kumar, A. Irshad Khan, Y. B. Abushark, M. M. Alam, A. Agrawal, and R. A. Khan, "An integrated approach of fuzzy logic, AHP and TOPSIS for estimating usable-security of Web applications," *IEEE Access*, vol. 8, pp. 50944–50957, 2020, doi: 10.1109/access.2020.2970245.
- [100] R. Kumar, A. I. Khan, Y. B. Abushark, M. M. Alam, A. Agrawal, and R. A. Khan, "A knowledge-based integrated system of hesitant fuzzy set, AHP and TOPSIS for evaluating security-durability of Web applications," *IEEE Access*, vol. 8, pp. 48870–48885, 2020, doi: 10.1109/access.2020.2978038.
- [101] R. Kumar, A. Baz, H. Alhakami, W. Alhakami, M. Baz, A. Agrawal, and R. A. Khan, "A hybrid model of hesitant fuzzy decision-making analysis for estimating usable-security of software," *IEEE Access*, vol. 8, pp. 72694–72712, 2020, doi: 10.1109/access.2020.2987941.
- [102] M. Alenezi, A. Agrawal, R. Kumar, and R. A. Khan, "Evaluating performance of Web application security through a fuzzy based hybrid multi-criteria decision-making approach: Design tactics perspective," *IEEE Access*, vol. 8, pp. 25543–25556, 2020, doi: 10.1109/access.2020.2970784.
- [103] A. K. Pandey, A. I. Khan, Y. B. Abushark, M. M. Alam, A. Agrawal, R. Kumar, and R. A. Khan, "Key issues in healthcare data integrity: Analysis and recommendations," *IEEE Access*, vol. 8, pp. 40612–40628, 2020, doi: 10.1109/access.2020.2976687.
- [104] M. Zarour, M. T. J. Ansari, M. Alenezi, A. K. Sarkar, M. Faizan, A. Agrawal, R. Kumar, and R. A. Khan, "Evaluating the impact of blockchain models for secure and trustworthy electronic healthcare records," *IEEE Access*, vol. 8, pp. 157959–157973, 2020, doi: 10.1109/access.2020.3019829.
- [105] B. Kitchenham and S. Charters, "Guidelines for performing systematic literature reviews in software engineering, version 2.3," School Comput. Sci. Math., Keele Univ., Keele, U.K., Tech. Rep. EBSE-2007-01, 2007.
- [106] K. Petersen, S. Vakkalanka, and L. Kuzniarz, "Guidelines for conducting systematic mapping studies in software engineering: An update," *Inf. Softw. Technol.*, vol. 64, pp. 1–18, Aug. 2015.
- [107] F. Q. B. da Silva, A. L. M. Santos, S. Soares, A. C. C. França, C. V. F. Monteiro, and F. F. Maciel, "Six years of systematic literature reviews in software engineering: An updated tertiary study," *Inf. Softw. Technol.*, vol. 53, no. 9, pp. 899–913, Sep. 2011, doi: 10.1016/j.infsof.2011.04.004.
- [108] J. Goedeke, M. Mueller, and O. Pankratz, "Uncovering the causes of information system project failure," presented at the 23rd Americas Conf. Inf. Syst., Boston, MA, USA, 2017.
- [109] A. Taghavi and C. Woo, "The role clarity framework to improve requirements gathering," ACM Trans. Manage. Inf. Syst., vol. 8, nos. 2–3, pp. 1–16, Aug. 2017, doi: 10.1145/3083726.
- [110] C. Symons, A. Abran, C. Ebert, and F. Vogelezang, "Measurement of software size: Advances made by the COSMIC community," presented at the Joint Conf. Int. Workshop Softw. Meas. Int. Conf. Softw. Process Product Meas., 2016.
- [111] A. Prasanth, S. Valsala, and S. Soomro, "A novel approach in calculating stakeholder priority in requirements elicitation," presented at 4th IEEE Int. Conf. Eng. Technol. Appl. Sci. (ICETAS), 2018.
- [112] A. Jarzębowicz and W. ólesiński, "Assessing effectiveness of recommendations to requirements-related problems through interviews with experts," presented at the Federated Conf. Comput. Sci. Inf. Syst., 2018.

- [113] W. Sardjono and A. Retnowardhani, "Analysis of failure factors in information systems project for software implementation at the organization," presented at the Int. Conf. Inf. Manage. Technol. (ICIMTech), 2019.
- [114] F. Fellir, G. E. Park, K. Nafil, and L. Chung, "Coping with uncertainties in estimating requirements development effort," presented at the 6th Int. Conf. Control Eng. Inf. Technol. (CEIT), Istanbul, Turkey, 2018.
- [115] J. A. Aguilar, A. Zaldívar-Colado, C. Tripp-Barba, R. Espinosa, S. Misra, and C. E. Zurita, "A survey about the impact of requirements engineering practice in small-sized software factories in Sinaloa, Mexico," in *Proc. Int. Conf. Comput. Sci. Appl.*, 2018, pp. 331–340.
- [116] E. A. de Carvalho, J. O. Gomes, A. Jatobá, M. F. da Silva, and P. V. R. de Carvalho, "Employing resilience engineering in eliciting software requirements for complex systems: Experiments with the functional resonance analysis method (FRAM)," *Cognition, Technol. Work*, Jan. 2020, doi: 10.1007/s10111-019-00620-0.
- [117] R. Yesudas, M. Efatmaneshnik, and K. Joiner, "Identifying focal points in IT project governance using a synthetic and systems thinking approach," in *Proc. Int. Conf. Complex Syst. Design Manage.*, 2019, pp. 79–92.
- [118] Y.-C. Tu, E. Tempero, and C. Thomborson, "An experiment on the impact of transparency on the effectiveness of requirements documents," *Empirical Softw. Eng.*, vol. 21, no. 3, pp. 1035–1066, Jun. 2016, doi: 10.1007/s10664-015-9374-8.
- [119] A. Wahbeh, S. Sarnikar, and O. El-Gayar, "A socio-technical-based process for questionnaire development in requirements elicitation via interviews," *Requirements Eng.*, vol. 25, no. 3, pp. 295–315, Sep. 2020, doi: 10.1007/s00766-019-00324-x.
- [120] M. R. Asadabadi, E. Chang, O. Zwikael, M. Saberi, and K. Sharpe, "Hidden fuzzy information: Requirement specification and measurement of project provider performance using the best worst method," *Fuzzy Sets Syst.*, vol. 383, pp. 127–145, Mar. 2020, doi: 10.1016/j.fss.2019.06.017.
- [121] F. Fellir, K. Nafil, R. Touahni, and L. Chung, "Improving case based software effort estimation using a multi-criteria decision technique," in *Software Engineering and Algorithms in Intelligent Systems* (Advances in Intelligent Systems and Computing). Cham, Switzerland: Springer, 2019, ch. 46, pp. 438–451.
- [122] Q. U. Ain, W. H. Butt, M. W. Anwar, F. Azam, and B. Maqbool, "A systematic review on code clone detection," *IEEE Access*, vol. 7, pp. 86121–86144, 2019.
- [123] A. Elhabbash, M. Salama, R. Bahsoon, and P. Tino, "Self-awareness in software engineering: A systematic literature review," ACM Trans. Auton. Adapt. Syst., vol. 14, no. 2, pp. 1–42, 2019.
- [124] A. Bajaj and O. P. Sangwan, "A systematic literature review of test case prioritization using genetic algorithms," *IEEE Access*, vol. 7, pp. 126355–126375, 2019.
- [125] M. Sánchez-Gordón and R. Colomo-Palacios, "Taking the emotional pulse of software engineering—A systematic literature review of empirical studies," *Inf. Softw. Technol.*, vol. 115, pp. 23–43, Nov. 2019.
- [126] M. F. Abrar, M. S. Khan, S. Ali, U. Ali, M. F. Majeed, A. Ali, B. Amin, and N. Rasheed, "Motivators for large-scale agile adoption from management perspective: A systematic literature review," *IEEE Access*, vol. 7, pp. 22660–22674, 2019.
- [127] H. U. Rahman, M. Raza, P. Afsar, H. U. Khan, and S. Nazir, "Analyzing factors that influence offshore outsourcing decision of application maintenance," *IEEE Access*, vol. 8, pp. 183913–183926, 2020.
- [128] T. J. Gandomani, Z. Tavakoli, H. Zulzalil, and H. K. Farsani, "The role of project manager in agile software teams: A systematic literature review," *IEEE Access*, vol. 8, pp. 117109–117121, 2020.
- [129] M. A. Akbar, S. Mahmood, A. Alsanad, M. Shafiq, A. Gumaei, and A. A.-A. Alsanad, "Organization type and size based identification of requirements change management challenges in global software development," *IEEE Access*, vol. 8, pp. 94089–94111, 2020.
- [130] M. A. Akbar, S. Mahmood, H. Alsalman, A. Razzaq, A. Gumaei, and M. T. Riaz, "Identification and prioritization of cloud based global software development best practices," *IEEE Access*, vol. 8, pp. 191242–191262, 2020.
- [131] A. A. B. Baqais and M. Alshayeb, "Automatic software refactoring: A systematic literature review," *Softw. Qual. J.*, vol. 28, no. 2, pp. 459–502, Jun. 2020.
- [132] H. Alsolai and M. Roper, "A systematic literature review of machine learning techniques for software maintainability prediction," *Inf. Softw. Technol.*, vol. 119, Mar. 2020, Art. no. 106214.
- [133] M. Pace, "A correlational study on project management methodology and project success," *J. Eng., Project, Prod. Manage.*, vol. 9, no. 2, pp. 56–65, Jul. 2019, doi: 10.2478/jeppm-2019-0007.

- [134] D. L. Hughes, N. P. Rana, and A. C. Siminitras, "The changing landscape of IS project failure: An examination of the key factors," *J. Enterprise Inf. Manage.*, vol. 30, no. 1, pp. 142–165, Feb. 2017, doi: 10.1108/jeim-01-2016-0029.
- [135] H. Alrumaih, A. Mirza, and H. Alsalamah, "Toward automated software requirements classification," presented at the 21st Saudi Comput. Soc. Nat. Comput. Conf. (NCC), 2018.
- [136] G. J. Hidding and J. M. Nicholas, "A new way of thinking about IT project management practices: Early empirical results," *J. Organizational Comput. Electron. Commerce*, vol. 27, no. 1, pp. 81–95, Jan. 2017, doi: 10.1080/10919392.2016.1263113.
- [137] B. Wachnik, "Critical success factors in IT projects-from the practice of polish clients," presented at the Commun. Papers of the Federated Conf. Comput. Sci. Inf. Syst., 2017.
- [138] M. Bezdrob, S. Brkić, D. Bajramović, M. Gram, and D. Burić, "Black Swans' in IT projects–Myth or reality," presented at the 9th Int. Conf. 'Enterprise Odyssey-Manag. Change Achieve Qual. Develop., 2018.
- [139] G. P. Sudhakar, "Critical failure factors (CFFs) of IT projects," Int. J. Manage. Res., vol. 4, no. 2, pp. 31–51, 2016.
- [140] H. Taherdoost and A. Keshavarzsaleh, "How to lead to sustainable and successful IT project management? Propose 5Ps guideline," *Int. J. Adv. Comput. Sci. Inf. Technol. (IJACSIT)*, pp. 1739–2296, 2018.
- [141] M. Aranyossy, B. Blaskovics, and Á. A. Horváth, "How universal are IT project success and failure factors? Evidence from hungary," *Inf. Syst. Manage.*, vol. 35, no. 1, pp. 15–28, Jan. 2018, doi: 10.1080/10580530.2017.1416943.
- [142] C. W. Chesterman, Jr., K. Castelle, and J. J. Shauger, "Interpreting barriers to success in software development and deployment using systems theory," *Int. J. Syst. Syst. Eng.*, vol. 7, no. 1/2/3, p. 143, 2016, doi: 10.1504/ijsse.2016.076122.
- [143] S. M. Sarif, S. Ramly, R. Yusof, N. A. A. Fadzillah, and N. Y. Bin Sulaiman, "Investigation of success and failure factors in IT project management," in *Proc. Int. Conf. Kansei Eng. Emotion Res.* Singapore: Springer, 2018, pp. 671–682.
- [144] D. L. Hughes, Y. K. Dwivedi, and N. P. Rana, "Mapping IS failure factors on PRINCE2 stages: An application of interpretive ranking process (IRP)," *Prod. Planning Control*, vol. 28, no. 9, pp. 776–790, Jul. 2017, doi: 10.1080/09537287.2017.1311431.
- [145] B. Yost, M. Coblenz, B. Myers, and J. Sunshine, "Software development practices, barriers in the field and the relationship to software quality," presented at the 10th ACM/IEEE Int. Symp. Empirical Softw. Eng. Meas. (ESEM), 2016.
- [146] I. Atoum, "A scalable operational framework for requirements validation using semantic and functional models," presented at the 2nd Int. Conf. Softw. Eng. Inf. Manage., 2019.
- [147] E. E. Surbakti, B. Purwandari, I. Solichah, and L. Kumaralalita, "Analysis of software development method selection," presented at the 3rd Int. Conf. Bus. Inf. Manage. (ICBIM), 2019.
- [148] G. S. Saini, S. K. Dubey, and S. K. Bharti, "Novel algorithm for software planning & development," presented at the 3rd Int. Conf. Adv. Inform. Comput. Res. (ICAICR), 2019.
- [149] W. Majerowicz and S. A. Shinn, "Schedule matters: Understanding the relationship between schedule delays and costs on overruns," presented at the IEEE Aerosp. Conf., 2016.

- [150] M. Haoues, A. Sellami, H. Ben-Abdallah, and O. Demirors, "Evaluating software security change requests: A COSMIC-based quantification approach," presented at the 45th Euromicro Conf. Softw. Eng. Adv. Appl. (SEAA), 2019.
- [151] W. Maalej, M. Nayebi, and G. Ruhe, "Data-driven requirements engineering-an update," presented at the IEEE/ACM 41st Int. Conf. Softw. Eng., Softw. Eng. Pract. (ICSE-SEIP), 2019.
- [152] K. Frakzkowski and B. Gładysz, "Stakeholder impact on the success and risk of failure of ICT projects in Poland," in *Proc. Asian Conf. Intell. Inf. Database Syst.*, 2016, pp. 530–540.

ISMA UL HASSAN received the M.S. degree in computer science from COMSATS University Islamabad, Pakistan, in 2014, where she is currently pursuing the Ph.D. degree with the Department of Computer Science. She is also a Lecturer with the Department of Computer Science, CUI. Her research interests include software engineering and software project management.

SOHAIL ASCHAR (Member, IEEE) received the degree (Hons.) in computer science from the University of Wales, U.K., in 1994, and the Ph.D. degree from the Faculty of Information Technology, Monash University, Clayton VIC, Australia, in 2006. In 2011, he joined as the Director of the University Institute of Information Technology, PMAS-Arid Agriculture University, Rawalpindi. He is currently working as a Professor and the Chairman of computer science with COMSATS

University Islamabad. He has taught and researched in data mining (including structural learning, classification, and privacy preservation in data mining, text, and web mining), big data analytics, data science, and information technology areas. He has published extensively (more than 150 publications) in international journals and conference proceedings. He has also consulted widely on information technology matters, especially in the framework of data mining and data science. He is a member of the Australian Computer Society (ACS) and the Higher Education Commission Approved Supervisor. He has also served as a Program Committee Member for numerous international conferences and regularly speaks at international conferences, seminars, and workshops. In 2004, he acquired the Australian Postgraduate Award for Industry. He is on the Editorial Team of well-reputed scientific journals.

. . .